\documentclass[conference,a4paper]{IEEEtran}
\IEEEoverridecommandlockouts
% The preceding line is only needed to identify funding in the first footnote. If that is unneeded, please comment it out.
\usepackage{hyperref}
\usepackage[english]{babel}
\usepackage{amsmath,amssymb,amsfonts}
\usepackage[utf8]{inputenc}
\usepackage{xcolor}
\usepackage{booktabs}
\usepackage{cite}
\usepackage{algorithmic}
\usepackage{graphicx}
\usepackage{pgfplots,filecontents}
\pgfplotsset{compat=1.14}
\usepackage{pgfplotstable}
\usepackage{subcaption}
\usepackage{textcomp}
\usepackage{siunitx}
\sisetup{detect-all}
\sisetup{separate-uncertainty = true}
\widowpenalty=5000  %to reduce effect of widow lines

\def\BibTeX{{\rm B\kern-.05em{\sc i\kern-.025em b}\kern-.08em
    T\kern-.1667em\lower.7ex\hbox{E}\kern-.125emX}}

\begin{document}
\begin{thebibliography}{00}
\bibitem{b1}  G. S. Kim and J. Yoon, “Development of intelligent foot with six-axis force/moment sensors for humanoid robot,” 2008 IEEE Int. Conf. Robot. Biomimetics, ROBIO 2008, pp. 939–944, 2008.
\bibitem{b2}  K. K. Şafak and T. B. Baturalp, “Design and analysis of a foot contact sensor for posture control of a biped robot,” ASME 2010 10th Bienn. Conf. Eng. Syst. Des. Anal. ESDA2010, vol. 3, no. July 2010, pp. 669–673, 2010.
\bibitem{b3} M. Crisóstomo, A. P. Coimbra, P. Ferreira, J. P. Ferreira, A. Vieira, “Calçado instrumentado para análise da marcha”, Patent PT 108143, 2017.
\bibitem{b4}  J. Park, S. J. Kim, Y. Na, and J. Kim, “Custom optoelectronic force sensor based ground reaction force (GRF) measurement system for providing absolute force,” 2016 13th Int. Conf. Ubiquitous Robot. Ambient Intell. URAI 2016, vol. 1, pp. 75–77, 2016.
\bibitem{b5}  J. Figueiredo et al., “Instrumented insole system for ambulatory and robotic walking assistance: First advances,” 2017 IEEE Int. Conf. Auton. Robot Syst. Compet. ICARSC 2017, pp. 116–121, 2017. 
\bibitem{b6}  H. Zhang, D. Zanotto, and S. K. Agrawal, “Estimating CoP Trajectories and Kinematic Gait Parameters in Walking and Running Using Instrumented Insoles,” IEEE Robot. Autom. Lett., vol. 2, no. 4, pp. 2159–2165, 2017.
\bibitem{b7}  S. Crea, M. Donati, S. M. M. De Rossi, C. Maria Oddo, and N. Vitiello, “A wireless flexible sensorized insole for gait analysis,” Sensors (Switzerland), vol. 14, no. 1, pp. 1073–1093, 2014.
\bibitem{b8}  L. S. Lincoln, S. J. M. Bamberg, E. Parsons, C. Salisbury, and J. Wheeler, “An elastomeric insole for 3-axis ground reaction force measurement,” Proc. IEEE RAS EMBS Int. Conf. Biomed. Robot. Biomechatronics, pp. 1512–1517, 2012.
\bibitem{b9} Flexiforce R : Standard and Custom OEM Force Sensing Solutions, 2017, $http://www.tekscan.com/flexiforce.html$.
\bibitem{b10} C. Lebosse, P. Renaud, B. Bayle and M. de Mathelin, "Modeling and Evaluation of Low-Cost Force Sensors," in IEEE Transactions on Robotics, vol. 27, no. 4, pp. 815-822, Aug. 2011.
\end{thebibliography}

\title{A Novel Wireless Instrumented Shoe for Ground Reaction Forces Analysis in Humanoids \\ \vspace{2.5 cm}
%\thanks{This work was supported by Fundação para a Ciência e Tecnologia (FCT) on the scope of project A2HR with reference PTDC/EEI-AUT/5141/2014}
}


% \author{\IEEEauthorblockN{Luís Almeida\IEEEauthorrefmark{1}, Vítor Santos\IEEEauthorrefmark{1}\IEEEauthorrefmark{2}, Filipe Silva\IEEEauthorrefmark{1}\IEEEauthorrefmark{3} }
% \IEEEauthorblockA{\IEEEauthorrefmark{1}Institute of Electronics and Informatics Engineering of Aveiro (IEETA), Aveiro, Portugal}
% \IEEEauthorblockA{\IEEEauthorrefmark{2}Department of Mechanical Engineering (DEM), Aveiro, Portugal}
% \IEEEauthorblockA{\IEEEauthorrefmark{3}Department of Electronics, Telecommunications and Informatics (DETI), Aveiro, Portugal}
% luis93.pa@gmail.com, vitor@ua.pt, fmsilva@ua.pt
% }

%\author{\IEEEauthorblockN{Lu\'{i}s Almeida}
%\IEEEauthorblockA{\textit{Department of Mechanical Engineering (DEM) } \\
%\textit{Aveiro University}\\
%Aveiro, Portugal \\
%luis93.pa@gmail.com}
%\and
%\IEEEauthorblockN{V\'{i}tor Santos}
%\IEEEauthorblockA{\textit{Department of Mechanical Engineering (DEM)} \\
%\textit{Aveiro University}\\
%Aveiro, Portugal \\
%vitor@ua.pt}
%\and
%\IEEEauthorblockN{Filipe Silva}
%\IEEEauthorblockA{\textit{Department of Electronics, Telecommunications and Informatics (DETI)} \\
%\textit{Aveiro University}\\
%Aveiro, Portugal  \\
%fmsilva@ua.pt}
%}

\maketitle


\begin{abstract}
The measurement of ground reaction forces (GRFs) is crucial in the biomechanical analysis of gait and other motor activities. Current humanoid robots mimic some of the biological adaptation methods found in humans. Therefore, studying the GRFs in these robots allows not only to improve their overall performance, but also to extrapolate possible techniques for human walking rehabilitation. The balance of a humanoid robot may be compromised when it walks and, even more, when it walks across multiple grounds with different frictions. This paper presents a system to be seamlessly installed on a walking humanoid robot to measure normal and tangential ground reaction forces. The proposed solution is a cost-effective, lightweight and wirelessly instrumented shoe (ITshoe) for real-time measuring of the GRFs. \\
%, in order to generate patterns that allows it to recognize different grounds. 
\end{abstract}

\begin{IEEEkeywords}
Force sensors, Ground reaction forces, Humanoid robot, Instrumented shoe.
\end{IEEEkeywords}




\section{Introduction}
Walking robots compete to mimic some of the biological adaptation methods found in humans. Studying these humanoid robots grants the possibility to deduce methods for human walking rehabilitation. During walking, the impulsive force between the stance foot and the ground may affect the stability and reliability of biped robots seriously. Bearing in mind the need to reach a properly and safely humanoid robot locomotion, with a precise and robust control, it is crucial to measure the contact force between its feet and the ground.\par

This work is part of a project entitled "Automatic adaptation of an humanoid robot gait to different floor-robot friction coefficients", that aims to achieve adaptive humanoid robot walking through several types of floors. Following this idea, the principal aim of this work is the development of a system capable of detecting and analysing the complete ground reaction forces (GRFs). The major requirements for such a system are small weight and size, self-contained and no impediment of functional mobility. \par

The current work presents a novel instrumented shoe (ITshoe) capable of measuring the three components of the GRFs. The solution detects and wirelessly transmits the GRFs in real-time, it is customizable for different robots, it allows more parameters than integrated solutions in robots, and it can be easily installed independently of the robot control. The ITshoe is designed to be used with the humanoid robot NAO and has two main parts: Outer Shoe (OSh) and Inner Shoe (ISh). The OSh is the instrumented part of the shoe, whereas the ISh is the link between the robot's foot and the OSh. \par

The remainder of the paper is divided as follows. Section II presents a review of modern developments connected to this work, such as commonly used sensors and existing instrumented systems. The developed prototype is presented in Section III. The system's architecture is outlined in Section IV. GRFs experiments and results are discussed in Section V. Finally, Section VI presents the conclusions and future challenges. 

\section{Related Work}

Multiple systems have been developed for GRFs estimation, being commonly used in two major fields: robotics and biomechanics. Most of today’s humanoid robots are equipped with force sensing capabilities at the feet for stability purposes, based on commercial or custom-made solutions. Besides being expensive, commercial solutions adopted are tailored to specific needs in which the main concern is to provide an accurate estimate of the robot’s center-of-pressure (CoP). Kim \textit{et al.} \cite{b1} developed an intelligent foot for a humanoid robot using two manufactured six-axis force/moment sensors (twelve sensors). The developed foot operates similarly to a human foot, but it turns out to be a little unpractical and an expensive solution, since it requires a challenging process to be used with existing humanoids. On the other hand, Şafak \textit{et al.} \cite{b2} use four force sensitive resistors (FSRs) to build a foot contact sensor for a biped robot. Although, no trials have been performed, over numerical simulations the authors achieved a viable method to monitor postural stability of biped robots.\par

Systems that allow quantitative analysis of human gait are normally classified into: ground mounted force platforms and wearable instrumented systems (e.g. shoes and insoles). Force plates have the disadvantage of limiting the number of steps that can be done and are expensive, hence, the increasing advantage of using portable instrumented systems. These systems are usually designed with strain gauges, piezoelectrics, force sensitive resistors (FSR) or 3- to 6-axis transducers. Different wearable systems have been designed and developed over the recent years in order to measure the GRFs, to estimate the coordinates of the CoP and to monitor gait events (see for example \cite{ b3, b4, b5, b6}). However, most of the solutions are designed to be used directly with humans, and few allow the measurement of the complete GRFs.

Zhang \textit{et al.} \cite{b6} features a fully-portable instrumented insoles capable of measuring spatio-temporal gait parameters and CoP trajectories during walking and running assignments, using a multi-cell piezoresistive sensor. Crea \textit{et al.} \cite{b7} presents a pressure-sensitive foot insole for real-time monitoring of plantar pressure distribution during walking. The system includes 64 pressure-sensitive elements and a electronic board to acquire and transmit the data through bluetooth with a sampling frequency of \SI{100}{\hertz}. The proposed solution is developed to do a gait analysis based only on the vertical ground reaction forces (vGRFs) and the coordinates of the CoP. The estimation of the vGRFs showed a high qualitative correlation despite the significant difference in terms of actual values measurements against a force-platform. Lincoln \textit{et al.} \cite{b8} designed a low-cost silicone insole capable of measuring the complete three dimensional reaction force vector. The developed sensor features a error within 10\% and appears to have a temperature dependency. 
Many solutions within this theme are developed, but none seems suited to be adapted and used with different humanoids.


\section{ITshoe Prototype Design}

%\Figure[h!][width=49mm]{osh}
%{ITshoe schematic structure.\label{fig.a5}}



Fig. \ref{fig:test} schematically illustrates the design of the ITshoe. It has a total weight of \SI{210}{g} (approximately 4\% of the humanoid robot NAO weight) and dimensions measuring \SI{125 x 65 x 18}{mm} to accommodate all the necessary hardware. The shoe is divided into two parts: OSh (Fig. \ref{fig:sub1}) and ISh (Fig. \ref{fig:sub2}). The OSh is the instrumented part of the shoe and the ISh is used to change the robot foot geometry allowing a tight fit with the robot's foot, thus ensuring a correct and complete force transfer between the foot and the OSh. \par

The building material used in the manufacture of the ITshoe is acrylic since it presents low density (\SI{1.18}{g/cm^3}) together with an acceptable modulus of elasticity (\SI{3.2}{GPa}). In addition, the acrylic allows the user to verify whether the hardware is well positioned after a specific experience, thanks to its transparency, and it is also an excellent electrical insulator.

\begin{figure}[h!]
\centering
\begin{subfigure}{.5\columnwidth}
  \centering
  \includegraphics[width=1\textwidth]{osh}
  \caption{OSh.}
  \label{fig:sub1}
\end{subfigure}%
\begin{subfigure}{.5\columnwidth}
  \centering
  \includegraphics[width=1\textwidth]{ish}
  \caption{ISh.}
  \label{fig:sub2}
\end{subfigure}
\caption{ITshoe model. The position of the force sensor that measures the tangential and the normal forces is represents by the numbers 1 and 2, respectively (left).}
\label{fig:test}
\end{figure}

\subsection{ITshoe structure}


%\begin{figure}[h] 
 %   \def\svgwidth{\columnwidth}
 %   \input{a89.pdf_tex}
 %   \caption{ITshosadastructure.\label{fisda}}
%\end{figure}


The ITshoe presented in Fig. \ref{fig.a6} is divided into three units: sensing, acquisition and streaming. The sensing unit is composed of piezo-resistive A301 flexiforce sensors (Fig. \ref{fig.a6} block C, number 6): 4 with a standard force of 4.44N for the measurement of tangential forces (positioned at \ang{45}) and 4 with a standard force of 111N to register the normal forces. The sensors have a reported typical linearity of $\pm$3\%, a repeatability of $\pm$2.5\% and a hysteresis of $\pm$4.5\% of the full scale output \cite{b9}. When compared to similar low price sensors (e.g. interlink FSR), they have a shorter response time, a better linearity, and offer the possibility of registering larger forces \cite{b10}. Over the sensitive area of these sensors is added a semi-sphere to ease the force transfer.  


%==== END OF NEW PART


\begin{figure}[h] 
\centering
     \includegraphics[width=0.45\textwidth]{itsh}
    \caption{ITshoes schematic structure. The green block (A) is the acquisition unit, the red block (B) is the streaming unit, and the blue block (C) is the sensing unit. The main elements of these units are subtitled with numbers: 1--Battery; 2--Step-up voltage regulator; 3--Micro-controller; 4--Bi-directional level converter; 5--WiFi module; 6--Force sensor.   \label{fig.a6}}
\end{figure}

The acquisition unit (Fig. \ref{fig.a6} block A) is responsible for the data acquisition. This unit deals with the electrical conditioning (Fig. \ref{fig.a6}, number 3 and 4) and the power supply (Fig. \ref{fig.a6} number 1 and 2). The voltage dividers assembled in the PCB allow the connection of the sensors to the micro-controller. The ADC has a 10-bit resolution and \SI{625}{kHz} frequency to convert the force signals into digital information. To synchronize and adjust the operating voltage level of this unit, a logic level bi-directional converter is used. The entire system is powered by a small \SI{750}{mAh} Li-Po battery through a voltage regulator. \par

The streaming unit receives the data across a serial communication with the micro-controller and forwards it to the server, through the ESP8266 module (Fig. \ref{fig.a6}, number 5). The PCB added to the ITshoe incorporates a voltage regulator LM2937 to match the ESP8266 required power \SI{3.3}{V}.% This module sends a message to the AP host each time the controller sends an order.%How relevant it to mention this here?

\begin{figure}[h]
\centering
 \begin{tikzpicture}[scale = 0.9]
    \begin{axis}[
    title={\textbf{Sensor Calibration}},
    xlabel={Force [\si{N}]},
    ylabel={Conductance [\si{S}]},
    xmin=0,xmax=21.5,
    ymin=0,ymax=0.003,
    xtick={0,4.3,8.6,12.9,17.2,21.5},
    ]
    \addplot[blue,scatter,only marks,
        scatter src=explicit symbolic, 
    error bars/.cd,
        y dir=both,y explicit] 
        coordinates{
    (4.34,0.00049) += (0,0.00007787) -= (0,0.0000629)[1]     % [meta information, viewed as strings]
    (8.62,0.001033) += (0,0.00008369) -=(0,0.0001128)[2]
    (12.90,0.00156) += (0,0.00008483) -=(0,0.00008414)[3]
    (17.17,0.0021) += (0,0.0001211) -=(0,0.00007113)[4]
    };
    \addplot [thick, red] table[
    y={create col/linear regression}]
    {
        4.34434596268752 0.00049154
        8.62004537517456 0.00102817
        12.8957447876615 0.00156318
        17.1714442001486 0.00209651
    };
    \addlegendentry{y(x)}
    \addlegendentry{%
$5 \cdot 10^{-4} x - 4 \cdot 10^{-5}$}
\end{axis}
\end{tikzpicture}
\caption{Force sensor calibration line. }%ref}
\label{g1}
\end{figure}

\subsection{Sensor calibration }

To ensure more accurate readings, a calibration procedure upon the sensors is required. The calibration procedure starts by conditioning the sensors and applying 110\% (or slightly more) of the maximum test load onto the sensor for approximately 3 seconds. After repeating this conditioning process, different weights are applied on the sensor and, using a multimeter, the resistance is recorded. 
The plot in Fig. \ref{g1} shows the calibration line obtained for one of the \SI{4.44}{N} standard force sensors, where it is visible that there is a linear evolution between the force applied to the sensor and its conductance.

%==== THIS IS NEW: UNDER CONSIDERTION
Besides calibration, the sensors were also tested for their dynamic response. This concern occurs because the reaction forces that arise when a biped robot walks are variable, thus emerging the necessity to validate the dynamic response of the sensors. Hence, an experiment was conducted to analyze the sensors' dynamic response by using a mechanical testing machine Shimadzu MMT-101N. With this machine, a \SI{4}{N} force was applied to the sensor with 3 different velocities (\SIlist[list-units=single]{50;
25;12.5}{N/s}). \par

Fig. \ref{fig:flexiforcedynresponse} shows the sensor responses for the applied forces. For the lower velocity, the sensor maximum force detected exceeds 3\% the
applied force, while for the higher velocity the sensor is not able to reach the
maximum applied force (5.5\% error). Indeed, there is a slight delay or saturation of the dynamic response of the sensors for some velocities, but those deviations are not expected to compromise the suitability of the sensors to monitor robot locomotion. Actually, and as might have been expected due to the piezoelectric nature of the sensors, their response improves with higher velocities, thus seeming quite adequate to catch the finer, or higher frequency, transitory force reactions.
\begin{figure}[h!]
\centering
 \begin{tikzpicture}[scale=0.9]
    \begin{axis}[
    title={\textbf{Dynamic response}},
    xlabel={Time[s]},
    ylabel={Force[N]},
    xmin=0,xmax=0.6,
    ymin=0,ymax=5,
    xtick={0,0.1,0.2,0.3,0.4,0.5,0.6},
    ytick={0,1,2,3,4,5}
    ]
    \addplot [thick, red] table[
    y]
    {
0.01	0.1
0.011024	0.118181818
0.01204	0.109090909
0.013064	0.118181818
0.01408	0.127272727
0.015104	0.136363636
0.01612	0.127272727
0.017144	0.118181818
0.01816	0.127272727
0.019184	0.172727273
0.0202	0.172727273
0.021224	0.163636364
0.02224	0.163636364
0.023264	0.172727273
0.02428	0.2
0.025304	0.190909091
0.026324	0.190909091
0.027344	0.209090909
0.028364	0.227272727
0.029384	0.218181818
0.0304	0.227272727
0.031424	0.254545455
0.03244	0.290909091
0.033464	0.318181818
0.03448	0.318181818
0.035504	0.363636364
0.03652	0.345454545
0.037544	0.390909091
0.03856	0.418181818
0.039584	0.418181818
0.0406	0.454545455
0.041624	0.509090909
0.04264	0.527272727
0.043664	0.536363636
0.04468	0.645454545
0.045704	0.645454545
0.04672	0.690909091
0.047744	0.727272727
0.04876	0.736363636
0.049784	0.790909091
0.0508	0.809090909
0.051824	0.836363636
0.05284	0.863636364
0.053864	0.872727273
0.05488	0.909090909
0.055984	0.9
0.057008	0.918181818
0.058112	0.945454545
0.059216	0.954545455
0.06032	0.954545455
0.061424	0.963636364
0.062536	0.972727273
0.06364	0.954545455
0.064744	0.972727273
0.065856	1
0.066952	1.027272727
0.068056	1.045454545
0.069164	1.454545455
0.070264	1.518181818
0.071376	1.572727273
0.07248	1.618181818
0.073584	1.663636364
0.074688	1.654545455
0.075792	1.709090909
0.076896	1.772727273
0.078	1.836363636
0.079104	1.881818182
0.08022	1.9
0.08132	1.918181818
0.082424	1.954545455
0.083532	1.972727273
0.084632	2.009090909
0.085736	2.027272727
0.08684	2.072727273
0.087944	2.1
0.089056	2.127272727
0.09016	2.136363636
0.091264	2.154545455
0.092368	2.190909091
0.09348	2.209090909
0.094584	2.245454545
0.09568	2.272727273
0.096792	2.272727273
0.097896	2.3
0.099	2.336363636
0.100104	2.363636364
0.101208	2.418181818
0.102312	2.436363636
0.103416	2.463636364
0.10452	2.463636364
0.105624	2.463636364
0.106736	2.490909091
0.107848	2.527272727
0.108944	2.563636364
0.110048	2.572727273
0.111156	2.581818182
0.112256	2.6
0.113364	2.654545455
0.114464	2.663636364
0.115576	2.690909091
0.11668	2.7
0.117784	2.718181818
0.118888	2.745454545
0.119992	2.763636364
0.121096	2.772727273
0.122208	2.763636364
0.123304	2.781818182
0.124416	2.790909091
0.12552	2.827272727
0.126624	2.872727273
0.127728	2.9
0.128832	2.909090909
0.129936	2.918181818
0.13104	2.936363636
0.132144	2.972727273
0.133256	3.027272727
0.13436	3.054545455
0.135472	3.063636364
0.136576	3.072727273
0.137672	3.090909091
0.13878	3.109090909
0.13988	3.136363636
0.140984	3.154545455
0.142096	3.172727273
0.1432	3.2
0.144304	3.209090909
0.145408	3.236363636
0.146512	3.254545455
0.147616	3.272727273
0.14872	3.3
0.149832	3.309090909
0.150936	3.336363636
0.15204	3.345454545
0.153144	3.381818182
0.154248	3.390909091
0.155352	3.427272727
0.156456	3.427272727
0.15756	3.436363636
0.158664	3.454545455
0.159776	3.472727273
0.16088	3.490909091
0.161984	3.490909091
0.163096	3.490909091
0.1642	3.490909091
0.165296	3.536363636
0.166404	3.554545455
0.167512	3.581818182
0.168616	3.581818182
0.16972	3.590909091
0.170824	3.609090909
0.171928	3.609090909
0.173032	3.618181818
0.174136	3.636363636
0.17524	3.654545455
0.176344	3.7
0.17746	3.718181818
0.178568	3.736363636
0.179664	3.745454545
0.180772	3.754545455
0.181872	3.763636364
0.182976	3.781818182
0.18408	3.8
0.185184	3.827272727
0.186296	3.836363636
0.1874	3.845454545
0.188504	3.854545455
0.189608	3.863636364
0.190712	3.890909091
0.191824	3.890909091
0.19292	3.9
0.194024	3.918181818
0.195136	3.936363636
0.19624	3.945454545
0.197344	3.963636364
0.198448	3.981818182
0.199552	4
0.200656	4.009090909
0.20176	4.027272727
0.202864	4.027272727
0.203976	4.072727273
0.205088	4.109090909
0.206192	4.109090909
0.207288	4.118181818
0.208396	4.109090909
0.2095	4.118181818
0.2106	4.136363636
0.211704	4.145454545
0.212816	4.163636364
0.21392	4.172727273
0.215024	4.181818182
0.216128	4.181818182
0.217232	4.190909091
0.218336	4.181818182
0.219448	4.190909091
0.220552	4.190909091
0.221656	4.181818182
0.22276	4.181818182
0.223864	4.190909091
0.224972	4.181818182
0.226072	4.181818182
0.227176	4.190909091
0.22828	4.181818182
0.229384	4.181818182
0.230496	4.190909091
0.2316	4.181818182
0.232704	4.181818182
0.233816	4.181818182
0.234912	4.172727273
0.23602	4.172727273
0.237124	4.181818182
0.238224	4.172727273
0.239336	4.181818182
0.24044	4.190909091
0.241544	4.172727273
0.242648	4.181818182
0.243752	4.190909091
0.244856	4.2
0.24596	4.181818182
0.247072	4.190909091
0.24818	4.181818182
0.24928	4.190909091
0.250384	4.190909091
0.251488	4.190909091
0.252592	4.2
0.253696	4.2
0.2548	4.190909091
0.255904	4.2
0.257016	4.2
0.25812	4.2
0.259224	4.190909091
0.260328	4.2
0.26144	4.2
0.262536	4.2
0.26364	4.2
0.264752	4.2
0.265856	4.2
0.26696	4.190909091
0.268064	4.190909091
0.269168	4.2
0.270272	4.181818182
0.271376	4.181818182
0.27248	4.181818182
0.273584	4.181818182
0.274696	4.172727273
0.275808	4.154545455
0.276904	4.154545455
0.278012	4.154545455
0.279116	4.154545455
0.280216	4.136363636
0.28132	4.145454545
0.282424	4.136363636
0.283536	4.127272727
0.28464	4.118181818
0.285744	4.109090909
0.286848	4.1
0.287952	4.1
0.289064	4.090909091
0.290168	4.063636364
0.291264	4.063636364
0.292376	4.063636364
0.29348	4.054545455
0.294584	4.054545455
0.295688	4.045454545
0.296792	4.036363636
0.297896	4.027272727
0.299	4.018181818
0.300104	4.009090909
0.301216	3.990909091
0.30232	4
0.303432	4
0.304528	3.990909091
0.305632	3.981818182
0.30674	3.981818182
0.30784	3.963636364
0.308952	3.954545455
0.310056	3.945454545
0.31116	3.936363636
0.312264	3.945454545
0.313368	3.927272727
0.314472	3.909090909
0.315576	3.909090909
0.31668	3.9
0.317792	3.881818182
0.318896	3.881818182
0.32	3.881818182
0.321104	3.872727273
0.322208	3.863636364
0.323312	3.854545455
0.324416	3.845454545
0.32552	3.836363636
0.326624	3.827272727
0.327736	3.818181818
0.32884	3.809090909
0.329944	3.8
0.331056	3.790909091
0.33216	3.790909091
0.333256	3.772727273
0.334364	3.754545455
0.335464	3.745454545
0.336576	3.745454545
0.33768	3.736363636
0.338784	3.718181818
0.339888	3.7
0.340992	3.690909091
0.342096	3.690909091
0.3432	3.690909091
0.344304	3.681818182
0.34542	3.672727273
0.34652	3.663636364
0.347624	3.654545455
0.348732	3.636363636
0.349832	3.609090909
0.350936	3.609090909
0.35204	3.6
0.353144	3.581818182
0.354256	3.581818182
0.35536	3.545454545
0.356464	3.545454545
0.357568	3.536363636
0.35868	3.536363636
0.359784	3.518181818
0.36088	3.509090909
0.361992	3.472727273
0.363096	3.454545455
0.3642	3.427272727
0.365304	3.427272727
0.366408	3.418181818
0.367512	3.4
0.368616	3.390909091
0.36972	3.363636364
0.370824	3.372727273
0.371936	3.363636364
0.373048	3.345454545
0.374152	3.345454545
0.375248	3.327272727
0.376356	3.327272727
0.377456	3.318181818
0.37856	3.309090909
0.379664	3.3
0.380776	3.281818182
0.38188	3.245454545
0.382984	3.236363636
0.384088	3.227272727
0.385192	3.209090909
0.386296	3.190909091
0.387408	3.154545455
0.388504	3.127272727
0.389616	3.118181818
0.39072	3.1
0.391824	3.081818182
0.392928	3.045454545
0.394032	3.018181818
0.395136	3
0.39624	3
0.397344	3
0.398456	2.981818182
0.39956	2.972727273
0.400672	2.918181818
0.401776	2.890909091
0.402872	2.881818182
0.40398	2.881818182
0.405084	2.854545455
0.406184	2.818181818
0.407296	2.790909091
0.4084	2.772727273
0.409504	2.754545455
0.410608	2.745454545
0.411712	2.727272727
0.412816	2.709090909
0.41392	2.663636364
0.415032	2.645454545
0.416136	2.627272727
0.41724	2.609090909
0.418344	2.581818182
0.419448	2.581818182
0.420556	2.581818182
0.421656	2.572727273
0.42276	2.536363636
0.423864	2.527272727
0.424976	2.518181818
0.42608	2.5
0.427184	2.472727273
0.428288	2.445454545
0.4294	2.445454545
0.430496	2.436363636
0.431604	2.390909091
0.432712	2.372727273
0.433816	2.345454545
0.43492	2.336363636
0.436024	2.318181818
0.437128	2.290909091
0.438232	2.272727273
0.439336	2.263636364
0.44044	2.254545455
0.441544	2.209090909
0.44266	2.163636364
0.443768	2.118181818
0.444864	2.090909091
0.445972	2.072727273
0.447072	2.063636364
0.448176	2.036363636
0.44928	2.018181818
0.450384	1.990909091
0.451496	1.972727273
0.4526	1.927272727
0.453704	1.909090909
0.454808	1.890909091
0.455912	1.872727273
0.457024	1.818181818
0.45812	1.772727273
0.459224	1.709090909
0.460336	1.7
0.46144	1.663636364
0.462544	1.645454545
0.463648	1.563636364
0.464752	1.536363636
0.465856	1.490909091
0.46696	1.454545455
0.468064	1.418181818
0.469176	1.409090909
0.47028	1.390909091
0.471392	1.345454545
0.472488	1.318181818
0.473596	1.281818182
0.4747	1.254545455
0.4758	1.245454545
0.476904	1.236363636
0.478016	1.236363636
0.47912	1.227272727
0.480224	1.2
0.481328	1.190909091
0.482432	1.163636364
0.483536	1.145454545
0.484648	1.118181818
0.485752	1.090909091
0.486856	1.063636364
0.48796	1.063636364
0.489064	1.036363636
0.490168	0.990909091
0.491272	0.981818182
0.492376	0.972727273
0.49348	0.945454545
0.494584	0.918181818
0.495696	0.890909091
0.496712	0.845454545
0.497736	0.836363636
0.498752	0.8
0.499776	0.772727273
0.500792	0.736363636
0.501816	0.709090909
0.502832	0.672727273
0.503856	0.663636364
0.504872	0.663636364
0.505896	0.654545455
0.506912	0.618181818
0.507936	0.618181818
0.508952	0.609090909
0.509976	0.554545455
0.510992	0.518181818
0.512016	0.481818182
0.513032	0.472727273
0.514056	0.445454545
0.515072	0.418181818
0.516096	0.418181818
0.517112	0.4
0.518136	0.427272727
0.519152	0.427272727
0.520176	0.381818182
0.521192	0.336363636
0.522216	0.318181818
0.523232	0.3
0.524256	0.263636364
0.525272	0.272727273
0.526296	0.272727273
0.527312	0.218181818
0.528336	0.2
0.529352	0.190909091
0.530376	0.163636364
0.531392	0.136363636
0.532416	0.127272727
0.533432	0.154545455
0.534456	0.145454545
0.535472	0.136363636
0.536496	0.145454545
0.537512	0.163636364
0.538536	0.136363636
0.539552	0.127272727
0.540576	0.118181818
0.541592	0.109090909
0.542616	0.109090909
0.543632	0.109090909
0.544656	0.1
0.545672	0.109090909
0.546696	0.1
0.547712	0.1
0.548736	0.1
0.549752	0.1
0.550776	0.090909091



     };
      \addplot [thick, blue, dashed] table[
    y]
    {
  
0	0.176568002
1.00E-03	0.17791
2.00E-03	0.175753996
3.00E-03	0.173638999
4.00E-03	0.176844999
5.00E-03	0.173830003
6.00E-03	0.178377002
7.00E-03	0.174999997
8.00E-03	0.173703
9.00E-03	0.171944007
1.00E-02	0.176970005
0.011000001	0.175646007
1.20E-02	0.172891006
1.30E-02	0.170219004
1.40E-02	0.170321003
1.50E-02	0.174559996
1.60E-02	0.177009001
1.70E-02	0.181586996
1.80E-02	0.188692003
1.90E-02	0.186507002
0.020000001	0.184562996
2.10E-02	0.187941998
2.20E-02	0.191864997
2.30E-02	0.201874003
2.40E-02	0.212592006
2.50E-02	0.218365997
2.60E-02	0.229040995
2.70E-02	0.233517006
2.80E-02	0.239748001
2.90E-02	0.251011997
3.00E-02	0.259451002
0.031000001	0.260675013
3.20E-02	0.264360011
3.30E-02	0.277269989
3.40E-02	0.289696008
3.50E-02	0.299852014
3.60E-02	0.31042701
3.70E-02	0.320816994
3.80E-02	0.327874005
3.90E-02	0.34198001
4.00E-02	0.35066399
4.10E-02	0.363189995
0.042000002	0.378980994
4.30E-02	0.395978004
4.40E-02	0.409649998
4.50E-02	0.414734006
4.60E-02	0.421175003
4.70E-02	0.431793004
4.80E-02	0.448006988
4.90E-02	0.468331993
5.00E-02	0.489003986
0.051000002	0.513604999
5.20E-02	0.542035997
5.30E-02	0.558625996
5.40E-02	0.578013003
5.50E-02	0.60057497
5.60E-02	0.624472022
5.70E-02	0.642054975
5.80E-02	0.658097982
5.90E-02	0.68403101
6.00E-02	0.713045001
6.10E-02	0.740257025
0.062000003	0.772516012
6.30E-02	0.798547029
6.40E-02	0.818413019
6.50E-02	0.834479988
6.60E-02	0.841606975
6.70E-02	0.84867698
6.80E-02	0.870644987
6.90E-02	0.893868983
7.00E-02	0.925440013
0.071000003	0.960565984
7.20E-02	0.985575974
7.30E-02	1.004238009
7.40E-02	1.018918037
7.50E-02	1.027917981
7.60E-02	1.033987045
7.70E-02	1.046607018
7.80E-02	1.06558001
7.90E-02	1.090708017
8.00E-02	1.11907804
8.10E-02	1.146505952
0.082000004	1.171200991
8.30E-02	1.192685008
8.40E-02	1.210618019
8.50E-02	1.22002399
8.60E-02	1.226771951
8.70E-02	1.236286044
8.80E-02	1.25475502
8.90E-02	1.28618896
9.00E-02	1.326279044
9.10E-02	1.350500941
9.20E-02	1.367722034
0.093000004	1.380825043
9.40E-02	1.387112975
9.50E-02	1.396932006
9.60E-02	1.41671896
9.70E-02	1.442749023
9.80E-02	1.466577053
9.90E-02	1.489094973
0.100000005	1.507433057
0.101000005	1.522964954
0.102000005	1.537809014
0.103000005	1.544545054
0.104000005	1.565369964
0.105000005	1.599110961
0.106000005	1.628476024
0.107000005	1.648493052
0.108000005	1.659693003
0.109000005	1.665812969
0.110000005	1.688151002
0.111000005	1.716091037
0.112000005	1.748227954
0.113000005	1.782953978
0.114000005	1.80793798
0.115000005	1.819383025
0.116000006	1.826200008
0.117000006	1.83137095
0.118000006	1.857854009
0.119000006	1.899517059
0.120000006	1.930585027
0.121000006	1.953753948
0.122000006	1.963907957
0.123000006	1.97024405
0.124000006	1.988170028
0.125000006	1.999958038
0.126000006	2.020462036
0.127000006	2.043508053
0.128000006	2.067670107
0.129000006	2.091613054
0.130000006	2.116081953
0.131000006	2.138607025
0.132000006	2.158406973
0.133000006	2.170577049
0.134000006	2.180224895
0.135000006	2.188431978
0.136000006	2.208194971
0.137000007	2.247028112
0.138000007	2.292042971
0.139000007	2.325385094
0.140000007	2.335630894
0.141000007	2.336982012
0.142000007	2.338027
0.143000007	2.348756075
0.144000007	2.379323006
0.145000007	2.412803888
0.146000007	2.442051888
0.147000007	2.453886986
0.148000007	2.46350503
0.149000007	2.476041079
0.150000007	2.500590086
0.151000007	2.524874926
0.152000007	2.55347991
0.153000007	2.57474494
0.154000007	2.587359905
0.155000007	2.602298975
0.156000007	2.624942064
0.157000007	2.65188098
0.158000008	2.673007965
0.159000008	2.691087008
0.160000008	2.705287933
0.161000008	2.726866007
0.162000008	2.742913008
0.163000008	2.754518986
0.164000008	2.772568941
0.165000008	2.803164005
0.166000008	2.838404894
0.167000008	2.867628098
0.168000008	2.887314081
0.169000008	2.895761967
0.170000008	2.912307978
0.171000008	2.938997984
0.172000008	2.968600988
0.173000008	2.98596096
0.174000008	2.987575054
0.175000008	2.996795893
0.176000008	3.018136024
0.177000008	3.05331111
0.178000008	3.085052967
0.179000009	3.117835999
0.180000009	3.131637096
0.181000009	3.139193058
0.182000009	3.153480053
0.183000009	3.172725916
0.184000009	3.19016695
0.185000009	3.202996969
0.186000009	3.226511002
0.187000009	3.260025978
0.188000009	3.299952984
0.189000009	3.338355064
0.190000009	3.354830027
0.191000009	3.353833914
0.192000009	3.358108044
0.193000009	3.376379967
0.194000009	3.398819923
0.195000009	3.425477028
0.196000009	3.448188066
0.197000009	3.479667902
0.198000009	3.506855011
0.199000009	3.524945974
0.200000009	3.532849073
0.20100001	3.55162096
0.20200001	3.577693939
0.20300001	3.602410078
0.20400001	3.623007059
0.20500001	3.642641068
0.20600001	3.665537119
0.20700001	3.687252998
0.20800001	3.708859921
0.20900001	3.728395939
0.21000001	3.754327059
0.21100001	3.784271955
0.21200001	3.80131793
0.21300001	3.802417994
0.21400001	3.818397045
0.21500001	3.857517004
0.21600001	3.906898975
0.21700001	3.932399035
0.21800001	3.938815117
0.21900001	3.942805052
0.22000001	3.949491024
0.22100001	3.976212978
0.222000011	4.012431145
0.223000011	4.041730881
0.224000011	4.060742855
0.225000011	4.074789047
0.226000011	4.081571102
0.227000011	4.084154129
0.228000011	4.082054138
0.229000011	4.083272934
0.230000011	4.08620882
0.231000011	4.08116293
0.232000011	4.078575134
0.233000011	4.076457024
0.234000011	4.075436115
0.235000011	4.076191902
0.236000011	4.072165012
0.237000011	4.0702672
0.238000011	4.073840141
0.239000011	4.075181007
0.240000011	4.06791687
0.241000011	4.066556931
0.242000011	4.066003799
0.243000012	4.064992905
0.244000012	4.064294815
0.245000012	4.061864853
0.246000012	4.063722134
0.247000012	4.06470108
0.248000012	4.060593128
0.249000012	4.05853796
0.250000012	4.060061932
0.251000012	4.061488152
0.252000012	4.057746887
0.253000012	4.052382946
0.254000012	4.047996044
0.255000012	4.056255817
0.256000012	4.052744865
0.257000012	4.049200058
0.258000012	4.042921066
0.259000012	4.045022011
0.260000012	4.047529221
0.261000012	4.04674387
0.262000012	4.045462132
0.263000012	4.044170856
0.264000013	4.035353184
0.265000013	4.03514719
0.266000013	4.030296803
0.267000013	4.03267622
0.268000013	4.029252052
0.269000013	4.033600807
0.270000013	4.027166843
0.271000013	4.027990818
0.272000013	4.027925968
0.273000013	4.029717922
0.274000013	4.022621155
0.275000013	4.017654896
0.276000013	4.015841961
0.277000013	4.014482021
0.278000013	4.004809856
0.279000013	4.000338078
0.280000013	3.993686914
0.281000013	3.9944489
0.282000013	3.978584051
0.283000013	3.974282026
0.284000013	3.969753981
0.285000014	3.970551968
0.286000014	3.951385975
0.287000014	3.932198048
0.288000014	3.914494038
0.289000014	3.911832094
0.290000014	3.908452988
0.291000014	3.896989107
0.292000014	3.88586092
0.293000014	3.872714996
0.294000014	3.858906984
0.295000014	3.844805956
0.296000014	3.836680889
0.297000014	3.828665972
0.298000014	3.813210964
0.299000014	3.803555965
0.300000014	3.792507887
0.301000014	3.779170036
0.302000014	3.761471033
0.303000014	3.749202013
0.304000014	3.745182037
0.305000014	3.742650032
0.306000015	3.728796005
0.307000015	3.707977057
0.308000015	3.691572905
0.309000015	3.667606115
0.310000015	3.655750036
0.311000015	3.643050909
0.312000015	3.629101992
0.313000015	3.619874001
0.314000015	3.618388891
0.315000015	3.615155935
0.316000015	3.599312067
0.317000015	3.580935001
0.318000015	3.56595397
0.319000015	3.552412987
0.320000015	3.53393507
0.321000015	3.516875982
0.322000015	3.509788036
0.323000015	3.501394033
0.324000015	3.492727995
0.325000015	3.471590996
0.326000015	3.448307037
0.327000016	3.432471991
0.328000016	3.418381929
0.329000016	3.404011011
0.330000016	3.400631905
0.331000016	3.392071962
0.332000016	3.375344992
0.333000016	3.360884905
0.334000016	3.35086894
0.335000016	3.334506035
0.336000016	3.316200972
0.337000016	3.304335117
0.338000016	3.285974979
0.339000016	3.271141052
0.340000016	3.255218983
0.341000016	3.247087002
0.342000016	3.23625803
0.343000016	3.221411943
0.344000016	3.207700014
0.345000016	3.193252087
0.346000016	3.173727989
0.347000016	3.167133093
0.348000017	3.156291008
0.349000017	3.137387991
0.350000017	3.119668007
0.351000017	3.096651077
0.352000017	3.084114075
0.353000017	3.073462963
0.354000017	3.059335947
0.355000017	3.053934097
0.356000017	3.038609028
0.357000017	3.027494907
0.358000017	3.010015965
0.359000017	2.989213943
0.360000017	2.966470003
0.361000017	2.953524113
0.362000017	2.94818902
0.363000017	2.938682079
0.364000017	2.923043966
0.365000017	2.90835309
0.366000017	2.88678503
0.367000017	2.863487959
0.368000017	2.846065044
0.369000018	2.835427999
0.370000018	2.833592892
0.371000018	2.822169065
0.372000018	2.793751955
0.373000018	2.76087904
0.374000018	2.743014097
0.375000018	2.737114906
0.376000018	2.730434895
0.377000018	2.714863062
0.378000018	2.694822073
0.379000018	2.67720294
0.380000018	2.662998915
0.381000018	2.65177393
0.382000018	2.6370399
0.383000018	2.625298023
0.384000018	2.606446028
0.385000018	2.592735052
0.386000018	2.578325987
0.387000018	2.562580109
0.388000018	2.541637897
0.389000018	2.524178028
0.390000019	2.51782608
0.391000019	2.506020069
0.392000019	2.482788086
0.393000019	2.456531048
0.394000019	2.440844059
0.395000019	2.435465097
0.396000019	2.423440933
0.397000019	2.402307034
0.398000019	2.382206917
0.399000019	2.365145922
0.400000019	2.341618061
0.401000019	2.335053921
0.402000019	2.324387074
0.403000019	2.310412884
0.404000019	2.290364981
0.405000019	2.259347916
0.406000019	2.240403891
0.407000019	2.234405994
0.408000019	2.227634907
0.409000019	2.221297979
0.410000019	2.209918022
0.41100002	2.176306963
0.41200002	2.142533064
0.41300002	2.121156931
0.41400002	2.116724014
0.41500002	2.108999968
0.41600002	2.091878891
0.41700002	2.072467089
0.41800002	2.050421
0.41900002	2.033682108
0.42000002	2.016836882
0.42100002	2.009548903
0.42200002	2.001318932
0.42300002	1.987627029
0.42400002	1.965620041
0.42500002	1.939929962
0.42600002	1.921442986
0.42700002	1.895264983
0.42800002	1.875311017
0.42900002	1.856753945
0.43000002	1.846547008
0.43100002	1.839517951
0.432000021	1.83178699
0.433000021	1.819888949
0.434000021	1.796488047
0.435000021	1.769965053
0.436000021	1.744132042
0.437000021	1.720785975
0.438000021	1.707937002
0.439000021	1.698236942
0.440000021	1.685745001
0.441000021	1.677651048
0.442000021	1.663159013
0.443000021	1.634461045
0.444000021	1.597707033
0.445000021	1.581323028
0.446000021	1.578933001
0.447000021	1.569051981
0.448000021	1.552263975
0.449000021	1.532518983
0.450000021	1.516780019
0.451000021	1.502578974
0.452000021	1.480221987
0.453000022	1.454012036
0.454000022	1.425042987
0.455000022	1.403604984
0.456000022	1.392035007
0.457000022	1.384297013
0.458000022	1.379062057
0.459000022	1.364711046
0.460000022	1.342579007
0.461000022	1.32169497
0.462000022	1.303614974
0.463000022	1.281491995
0.464000022	1.256716967
0.465000022	1.235968947
0.466000022	1.22211802
0.467000022	1.211822033
0.468000022	1.189393997
0.469000022	1.169513941
0.470000022	1.157742023
0.471000022	1.142531991
0.472000022	1.132328987
0.473000022	1.119346976
0.474000023	1.095201015
0.475000023	1.065467954
0.476000023	1.041713953
0.477000023	1.011265993
0.478000023	0.992350996
0.479000023	0.984983027
0.480000023	0.977602005
0.481000023	0.975395977
0.482000023	0.961429
0.483000023	0.933827996
0.484000023	0.905901015
0.485000023	0.875470996
0.486000023	0.851786971
0.487000023	0.842872977
0.488000023	0.834254026
0.489000023	0.819589019
0.490000023	0.807896018
0.491000023	0.788993001
0.492000023	0.77362299
0.493000023	0.756725013
0.494000023	0.737321973
0.495000024	0.716108024
0.496000024	0.691084027
0.497000024	0.667259991
0.498000024	0.651372015
0.499000024	0.634088993
0.500000024	0.618646979
0.501000024	0.613483012
0.502000024	0.597675025
0.503000024	0.584106028
0.504000024	0.564486027
0.505000024	0.543839991
0.506000024	0.521893978
0.507000024	0.502323985
0.508000024	0.483653009
0.509000024	0.470490009
0.510000024	0.460678011
0.511000024	0.446743995
0.512000024	0.433277011
0.513000024	0.421851009
0.514000024	0.404870003
0.515000024	0.376552999
0.516000025	0.360574007
0.517000025	0.349480987
0.518000025	0.332702011
0.519000025	0.327250004
0.520000025	0.308824003
0.521000025	0.289209992
0.522000025	0.277114004
0.523000025	0.265415996
0.524000025	0.251486987
0.525000025	0.238431007
0.526000025	0.225213006
0.527000025	0.211052001
0.528000025	0.195017993
0.529000025	0.178547993
0.530000025	0.169038996
0.531000025	0.160376996
0.532000025	0.146498993
0.533000025	0.131707996
0.534000025	0.123634003
0.535000025	0.112079002
0.536000025	0.106427997
0.537000026	0.101223998
0.538000026	0.095922001
0.539000026	0.084141001
0.540000026	0.071511


     };
       \addplot [thick, red] table[
    y]
    {
     0.01	0.1
0.011024	0.1
0.01204	0.109090909
0.013064	0.118181818
0.01408	0.109090909
0.015104	0.1
0.01612	0.1
0.017144	0.1
0.01816	0.118181818
0.019184	0.118181818
0.0202	0.136363636
0.021224	0.145454545
0.02224	0.181818182
0.023264	0.209090909
0.02428	0.245454545
0.025304	0.290909091
0.02632	0.3
0.027344	0.354545455
0.02836	0.418181818
0.029384	0.472727273
0.0304	0.490909091
0.031424	0.536363636
0.03244	0.672727273
0.033464	0.754545455
0.03448	0.790909091
0.035504	0.909090909
0.036608	0.927272727
0.03772	1.072727273
0.038816	1.072727273
0.03992	1.136363636
0.041032	1.218181818
0.042136	1.281818182
0.04324	1.327272727
0.044344	1.436363636
0.045448	1.509090909
0.046552	1.636363636
0.047656	1.718181818
0.04876	1.781818182
0.049864	1.845454545
0.050976	1.881818182
0.052088	1.963636364
0.053184	2.027272727
0.054292	2.109090909
0.055396	2.181818182
0.056496	2.227272727
0.0576	2.245454545
0.058704	2.263636364
0.059816	2.318181818
0.06092	2.390909091
0.062024	2.436363636
0.063128	2.481818182
0.064232	2.490909091
0.065344	2.490909091
0.066448	2.545454545
0.067544	2.609090909
0.068656	2.672727273
0.06976	2.690909091
0.070864	2.718181818
0.071968	2.736363636
0.073072	2.790909091
0.074176	2.872727273
0.07528	2.918181818
0.076384	2.936363636
0.077496	2.954545455
0.0786	2.981818182
0.079712	3.027272727
0.080808	3.081818182
0.081912	3.136363636
0.08302	3.172727273
0.08412	3.181818182
0.085232	3.190909091
0.086336	3.218181818
0.08744	3.245454545
0.088544	3.290909091
0.089648	3.336363636
0.090752	3.345454545
0.091856	3.363636364
0.09296	3.372727273
0.094072	3.436363636
0.095176	3.490909091
0.09628	3.527272727
0.097384	3.536363636
0.098488	3.536363636
0.099592	3.554545455
0.100696	3.590909091
0.1018	3.645454545
0.102904	3.690909091
0.104016	3.718181818
0.10512	3.718181818
0.106224	3.736363636
0.107336	3.781818182
0.10844	3.827272727
0.109536	3.854545455
0.110644	3.872727273
0.111744	3.890909091
0.112856	3.9
0.11396	3.918181818
0.115064	3.963636364
0.116168	4
0.117272	4.027272727
0.118376	4.045454545
0.11948	4.063636364
0.120584	4.081818182
0.1217	4.118181818
0.1228	4.145454545
0.123904	4.163636364
0.125012	4.172727273
0.126112	4.172727273
0.127216	4.181818182
0.12832	4.181818182
0.129424	4.172727273
0.130536	4.172727273
0.13164	4.172727273
0.132744	4.172727273
0.133848	4.163636364
0.13496	4.154545455
0.136064	4.163636364
0.13716	4.163636364
0.138272	4.163636364
0.139376	4.172727273
0.14048	4.172727273
0.141584	4.172727273
0.142688	4.172727273
0.143792	4.172727273
0.144896	4.172727273
0.146	4.163636364
0.147104	4.172727273
0.148216	4.172727273
0.149328	4.154545455
0.150432	4.136363636
0.151528	4.127272727
0.152636	4.118181818
0.153736	4.109090909
0.15484	4.1
0.155944	4.090909091
0.157056	4.090909091
0.15816	4.072727273
0.159264	4.054545455
0.160368	4.045454545
0.161472	4.036363636
0.162576	4.027272727
0.163688	4.009090909
0.164784	3.990909091
0.165896	3.990909091
0.167	3.972727273
0.168104	3.945454545
0.169208	3.936363636
0.170312	3.927272727
0.171416	3.909090909
0.17252	3.890909091
0.173624	3.863636364
0.174736	3.827272727
0.17584	3.818181818
0.176952	3.809090909
0.178056	3.8
0.179152	3.781818182
0.18026	3.745454545
0.181364	3.690909091
0.182464	3.681818182
0.183576	3.672727273
0.18468	3.663636364
0.185784	3.636363636
0.186888	3.6
0.187992	3.563636364
0.189096	3.536363636
0.1902	3.536363636
0.191312	3.536363636
0.192416	3.509090909
0.19352	3.472727273
0.194624	3.445454545
0.195728	3.427272727
0.196836	3.409090909
0.197936	3.4
0.19904	3.372727273
0.200144	3.336363636
0.201256	3.336363636
0.20236	3.318181818
0.203464	3.3
0.204568	3.272727273
0.20568	3.245454545
0.206776	3.236363636
0.207884	3.209090909
0.208992	3.172727273
0.210096	3.136363636
0.2112	3.118181818
0.212304	3.090909091
0.213408	3.072727273
0.214512	3.018181818
0.215616	2.945454545
0.21672	2.927272727
0.217824	2.918181818
0.21894	2.9
0.220048	2.836363636
0.221144	2.754545455
0.222252	2.709090909
0.223352	2.672727273
0.224456	2.663636364
0.22556	2.663636364
0.226664	2.609090909
0.227776	2.536363636
0.22888	2.472727273
0.229984	2.463636364
0.231088	2.445454545
0.232192	2.427272727
0.233304	2.409090909
0.2344	2.372727273
0.235504	2.327272727
0.236616	2.281818182
0.23772	2.245454545
0.238824	2.236363636
0.239928	2.227272727
0.241032	2.2
0.242136	2.118181818
0.24324	2
0.244344	1.963636364
0.245456	1.890909091
0.24656	1.854545455
0.247672	1.818181818
0.248768	1.772727273
0.249876	1.727272727
0.25098	1.636363636
0.25208	1.536363636
0.253184	1.472727273
0.254296	1.454545455
0.2554	1.436363636
0.256504	1.418181818
0.257608	1.354545455
0.258712	1.236363636
0.259816	1.218181818
0.260928	1.172727273
0.262032	1.163636364
0.263136	1.154545455
0.26424	1.072727273
0.265344	1.045454545
0.266448	1.009090909
0.267552	0.954545455
0.268656	0.936363636
0.26976	0.9
0.270784	0.890909091
0.2718	0.854545455
0.272824	0.8
0.27384	0.772727273
0.274864	0.645454545
0.27588	0.609090909
0.276904	0.554545455
0.27792	0.536363636
0.278944	0.518181818
0.27996	0.427272727
0.280984	0.381818182
0.282	0.381818182
0.283024	0.372727273
0.28404	0.354545455
0.285064	0.372727273
0.28608	0.363636364
0.287104	0.336363636
0.28812	0.354545455
0.289144	0.3
0.29016	0.281818182
0.291184	0.272727273
0.2922	0.254545455
0.293224	0.254545455
0.29424	0.272727273
0.295264	0.254545455
0.29628	0.218181818
0.297304	0.145454545
0.29832	0.145454545
0.299344	0.172727273
0.30036	0.154545455
0.301384	0.136363636
0.3024	0.118181818
0.303424	0.145454545
0.30444	0.136363636
0.305464	0.136363636
0.30648	0.118181818
0.307504	0.109090909
0.30852	0.109090909
0.309544	0.1
0.31056	0.090909091
0.311584	0.090909091
0.3126	0.081818182
0.313536	0.1
0.31456	0.072727273
0.315496	0.081818182
0.316432	0.090909091
0.317452	0.1
0.318464	0.090909091
0.319488	0.090909091
0.320504	0.081818182
0.32144	0.081818182
0.322384	0.072727273
0.323312	0.090909091
0.324336	0.090909091
0.325352	0.081818182
0.326288	0.081818182
0.327224	0.081818182
0.32816	0.081818182
0.329096	0.081818182
0.330024	0.081818182
0.33096	0.090909091
0.331984	0.1
0.333	0.090909091
0.334024	0.1
0.33504	0.090909091

};

  \addplot [thick, blue, dashed] table[
    y]
    {
    0	0.174965993
1.00E-03	0.178804994
2.00E-03	0.179010004
3.00E-03	0.175623998
4.00E-03	0.173051998
5.00E-03	0.176982
6.00E-03	0.180311993
7.00E-03	0.181565002
8.00E-03	0.176445007
9.00E-03	0.178402007
1.00E-02	0.178370997
0.011000001	0.174795002
1.20E-02	0.17881
1.30E-02	0.180299997
1.40E-02	0.187325001
1.50E-02	0.192569003
1.60E-02	0.203105003
1.70E-02	0.214948997
1.80E-02	0.222369999
1.90E-02	0.237978995
0.020000001	0.251960993
2.10E-02	0.265962988
2.20E-02	0.279444993
2.30E-02	0.295807987
2.40E-02	0.32000801
2.50E-02	0.351274997
2.60E-02	0.371147007
2.70E-02	0.395105004
2.80E-02	0.41812399
2.90E-02	0.441873997
3.00E-02	0.458243996
0.031000001	0.478289008
3.20E-02	0.512037992
3.30E-02	0.558438003
3.40E-02	0.611835003
3.50E-02	0.657522976
3.60E-02	0.705434978
3.70E-02	0.763881028
3.80E-02	0.814256012
3.90E-02	0.861437023
4.00E-02	0.901106
4.10E-02	0.946439028
0.042000002	0.992932975
4.30E-02	1.041854024
4.40E-02	1.084301949
4.50E-02	1.122869015
4.60E-02	1.159494042
4.70E-02	1.194708943
4.80E-02	1.240442038
4.90E-02	1.28590095
5.00E-02	1.336472034
0.051000002	1.387786031
5.20E-02	1.422029972
5.30E-02	1.445080042
5.40E-02	1.46301496
5.50E-02	1.483767033
5.60E-02	1.516224027
5.70E-02	1.561463952
5.80E-02	1.621376038
5.90E-02	1.676591992
6.00E-02	1.704357982
6.10E-02	1.721413016
0.062000003	1.74209702
6.30E-02	1.77581501
6.40E-02	1.820201993
6.50E-02	1.873939991
6.60E-02	1.913231969
6.70E-02	1.941424012
6.80E-02	1.953178048
6.90E-02	1.977743983
7.00E-02	2.023921967
0.071000003	2.079987049
7.20E-02	2.130961895
7.30E-02	2.152858019
7.40E-02	2.16635108
7.50E-02	2.186674118
7.60E-02	2.225364923
7.70E-02	2.282206059
7.80E-02	2.342091084
7.90E-02	2.385648966
8.00E-02	2.406640053
8.10E-02	2.415997982
0.082000004	2.436357975
8.30E-02	2.475931883
8.40E-02	2.523401022
8.50E-02	2.577054977
8.60E-02	2.616622925
8.70E-02	2.647191048
8.80E-02	2.664262056
8.90E-02	2.692466021
9.00E-02	2.735351086
9.10E-02	2.792248964
9.20E-02	2.856035948
0.093000004	2.898904085
9.40E-02	2.913616896
9.50E-02	2.914776087
9.60E-02	2.935409069
9.70E-02	2.981895924
9.80E-02	3.040999889
9.90E-02	3.104413986
0.100000005	3.152417898
0.101000005	3.16516304
0.102000005	3.16877389
0.103000005	3.189536095
0.104000005	3.237375021
0.105000005	3.303786039
0.106000005	3.361809015
0.107000005	3.399066925
0.108000005	3.419929028
0.109000005	3.429234028
0.110000005	3.458909988
0.111000005	3.509226084
0.112000005	3.576847076
0.113000005	3.62518692
0.114000005	3.646194935
0.115000005	3.663835049
0.116000006	3.691343069
0.117000006	3.717869997
0.118000006	3.76551795
0.119000006	3.824012041
0.120000006	3.879959106
0.121000006	3.915862083
0.122000006	3.939342022
0.123000006	3.964529037
0.124000006	3.996330023
0.125000006	4.042925835
0.126000006	4.088387966
0.127000006	4.123037815
0.128000006	4.145789146
0.129000006	4.154273033
0.130000006	4.152852058
0.131000006	4.147152901
0.132000006	4.143918037
0.133000006	4.141702175
0.134000006	4.141783237
0.135000006	4.137386799
0.136000006	4.133883953
0.137000007	4.139293194
0.138000007	4.132524014
0.139000007	4.127342224
0.140000007	4.129033089
0.141000007	4.122837067
0.142000007	4.121077061
0.143000007	4.117085934
0.144000007	4.113282204
0.145000007	4.115606785
0.146000007	4.114596844
0.147000007	4.10806179
0.148000007	4.101728916
0.149000007	4.093369007
0.150000007	4.087317944
0.151000007	4.081645966
0.152000007	4.066899776
0.153000007	4.038243771
0.154000007	4.004401207
0.155000007	3.981415987
0.156000007	3.96608305
0.157000007	3.947314024
0.158000008	3.920557022
0.159000008	3.898926973
0.160000008	3.88299799
0.161000008	3.863559008
0.162000008	3.825835943
0.163000008	3.782032013
0.164000008	3.751621008
0.165000008	3.736294031
0.166000008	3.729782104
0.167000008	3.717026949
0.168000008	3.684700012
0.169000008	3.653078079
0.170000008	3.616106033
0.171000008	3.586586952
0.172000008	3.569205999
0.173000008	3.549962997
0.174000008	3.527087927
0.175000008	3.497467041
0.176000008	3.465903044
0.177000008	3.425584078
0.178000008	3.392007113
0.179000009	3.375201941
0.180000009	3.366446018
0.181000009	3.35324502
0.182000009	3.318516016
0.183000009	3.270463943
0.184000009	3.225958109
0.185000009	3.200553894
0.186000009	3.185769081
0.187000009	3.17275095
0.188000009	3.157960892
0.189000009	3.122659922
0.190000009	3.071595907
0.191000009	3.022605896
0.192000009	2.988446951
0.193000009	2.981760025
0.194000009	2.975037098
0.195000009	2.962264061
0.196000009	2.937618017
0.197000009	2.892659903
0.198000009	2.851181984
0.199000009	2.814479113
0.200000009	2.789681911
0.20100001	2.773128033
0.20200001	2.747643948
0.20300001	2.707916975
0.20400001	2.670495033
0.20500001	2.647305965
0.20600001	2.62558198
0.20700001	2.599109888
0.20800001	2.579161882
0.20900001	2.546238899
0.21000001	2.513386965
0.21100001	2.473117113
0.21200001	2.438045979
0.21300001	2.405132055
0.21400001	2.383920908
0.21500001	2.365066051
0.21600001	2.343703985
0.21700001	2.30833292
0.21800001	2.261488914
0.21900001	2.222382069
0.22000001	2.206082106
0.22100001	2.199101925
0.222000011	2.178973913
0.223000011	2.144042015
0.224000011	2.084117889
0.225000011	2.025445938
0.226000011	1.984418988
0.227000011	1.968994975
0.228000011	1.969203949
0.229000011	1.959036946
0.230000011	1.933696032
0.231000011	1.891052008
0.232000011	1.840476036
0.233000011	1.798329949
0.234000011	1.764565945
0.235000011	1.740478992
0.236000011	1.727329016
0.237000011	1.711572051
0.238000011	1.67127502
0.239000011	1.625184059
0.240000011	1.578001022
0.241000011	1.547988057
0.242000011	1.536013007
0.243000012	1.522457004
0.244000012	1.497079015
0.245000012	1.457206964
0.246000012	1.411782026
0.247000012	1.369508028
0.248000012	1.334010959
0.249000012	1.310243964
0.250000012	1.29389596
0.251000012	1.273450017
0.252000012	1.241225958
0.253000012	1.198403001
0.254000012	1.156983972
0.255000012	1.118075967
0.256000012	1.093508005
0.257000012	1.075628042
0.258000012	1.058814049
0.259000012	1.029484034
0.260000012	0.990379989
0.261000012	0.941923976
0.262000012	0.896991014
0.263000012	0.859596014
0.264000013	0.838280022
0.265000013	0.830720007
0.266000013	0.809598982
0.267000013	0.781979024
0.268000013	0.746237993
0.269000013	0.708217025
0.270000013	0.662768006
0.271000013	0.626789987
0.272000013	0.60136801
0.273000013	0.587046027
0.274000013	0.570330024
0.275000013	0.541472018
0.276000013	0.511846006
0.277000013	0.476592988
0.278000013	0.440847993
0.279000013	0.411087006
0.280000013	0.391357988
0.281000013	0.371829987
0.282000013	0.349985987
0.283000013	0.327553004
0.284000013	0.30012399
0.285000014	0.27195701
0.286000014	0.249195993
0.287000014	0.229559004
0.288000014	0.207479998
0.289000014	0.185910001
0.290000014	0.165367007
0.291000014	0.143022999
0.292000014	0.123816997
0.293000014	0.106270999
0.294000014	0.095431998
0.295000014	0.085505001
0.296000014	0.073491
0.297000014	0.060469002
0.298000014	0.050517999
0.299000014	0.041076001
0.300000014	0.036125001
0.301000014	0.032513
0.302000014	0.031314999
0.303000014	0.027524
0.304000014	0.027438
0.305000014	0.022208
0.306000015	0.018145001
0.307000015	0.010954
0.308000015	0.010045
0.309000015	0.007821
0.310000015	0.008266
0.311000015	0.008038
0.312000015	0.002474
0.313000015	0.00402
0.314000015	0.001192
0.315000015	-0.002597
0.316000015	-0.008283
0.317000015	-0.009751
0.318000015	-0.013525
0.319000015	-0.010867
0.320000015	-0.006311

    };
    
    
     \addplot [thick, red] table[
    y]
    {
  -0.001	0.136363636
2.4E-05	0.163636364
0.00104	0.163636364
0.002064	0.145454545
0.00308	0.163636364
0.004104	0.145454545
0.00512	0.172727273
0.006144	0.172727273
0.00716	0.163636364
0.008184	0.163636364
0.009192	0.163636364
0.010216	0.163636364
0.011232	0.154545455
0.012256	0.154545455
0.013272	0.172727273
0.014296	0.181818182
0.015312	0.163636364
0.016336	0.236363636
0.017352	0.263636364
0.018376	0.354545455
0.019392	0.363636364
0.020416	0.445454545
0.021432	0.590909091
0.022456	0.736363636
0.023472	0.945454545
0.024584	1.054545455
0.025688	1.163636364
0.026792	1.236363636
0.027896	1.354545455
0.029	1.545454545
0.030104	1.772727273
0.031208	1.945454545
0.032312	2.036363636
0.033416	2.081818182
0.03452	2.136363636
0.035632	2.172727273
0.036736	2.245454545
0.03784	2.390909091
0.038952	2.545454545
0.040048	2.545454545
0.041152	2.563636364
0.04226	2.6
0.04336	2.681818182
0.044472	2.827272727
0.045576	2.918181818
0.04668	2.981818182
0.047784	2.981818182
0.048888	3
0.049992	3.018181818
0.051096	3.081818182
0.0522	3.209090909
0.053316	3.254545455
0.054416	3.254545455
0.05552	3.3
0.056628	3.363636364
0.057728	3.427272727
0.058832	3.536363636
0.059936	3.581818182
0.06104	3.618181818
0.062152	3.618181818
0.063256	3.636363636
0.06436	3.681818182
0.065464	3.790909091
0.066576	3.845454545
0.06768	3.854545455
0.068776	3.863636364
0.069888	3.881818182
0.070992	3.881818182
0.072096	3.9
0.0732	3.890909091
0.074304	3.909090909
0.075408	3.881818182
0.076512	3.918181818
0.077616	3.909090909
0.07872	3.909090909
0.079832	3.909090909
0.080944	3.9
0.08204	3.890909091
0.083144	3.854545455
0.084252	3.818181818
0.085352	3.827272727
0.086456	3.8
0.08756	3.790909091
0.088672	3.772727273
0.089776	3.690909091
0.09088	3.645454545
0.091984	3.609090909
0.093088	3.6
0.0942	3.590909091
0.095304	3.563636364
0.0964	3.472727273
0.097512	3.418181818
0.098616	3.381818182
0.09972	3.372727273
0.100824	3.363636364
0.101928	3.327272727
0.103032	3.245454545
0.104136	3.181818182
0.10524	3.154545455
0.106352	3.127272727
0.107456	3.109090909
0.108568	3.072727273
0.109672	2.981818182
0.110768	2.881818182
0.111876	2.781818182
0.112976	2.772727273
0.11408	2.763636364
0.115192	2.727272727
0.116296	2.627272727
0.1174	2.5
0.118504	2.381818182
0.119608	2.327272727
0.120712	2.309090909
0.121816	2.3
0.122928	2.236363636
0.124032	2.190909091
0.125136	2.090909091
0.12624	1.990909091
0.127344	1.9
0.128448	1.872727273
0.129552	1.836363636
0.130656	1.727272727
0.13176	1.609090909
0.132872	1.454545455
0.133976	1.381818182
0.13508	1.336363636
0.136192	1.281818182
0.137296	1.190909091
0.138392	1.081818182
0.1395	0.990909091
0.140608	0.963636364
0.141712	0.890909091
0.142728	0.863636364
0.143752	0.836363636
0.144768	0.727272727
0.145792	0.718181818
0.146808	0.6
0.147832	0.545454545
0.148848	0.5
0.149872	0.436363636
0.150888	0.390909091
0.151912	0.372727273
0.152928	0.372727273
0.153952	0.345454545
0.154968	0.281818182
0.155992	0.290909091
0.157008	0.263636364
0.158032	0.263636364
0.159048	0.236363636
0.160072	0.227272727
0.161088	0.181818182
0.162112	0.136363636
0.163132	0.190909091
0.164152	0.127272727
0.165172	0.145454545
0.166192	0.136363636
0.167212	0.127272727
0.168232	0.118181818
0.169252	0.109090909
0.170272	0.1
0.171292	0.081818182


    };
    
     \addplot [thick, blue, dashed] table[
    y]
    {
    0	0.218664005
1.00E-03	0.224258006
2.00E-03	0.221027002
3.00E-03	0.224301994
4.00E-03	0.223251
5.00E-03	0.215376005
6.00E-03	0.215874001
7.00E-03	0.214668006
8.00E-03	0.217859998
9.00E-03	0.220018998
1.00E-02	0.220771
0.011000001	0.227282003
1.20E-02	0.237001002
1.30E-02	0.257304996
1.40E-02	0.287059009
1.50E-02	0.322021008
1.60E-02	0.360377014
1.70E-02	0.407115996
1.80E-02	0.452547997
1.90E-02	0.503705025
0.020000001	0.554731011
2.10E-02	0.610468984
2.20E-02	0.67789501
2.30E-02	0.764523029
2.40E-02	0.877829015
2.50E-02	1.004194021
2.60E-02	1.115347028
2.70E-02	1.188171983
2.80E-02	1.225635052
2.90E-02	1.275542021
3.00E-02	1.358500004
0.031000001	1.477144003
3.20E-02	1.602455974
3.30E-02	1.69255805
3.40E-02	1.735095978
3.50E-02	1.757030964
3.60E-02	1.781474948
3.70E-02	1.842911959
3.80E-02	1.944740057
3.90E-02	2.058753014
4.00E-02	2.165983915
4.10E-02	2.227641106
0.042000002	2.247035027
4.30E-02	2.261790037
4.40E-02	2.309355021
4.50E-02	2.404432058
4.60E-02	2.537523031
4.70E-02	2.666587114
4.80E-02	2.733292103
4.90E-02	2.741765976
5.00E-02	2.736298084
0.051000002	2.767746925
5.20E-02	2.853780985
5.30E-02	2.997373104
5.40E-02	3.132855892
5.50E-02	3.201487064
5.60E-02	3.20264411
5.70E-02	3.208693981
5.80E-02	3.253757
5.90E-02	3.363452911
6.00E-02	3.504626036
6.10E-02	3.620284081
0.062000003	3.659827948
6.30E-02	3.658649921
6.40E-02	3.670836926
6.50E-02	3.729419947
6.60E-02	3.849766016
6.70E-02	3.993725061
6.80E-02	4.099163055
6.90E-02	4.133265018
7.00E-02	4.129829884
0.071000003	4.126695156
7.20E-02	4.123055935
7.30E-02	4.121329784
7.40E-02	4.118787766
7.50E-02	4.115839958
7.60E-02	4.11125803
7.70E-02	4.105840206
7.80E-02	4.101448059
7.90E-02	4.10031414
8.00E-02	4.090068817
8.10E-02	4.080126762
0.082000004	4.065956116
8.30E-02	4.026463032
8.40E-02	3.966536045
8.50E-02	3.903388977
8.60E-02	3.847882986
8.70E-02	3.803672075
8.80E-02	3.77888608
8.90E-02	3.744364977
9.00E-02	3.687768936
9.10E-02	3.606170893
9.20E-02	3.511447906
0.093000004	3.444077015
9.40E-02	3.419089079
9.50E-02	3.407258034
9.60E-02	3.371182919
9.70E-02	3.298603058
9.80E-02	3.20062995
9.90E-02	3.108127117
0.100000005	3.059220076
0.101000005	3.044122934
0.102000005	3.024871111
0.103000005	2.979624033
0.104000005	2.896147966
0.105000005	2.796395063
0.106000005	2.716042995
0.107000005	2.67633605
0.108000005	2.656296015
0.109000005	2.631401062
0.110000005	2.58316803
0.111000005	2.497050047
0.112000005	2.394267082
0.113000005	2.309604883
0.114000005	2.276000023
0.115000005	2.263103008
0.116000006	2.233514071
0.117000006	2.18001008
0.118000006	2.092425108
0.119000006	1.991070986
0.120000006	1.909914017
0.121000006	1.864012957
0.122000006	1.849153042
0.123000006	1.822656989
0.124000006	1.767042041
0.125000006	1.692978024
0.126000006	1.605648994
0.127000006	1.520702004
0.128000006	1.458199024
0.129000006	1.417006016
0.130000006	1.386875987
0.131000006	1.340664983
0.132000006	1.279585004
0.133000006	1.196527004
0.134000006	1.116742015
0.135000006	1.056923985
0.136000006	1.015041947
0.137000007	0.983848989
0.138000007	0.940479994
0.139000007	0.879584014
0.140000007	0.804207027
0.141000007	0.731953979
0.142000007	0.668950975
0.143000007	0.618907988
0.144000007	0.582606971
0.145000007	0.539730012
0.146000007	0.499053001
0.147000007	0.455323994
0.148000007	0.400920987
0.149000007	0.349386007
0.150000007	0.295641005
0.151000007	0.250986993
0.152000007	0.214969993
0.153000007	0.186821997
0.154000007	0.156165004
0.155000007	0.126553997
0.156000007	0.100657001
0.157000007	0.074551001
0.158000008	0.058079999
0.159000008	0.043255001
0.160000008	0.027959

    };
    
    \addlegendentry[scale=0.8]{Sensor A201}
    \addlegendentry[scale=0.8]{MMT-101N}
\node[black, scale=0.6] at (axis cs: 0.06,4.3) {$v=50$};
\node[black, scale=0.6] at (axis cs: 0.15,4.4) {$v=25$};
\node[black, scale=0.6] at (axis cs: 0.25,4.4) {$v=12.5$};
\end{axis}
\end{tikzpicture}
\caption{Flexiforce sensor dynamic response.}
\label{fig:flexiforcedynresponse}
\end{figure}

\section{System Architecture}

%\Figure[h!]{SC}
%{System architecture.\label{fig.1}}

%\begin{figure}[h!] 
 %   \def\svgwidth{\columnwidth}
  %  \input{SC.pdf_tex}
   % \caption{System architecture.\label{fig1}}
%\end{figure}


The proposed system architecture (Fig. \ref{fig1}) aims to create a data provider--data processor system in which the data provider is a measurer module responsible for measuring data regarding all components of GRFs for each robot foot separately. This module is also responsible for relaying this information to the data processor. The data processor is accountable for storing, processing and presenting such data.
\begin{figure}[h!] 
     \includegraphics[width=0.5\textwidth]{adc}
    \caption{System basic architecture.  \label{fig1}}
\end{figure}

The data provider is assembled by two major modules: the control module and the capture module. The control module is responsible for all the initial settings required, from the time the data provider runs until it connects to the data processor. Firstly, starts by setting the WiFi module using AT commands. These commands allow the WiFi module to establish a connection with the data processor and further define the protocol as UDP. The capture module implements the capture process, being responsible for measuring data from all the sensors synchronously. The module operates based on a timer, ensuring periodic data acquisition for each sensor. The diagram in Fig. \ref{fig333} illustrates the acquisition process. 

\begin{figure}[h!] 
     \includegraphics[width=0.48\textwidth]{qwe}
    \caption{Acquisition process. F = Force applied to the sensor; S = Sensor; Rv = Resistance variation; M = capture Module; T = Timer.  \label{fig333}}
\end{figure}

The data processor is also formed by two major modules: the storage and the analysis modules. On the one hand, the storage module receives, stores, and displays the gathered data for further analysis. Essentially, in ROS environment, two functions (nodes) are used to receive and print the raw data. The data processor is configured to work as an access point (AP) creating a wireless local area network (WLAN). It waits for the ITshoe to be connected and starts receiving UDP messages. Well-formed messages are processed and made available for display, and also stored in a comma separated format file for further analysis by external applications. The second function subscribes the received message and generates a graphic environment (Fig. \ref{fig444}) that facilitates the understanding of the data evolution, as well as the CoP behaviour. \par

On the other hand, the analysis module implements a data analysis interface in order to convert the raw data into forces and to extract relevant information.
The proposed system is designed to measure and transmit raw data at a frequency of \SI{100}{\hertz}.

\begin{figure}[h!] 
     \includegraphics[width=0.48\textwidth]{ret}
    \caption{Graphical representation of the force values (example using the LITshoe). 1- ITshoes; 2- Humanoid robot NAO. \label{fig444}}
\end{figure}

\section{Experiments and Results}

This section describes the experiments and data acquisition results. The raw data upload rate is evaluated, and the GRFs are detected during two humanoid robot situations: static position and locomotion for two different grounds. 

\subsection{Data upload rate percentage loss}

The ITshoes were turned on and data was collected for one hour; table \ref{table1} shows the data upload percentage loss for that time interval. By observing the results, it can be concluded that the developed ITshoes have a very good upload rate since the data loss is negligible. The loss of data can be justified with the instability of the WiFi module.

\begin{table}[h]
\centering
\caption{Data upload rate percentage loss for both ITshoes.}
\label{table1}
\begin{tabular}{lccc}%{l|c|c|c|}
%\cline{2-4}
%                                    & \begin{tabular}[c]{@{}c@{}}Transmitted \\ data\end{tabular} & \begin{tabular}[c]{@{}c@{}}Received\\ data\end{tabular} & \begin{tabular}[c]{@{}c@{}}data loss\\  {[}\%{]}\end{tabular} \\
%\hline
% \multicolumn{1}{|l|}{Left ITshoe}  & 360000                                                      & 359894                                                  & 0.0295                                                        \\ %\hline
% \multicolumn{1}{|l|}{Right ITshoe} & 360000                                                      & 359783                                                  & 0.0603                                                        \\ %\hline
%------------------------------------
\toprule
  & Bytes sent &Bytes received & Data loss [\%]\\
\midrule
Left ITshoe  & 360000 & 359894 & 0.03 \\
Right ITshoe & 360000 & 359783 & 0.06 \\
\bottomrule
%-----------------------
\end{tabular}
\end{table}

\subsection{Detection of the vGRFs for a static robot position}



Fig. \ref{fig:sta1} illustrates the position of the humanoid robot where the vGRFs (sum of the values measured by the four sensors responsible to measure this component of the GRFs) are gathered. As can be seen in Fig. \ref{g2}, despite the visible differences between the total forces measured by the two ITshoes, the sum \SI{53.19 +- 0.32}{N} ends up to be close to the robot weight: \SI{5.4}{kg}, approx. \SI{52.96}{N}.

\begin{figure}[h]
\centering
  \includegraphics[width=0.9\columnwidth]{sta1}
\caption{Robot in a static position.}
\label{fig:sta1}
\end{figure}

\begin{figure}[h]
\centering
 \begin{tikzpicture}[scale = 0.8]
    \begin{axis}[
    title={\textbf{vGRFs (static robot position)}},
    legend style={at={(axis cs:2.75,40)},anchor=south west},
    xlabel={Time[s]},
    ylabel={Force[N]},
    xmin=0,xmax=5,
    ymin=20,ymax=55,
    xtick={0,0.5,1,1.5,2,2.5,3,3.5,4,4.5,5},
    ytick={20, 30, 40, 50, 55}
    ]
    \addplot [thick, red] table[
    y]
    {
  
       0	24.76991669
0.01	24.76940829
0.02	25.03882459
0.03	24.81944839
0.04	24.75679275
0.05	24.72788682
0.06	24.79827574
0.07	24.74011857
0.08	24.72950644
0.09	24.71043823
0.1	24.76847762
0.11	24.65913805
0.12	24.72699462
0.13	24.6980887
0.14	24.7782948
0.15	24.66840835
0.16	24.56993967
0.17	24.55085092
0.18	24.54083787
0.19	24.52174913
0.2	24.59213805
0.21	24.5730493
0.22	24.64370611
0.23	24.6829353
0.24	24.82004568
0.25	24.81333594
0.26	24.74765433
0.27	24.53799141
0.28	24.52842911
0.29	24.53799141
0.3	24.69554416
0.31	24.59562925
0.32	24.82195156
0.33	24.82550551
0.34	24.83845198
0.35	24.65123046
0.36	24.67033981
0.37	24.60843173
0.38	24.72354054
0.39	24.67540167
0.4	24.74712693
0.41	24.63719447
0.42	24.73730975
0.43	24.64321963
0.44	24.7141877
0.45	24.66606957
0.46	24.5989107
0.47	24.63719447
0.48	24.7756507
0.49	24.68165352
0.5	24.74666259
0.51	24.77056417
0.52	24.68902475
0.53	24.74168907
0.54	24.79904702
0.55	24.83255713
0.56	24.90464738
0.57	24.76558719
0.58	24.93363549
0.59	24.82322503
0.6	24.84778408
0.61	24.73237763
0.62	24.72752617
0.63	24.54660145
0.64	24.67967191
0.65	24.71817973
0.66	24.76081908
0.67	24.66054189
0.68	24.79904702
0.69	24.83255713
0.7	24.71275813
0.71	24.60831798
0.72	24.6801296
0.73	24.67996656
0.74	24.7847172
0.75	24.82785879
0.76	24.92899262
0.77	24.85242808
0.78	24.72997656
0.79	24.62027906
0.8	24.69262941
0.81	24.62027906
0.82	24.78628943
0.83	24.75780658
0.84	24.80222933
0.85	24.6734994
0.86	24.70198225
0.87	24.59948967
0.88	24.55606724
0.89	24.5077646
0.9	24.52598857
0.91	24.56126802
0.92	24.57977735
0.93	24.53359658
0.94	24.59700511
0.95	24.50712979
0.96	24.63528484
0.97	24.49852381
0.98	24.52743252
0.99	24.53484953
1	24.52743252
1.01	24.49852381
1.02	24.52698745
1.03	24.53440293
1.04	24.55353447
1.05	24.39008101
1.06	24.25803211
1.07	24.18596107
1.08	24.17897877
1.09	24.08091578
1.1	24.15324097
1.11	24.18770378
1.12	24.20683533
1.13	24.24008821
1.14	24.33882948
1.15	24.39968818
1.16	24.44498932
1.17	24.4793612
1.18	24.56997943
1.19	24.55094544
1.2	24.51646189
1.21	24.49735108
1.22	24.59757507
1.23	24.63312293
1.24	24.67908128
1.25	24.61405195
1.26	24.7149238
1.27	24.67760791
1.28	24.70603015
1.29	24.55939329
1.3	24.60531181
1.31	24.50502691
1.32	24.52409789
1.33	24.45095049
1.34	24.45285894
1.35	24.23749845
1.36	24.34531456
1.37	24.31691148
1.38	24.31866793
1.39	24.37972634
1.4	24.50620112
1.41	24.62250113
1.42	24.69625144
1.43	24.67715979
1.44	24.77825551
1.45	24.75916386
1.46	24.77950627
1.47	24.73211312
1.48	24.69625144
1.49	24.62336338
1.5	24.69625144
1.51	24.67715979
1.52	24.56963038
1.53	24.40647315
1.54	24.53342999
1.55	24.48710947
1.56	24.34568858
1.57	24.32659693
1.58	24.53314998
1.59	24.40647315
1.6	24.42556479
1.61	24.48759154
1.62	24.58779637
1.63	24.53264648
1.64	24.67876482
1.65	24.65044414
1.66	24.66947399
1.67	24.65971444
1.68	24.74332662
1.69	24.57977398
1.7	24.63468281
1.71	24.50766353
1.72	24.52673451
1.73	24.56065851
1.74	24.62537138
1.75	24.59549082
1.76	24.61452067
1.77	24.49670901
1.78	24.59661658
1.79	24.48743871
1.8	24.50646856
1.81	24.56874401
1.82	24.53311519
1.83	24.48743871
1.84	24.50646856
1.85	24.51408535
1.86	24.47992154
1.87	24.40652532
1.88	24.53311519
1.89	24.51408535
1.9	24.64272718
1.91	24.66020369
1.92	24.70561237
1.93	24.63296763
1.94	24.65201801
1.95	24.60476112
1.96	24.50646856
1.97	24.52335565
1.98	24.40839804
1.99	24.42508645
2	24.39985971
2.01	24.37147875
2.02	24.43396323
2.03	24.38934766
2.04	24.35579707
2.05	24.41579562
2.06	24.56104956
2.07	24.59555227
2.08	24.59884029
2.09	24.52507082
2.1	24.60819313
2.11	24.54195791
2.12	24.52581473
2.13	24.51531128
2.14	24.51763461
2.15	24.36363305
2.16	24.53484953
2.17	24.48803965
2.18	24.4073166
2.19	24.38875547
2.2	24.46139302
2.21	24.38076958
2.22	24.42640673
2.23	24.37147875
2.24	24.35480949
2.25	24.33576048
2.26	24.46188089
2.27	24.46947852
2.28	24.43533387
2.29	24.55185693
2.3	24.65114692
2.31	24.60438811
2.32	24.5696407
2.33	24.49719826
2.34	24.40780447
2.35	24.28146326
2.36	24.30051226
2.37	24.28146326
2.38	24.35401569
2.39	24.30810989
2.4	24.39046023
2.41	24.36161332
2.42	24.44374193
2.43	24.33496669
2.44	24.44374193
2.45	24.4154021
2.46	24.47048874
2.47	24.39763043
2.48	24.52428515
2.49	24.42447798
2.5	24.39975107
2.51	24.39763043
2.52	24.41669996
2.53	24.344127
2.54	24.33644973
2.55	24.37250797
2.56	24.33644973
2.57	24.344127
2.58	24.39935552
2.59	24.344127
2.6	24.33644973
2.61	24.3539057
2.62	24.39935552
2.63	24.35341783
2.64	24.39004408
2.65	24.344127
2.66	24.47049637
2.67	24.37097455
2.68	24.39004408
2.69	24.3539057
2.7	24.36319653
2.71	24.3539057
2.72	24.36319653
2.73	24.36272926
2.74	24.36319653
2.75	24.344127
2.76	24.37250797
2.77	24.31738019
2.78	24.33644973
2.79	24.31738019
2.8	24.31911453
2.81	24.29073356
2.82	24.34576116
2.83	24.37088362
2.84	24.39926459
2.85	24.36330652
2.86	24.34576116
2.87	24.33598246
2.88	24.40859669
2.89	24.30933582
2.9	24.38195006
2.91	24.36283925
2.92	24.41794953
2.93	24.36283925
2.94	24.37261795
2.95	24.35352782
2.96	24.37261795
2.97	24.30051226
2.98	24.31911453
2.99	24.36330652
3	24.31911453
3.01	24.30933582
3.02	24.32844663
3.03	24.30002439
3.04	24.31911453
3.05	24.29073356
3.06	24.30980309
3.07	24.27396524
3.08	24.37261795
3.09	24.37261795
3.1	24.40859669
3.11	24.38948588
3.12	24.35509326
3.13	24.38995315
3.14	24.31911453
3.15	24.23003573
3.16	24.29256751
3.17	24.30980309
3.18	24.26589419
3.19	24.28325607
3.2	24.27545165
3.21	24.31962297
3.22	24.32895507
3.23	24.31911453
3.24	24.40859669
3.25	24.36272926
3.26	24.34613518
3.27	24.35574623
3.28	24.31083311
3.29	24.27544429
3.3	24.26748368
3.31	24.21376902
3.32	24.23283856
3.33	24.21376902
3.34	24.23283856
3.35	24.22354773
3.36	24.24263647
3.37	24.22354773
3.38	24.20578781
3.39	24.22354773
3.4	24.23283856
3.41	24.29553839
3.42	24.29927332
3.43	24.28996188
3.44	24.35186996
3.45	24.19859077
3.46	24.18072839
3.47	24.24092224
3.48	24.25817224
3.49	24.29330667
3.5	24.3123762
3.51	24.22354773
3.52	24.17883895
3.53	24.1519914
3.54	24.1519914
3.55	24.14270057
3.56	24.195085
3.57	24.1858147
3.58	24.10668346
3.59	24.14124928
3.6	24.16954812
3.61	24.14124928
3.62	24.14438681
3.63	24.17954177
3.64	24.18932047
3.65	24.19754819
3.66	24.24713165
3.67	24.30258643
3.68	24.18120809
3.69	24.06035409
3.7	24.1071835
3.71	24.02837574
3.72	23.93791365
3.73	24.03109974
3.74	24.16404833
3.75	24.01891317
3.76	24.13539729
3.77	24.18808994
3.78	24.14454472
3.79	24.20440939
3.8	24.27994804
3.81	24.2478283
3.82	24.32404788
3.83	24.19731938
3.84	24.10685729
3.85	24.15426963
3.86	24.10632843
3.87	24.06665331
3.88	24.01891317
3.89	23.96020164
3.9	23.97868092
3.91	23.88479839
3.92	23.9596314
3.93	23.9503611
3.94	23.9503611
3.95	23.90113083
3.96	23.93872095
3.97	23.97868092
3.98	23.93872095
3.99	23.96892137
4	24.00711051
4.01	24.07588275
4.02	24.08513258
4.03	24.20426118
4.04	24.33995922
4.05	24.43194794
4.06	24.41236368
4.07	24.43558708
4.08	24.45461562
4.09	24.30501934
4.1	24.33378208
4.11	24.17561013
4.12	24.15426963
4.13	24.20267803
4.14	24.13486842
4.15	24.09707859
4.16	24.19228221
4.17	24.03758321
4.18	24.0356505
4.19	24.13329653
4.2	24.14254637
4.21	24.09707859
4.22	24.04734275
4.23	24.11638914
4.24	24.19228221
4.25	24.18303237
4.26	24.23971844
4.27	24.30013697
4.28	24.22278131
4.29	24.13486842
4.3	24.20072413
4.31	24.28041916
4.32	24.46848336
4.33	24.28561813
4.34	24.41285155
4.35	24.47371201
4.36	24.43194794
4.37	24.46144446
4.38	24.46032891
4.39	24.43194794
4.4	24.42214238
4.41	24.58732285
4.42	24.53832352
4.43	24.52903269
4.44	24.47826206
4.45	24.54755435
4.46	24.5962837
4.47	24.57589419
4.48	24.56660336
4.49	24.42775314
4.5	24.30758704
4.51	24.35684352
4.52	24.4161259
4.53	24.40683507
4.54	24.39558122
4.55	24.4565749
4.56	24.5962837
4.57	24.69699035
4.58	24.67685245
4.59	24.6980887
4.6	24.77918699
4.61	24.74938887
4.62	24.70737953
4.63	24.73717765
4.64	24.6980887
4.65	24.64705433
4.66	24.62608182
4.67	24.65599842
4.68	24.65599842
4.69	24.5962192
4.7	24.63552529
4.71	24.73680484
4.72	24.62608182
4.73	24.65599842
4.74	24.63552529
4.75	24.55524026
4.76	24.5755729
4.77	24.52532366
4.78	24.53461449
4.79	24.49552554
4.8	24.53461449
4.81	24.50524871
4.82	24.42559627
4.83	24.39579815
4.84	24.46480371
4.85	24.46606181
4.86	24.54457534
4.87	24.61543175
4.88	24.74646623
4.89	24.81836422
4.9	24.80738364
4.91	24.84888397
4.92	24.71618991
4.93	24.5663026
4.94	24.52532366
4.95	24.46606181
4.96	24.48513002
4.97	24.56503818
4.98	24.63552529
4.99	24.76748815

    };
    \addplot [thick, green] table[
    y]
    {
  
0	28.5060241
0.01	28.48192771
0.02	28.3253012
0.03	28.46987952
0.04	28.46987952
0.05	28.39759036
0.06	28.36144578
0.07	28.43373494
0.08	28.38554217
0.09	28.3373494
0.1	28.3253012
0.11	28.39759036
0.12	28.37349398
0.13	28.42168675
0.14	28.4939759
0.15	28.51807229
0.16	28.43373494
0.17	28.48192771
0.18	28.51807229
0.19	28.55421687
0.2	28.53012048
0.21	28.5060241
0.22	28.4939759
0.23	28.48192771
0.24	28.40963855
0.25	28.3253012
0.26	28.28915663
0.27	28.28915663
0.28	28.28915663
0.29	28.31325301
0.3	28.28915663
0.31	28.30120482
0.32	28.27710843
0.33	28.30120482
0.34	28.28915663
0.35	28.37349398
0.36	28.42168675
0.37	28.37349398
0.38	28.39759036
0.39	28.45783133
0.4	28.42168675
0.41	28.43373494
0.42	28.44578313
0.43	28.45783133
0.44	28.43373494
0.45	28.44578313
0.46	28.44578313
0.47	28.39759036
0.48	28.38554217
0.49	28.39759036
0.5	28.37349398
0.51	28.36144578
0.52	28.38554217
0.53	28.39759036
0.54	28.38554217
0.55	28.40963855
0.56	28.43373494
0.57	28.44578313
0.58	28.42168675
0.59	28.44578313
0.6	28.44578313
0.61	28.48192771
0.62	28.46987952
0.63	28.4939759
0.64	28.45783133
0.65	28.42168675
0.66	28.40963855
0.67	28.42168675
0.68	28.37349398
0.69	28.37349398
0.7	28.34939759
0.71	28.34939759
0.72	28.3373494
0.73	28.34939759
0.74	28.34939759
0.75	28.3373494
0.76	28.34939759
0.77	28.37349398
0.78	28.37349398
0.79	28.38554217
0.8	28.36144578
0.81	28.39759036
0.82	28.39759036
0.83	28.39759036
0.84	28.38554217
0.85	28.37349398
0.86	28.34939759
0.87	28.36144578
0.88	28.36144578
0.89	28.34939759
0.9	28.37349398
0.91	28.3373494
0.92	28.31325301
0.93	28.3373494
0.94	28.36144578
0.95	28.3373494
0.96	28.3253012
0.97	28.3373494
0.98	28.36144578
0.99	28.37349398
1	28.38554217
1.01	28.42168675
1.02	28.42168675
1.03	28.40963855
1.04	28.39759036
1.05	28.38554217
1.06	28.39759036
1.07	28.37349398
1.08	28.39759036
1.09	28.37349398
1.1	28.34939759
1.11	28.31325301
1.12	28.3253012
1.13	28.34939759
1.14	28.3373494
1.15	28.36144578
1.16	28.37349398
1.17	28.38554217
1.18	28.37349398
1.19	28.38554217
1.2	28.38554217
1.21	28.38554217
1.22	28.38554217
1.23	28.37349398
1.24	28.31325301
1.25	28.3253012
1.26	28.3253012
1.27	28.34939759
1.28	28.31325301
1.29	28.31325301
1.3	28.30120482
1.31	28.28915663
1.32	28.27710843
1.33	28.26506024
1.34	28.31325301
1.35	28.30120482
1.36	28.26506024
1.37	28.27710843
1.38	28.36144578
1.39	28.36144578
1.4	28.37349398
1.41	28.37349398
1.42	28.36144578
1.43	28.34939759
1.44	28.34939759
1.45	28.42168675
1.46	28.42168675
1.47	28.3373494
1.48	28.24096386
1.49	28.25301205
1.5	28.19277108
1.51	28.10843373
1.52	28.09638554
1.53	28.06024096
1.54	28.06024096
1.55	28.08433735
1.56	28.09638554
1.57	28.04819277
1.58	28.04819277
1.59	28.03614458
1.6	28.02409639
1.61	28.04819277
1.62	28
1.63	27.93975904
1.64	27.89156627
1.65	27.98795181
1.66	27.97590361
1.67	27.95180723
1.68	27.96385542
1.69	28
1.7	28.03614458
1.71	28.06024096
1.72	28.14457831
1.73	28.19277108
1.74	28.21686747
1.75	28.21686747
1.76	28.30120482
1.77	28.31325301
1.78	28.24096386
1.79	28.27710843
1.8	28.28915663
1.81	28.28915663
1.82	28.3253012
1.83	28.40963855
1.84	28.37349398
1.85	28.3253012
1.86	28.31325301
1.87	28.31325301
1.88	28.22891566
1.89	28.21686747
1.9	28.19277108
1.91	28.1686747
1.92	28.18072289
1.93	28.28915663
1.94	28.30120482
1.95	28.31325301
1.96	28.3373494
1.97	28.36144578
1.98	28.40963855
1.99	28.40963855
2	28.38554217
2.01	28.42168675
2.02	28.43373494
2.03	28.44578313
2.04	28.53012048
2.05	28.53012048
2.06	28.56626506
2.07	28.59036145
2.08	28.62650602
2.09	28.63855422
2.1	28.60240964
2.11	28.56626506
2.12	28.55421687
2.13	28.53012048
2.14	28.51807229
2.15	28.55421687
2.16	28.56626506
2.17	28.53012048
2.18	28.45783133
2.19	28.51807229
2.2	28.60240964
2.21	28.63855422
2.22	28.57831325
2.23	28.61445783
2.24	28.63855422
2.25	28.59036145
2.26	28.59036145
2.27	28.60240964
2.28	28.61445783
2.29	28.61445783
2.3	28.61445783
2.31	28.61445783
2.32	28.69879518
2.33	28.80722892
2.34	28.84337349
2.35	28.81927711
2.36	28.85542169
2.37	28.87951807
2.38	28.84337349
2.39	28.84337349
2.4	28.75903614
2.41	28.74698795
2.42	28.69879518
2.43	28.69879518
2.44	28.63855422
2.45	28.71084337
2.46	28.69879518
2.47	28.63855422
2.48	28.6746988
2.49	28.71084337
2.5	28.72289157
2.51	28.8313253
2.52	28.85542169
2.53	28.90361446
2.54	28.92771084
2.55	28.92771084
2.56	28.96385542
2.57	28.97590361
2.58	28.92771084
2.59	28.91566265
2.6	28.95180723
2.61	28.96385542
2.62	28.96385542
2.63	28.93975904
2.64	28.96385542
2.65	28.97590361
2.66	28.97590361
2.67	28.93975904
2.68	28.92771084
2.69	28.93975904
2.7	28.95180723
2.71	28.95180723
2.72	29.01204819
2.73	29.02409639
2.74	29.03614458
2.75	29.04819277
2.76	29.06024096
2.77	29.08433735
2.78	29.13253012
2.79	29.12048193
2.8	29.19277108
2.81	29.13253012
2.82	29.13253012
2.83	29.14457831
2.84	29.1686747
2.85	29.14457831
2.86	29.07228916
2.87	29.14457831
2.88	29.13253012
2.89	29.04819277
2.9	29.09638554
2.91	29.18072289
2.92	29.19277108
2.93	29.15662651
2.94	29.19277108
2.95	29.26506024
2.96	29.27710843
2.97	29.26506024
2.98	29.22891566
2.99	29.25301205
3	29.27710843
3.01	29.30120482
3.02	29.3373494
3.03	29.34939759
3.04	29.34939759
3.05	29.34939759
3.06	29.31325301
3.07	29.26506024
3.08	29.27710843
3.09	29.21686747
3.1	29.21686747
3.11	29.3373494
3.12	29.34939759
3.13	29.34939759
3.14	29.37349398
3.15	29.38554217
3.16	29.36144578
3.17	29.3253012
3.18	29.36144578
3.19	29.39759036
3.2	29.42168675
3.21	29.43373494
3.22	29.40963855
3.23	29.38554217
3.24	29.36144578
3.25	29.40963855
3.26	29.43373494
3.27	29.40963855
3.28	29.36144578
3.29	29.37349398
3.3	29.44578313
3.31	29.46987952
3.32	29.44578313
3.33	29.43373494
3.34	29.37349398
3.35	29.34939759
3.36	29.42168675
3.37	29.45783133
3.38	29.40963855
3.39	29.39759036
3.4	29.34939759
3.41	29.38554217
3.42	29.40963855
3.43	29.37349398
3.44	29.38554217
3.45	29.3253012
3.46	29.34939759
3.47	29.43373494
3.48	29.45783133
3.49	29.42168675
3.5	29.37349398
3.51	29.37349398
3.52	29.38554217
3.53	29.37349398
3.54	29.42168675
3.55	29.39759036
3.56	29.38554217
3.57	29.43373494
3.58	29.43373494
3.59	29.36144578
3.6	29.36144578
3.61	29.34939759
3.62	29.40963855
3.63	29.43373494
3.64	29.43373494
3.65	29.43373494
3.66	29.43373494
3.67	29.36144578
3.68	29.42168675
3.69	29.44578313
3.7	29.42168675
3.71	29.42168675
3.72	29.39759036
3.73	29.38554217
3.74	29.40963855
3.75	29.46987952
3.76	29.4939759
3.77	29.43373494
3.78	29.39759036
3.79	29.38554217
3.8	29.37349398
3.81	29.42168675
3.82	29.43373494
3.83	29.39759036
3.84	29.38554217
3.85	29.36144578
3.86	29.39759036
3.87	29.37349398
3.88	29.42168675
3.89	29.43373494
3.9	29.44578313
3.91	29.40963855
3.92	29.45783133
3.93	29.53012048
3.94	29.57831325
3.95	29.59036145
3.96	29.56626506
3.97	29.51807229
3.98	29.42168675
3.99	29.38554217
4	29.40963855
4.01	29.3373494
4.02	29.26506024
4.03	29.18072289
4.04	29.24096386
4.05	29.21686747
4.06	29.21686747
4.07	29.15662651
4.08	29.1686747
4.09	29.06024096
4.1	29.09638554
4.11	29.09638554
4.12	29.07228916
4.13	29.06024096
4.14	29.06024096
4.15	29.04819277
4.16	29.12048193
4.17	29.12048193
4.18	29.13253012
4.19	29.09638554
4.2	29.13253012
4.21	29.18072289
4.22	29.20481928
4.23	29.18072289
4.24	29.06024096
4.25	29.08433735
4.26	29.09638554
4.27	29.10843373
4.28	28.96385542
4.29	28.92771084
4.3	28.86746988
4.31	28.89156627
4.32	28.90361446
4.33	28.98795181
4.34	28.90361446
4.35	28.91566265
4.36	29.09638554
4.37	29.07228916
4.38	29
4.39	29.06024096
4.4	29.13253012
4.41	29.13253012
4.42	29.07228916
4.43	29.04819277
4.44	29.10843373
4.45	29.03614458
4.46	28.95180723
4.47	29.01204819
4.48	28.95180723
4.49	28.86746988
4.5	28.93975904
4.51	28.79518072
4.52	28.8313253
4.53	28.85542169
4.54	28.8313253
4.55	28.80722892
4.56	28.81927711
4.57	28.84337349
4.58	28.87951807
4.59	28.87951807
4.6	28.8313253
4.61	28.80722892
4.62	28.84337349
4.63	28.8313253
4.64	28.79518072
4.65	28.79518072
4.66	28.79518072
4.67	28.87951807
4.68	28.85542169
4.69	28.84337349
4.7	28.84337349
4.71	28.85542169
4.72	28.84337349
4.73	28.8313253
4.74	28.79518072
4.75	28.69879518
4.76	28.57831325
4.77	28.71084337
4.78	28.61445783
4.79	28.59036145
4.8	28.65060241
4.81	28.60240964
4.82	28.63855422
4.83	28.71084337
4.84	28.6746988
4.85	28.72289157
4.86	28.65060241
4.87	28.62650602
4.88	28.62650602
4.89	28.62650602
4.9	28.65060241
4.91	28.63855422
4.92	28.56626506
4.93	28.53012048
4.94	28.5060241
4.95	28.48192771
4.96	28.4939759
4.97	28.48192771
4.98	28.43373494
4.99	28.5060241


    };
    \addplot [thick, blue] table[
    y]
    {
  
      0	53.27594079
0.01	53.251336
0.02	53.36412579
0.03	53.28932791
0.04	53.22667226
0.05	53.12547718
0.06	53.15972152
0.07	53.17385351
0.08	53.11504861
0.09	53.04778763
0.1	53.09377882
0.11	53.05672842
0.12	53.1004886
0.13	53.11977544
0.14	53.2722707
0.15	53.18648064
0.16	53.00367461
0.17	53.03277863
0.18	53.05891016
0.19	53.075966
0.2	53.12225853
0.21	53.0790734
0.22	53.13768201
0.23	53.16486301
0.24	53.22968423
0.25	53.13863714
0.26	53.03681095
0.27	52.82714804
0.28	52.81758574
0.29	52.85124442
0.3	52.98470079
0.31	52.89683407
0.32	53.09905999
0.33	53.12671033
0.34	53.12760861
0.35	53.02472444
0.36	53.09202655
0.37	52.9819257
0.38	53.1211309
0.39	53.133233
0.4	53.16881368
0.41	53.07092941
0.42	53.18309288
0.43	53.10105096
0.44	53.14792264
0.45	53.1118527
0.46	53.04469383
0.47	53.03478483
0.48	53.16119287
0.49	53.07924388
0.5	53.12015657
0.51	53.13200995
0.52	53.07456692
0.53	53.13927943
0.54	53.18458919
0.55	53.24219569
0.56	53.33838232
0.57	53.21137032
0.58	53.35532224
0.59	53.26900816
0.6	53.29356722
0.61	53.21430535
0.62	53.19740568
0.63	53.04057735
0.64	53.13750323
0.65	53.13986648
0.66	53.17045764
0.67	53.08222864
0.68	53.172541
0.69	53.20605111
0.7	53.06215572
0.71	52.95771557
0.72	53.017479
0.73	53.02936415
0.74	53.13411479
0.75	53.16520818
0.76	53.27839021
0.77	53.22592206
0.78	53.10347054
0.79	53.00582122
0.8	53.0540752
0.81	53.01786942
0.82	53.1838798
0.83	53.15539694
0.84	53.1877715
0.85	53.04699337
0.86	53.05137984
0.87	52.96093545
0.88	52.91751302
0.89	52.85716219
0.9	52.89948254
0.91	52.89861742
0.92	52.89303036
0.93	52.87094598
0.94	52.95845089
0.95	52.84447918
0.96	52.96058605
0.97	52.8358732
0.98	52.88887831
0.99	52.9083435
1	52.91297469
1.01	52.92021055
1.02	52.9486742
1.03	52.94404148
1.04	52.95112483
1.05	52.77562318
1.06	52.65562247
1.07	52.55945505
1.08	52.57656913
1.09	52.45440975
1.1	52.50263856
1.11	52.5009568
1.12	52.53213653
1.13	52.58948581
1.14	52.67617888
1.15	52.76113397
1.16	52.81848329
1.17	52.86490337
1.18	52.94347341
1.19	52.93648761
1.2	52.90200406
1.21	52.88289325
1.22	52.98311724
1.23	53.00661691
1.24	52.9923343
1.25	52.93935316
1.26	53.04022501
1.27	53.0270055
1.28	53.01928316
1.29	52.8726463
1.3	52.90651663
1.31	52.79418354
1.32	52.80120633
1.33	52.71601073
1.34	52.76611196
1.35	52.53870327
1.36	52.6103748
1.37	52.59401991
1.38	52.68011371
1.39	52.74117212
1.4	52.8796951
1.41	52.9959951
1.42	53.05769723
1.43	53.02655739
1.44	53.1276531
1.45	53.18085061
1.46	53.20119302
1.47	53.06946251
1.48	52.9372153
1.49	52.87637543
1.5	52.88902253
1.51	52.78559353
1.52	52.66601592
1.53	52.46671411
1.54	52.59367096
1.55	52.57144682
1.56	52.44207412
1.57	52.3747897
1.58	52.58134275
1.59	52.44261772
1.6	52.44966118
1.61	52.53578431
1.62	52.58779637
1.63	52.47240551
1.64	52.57033109
1.65	52.63839595
1.66	52.6453776
1.67	52.61152167
1.68	52.70718204
1.69	52.57977398
1.7	52.67082739
1.71	52.56790449
1.72	52.67131282
1.73	52.7534296
1.74	52.84223885
1.75	52.81235829
1.76	52.91572549
1.77	52.80996203
1.78	52.83758043
1.79	52.76454715
1.8	52.79562519
1.81	52.85790064
1.82	52.8584164
1.83	52.89707727
1.84	52.87996254
1.85	52.83938655
1.86	52.79317455
1.87	52.71977833
1.88	52.76203086
1.89	52.73095282
1.9	52.83549826
1.91	52.82887839
1.92	52.88633526
1.93	52.92212426
1.94	52.95322283
1.95	52.91801413
1.96	52.84381796
1.97	52.88480143
1.98	52.81803659
1.99	52.83472501
2	52.78540188
2.01	52.79316549
2.02	52.86769817
2.03	52.83513079
2.04	52.88591755
2.05	52.9459161
2.06	53.12731462
2.07	53.18591371
2.08	53.22534632
2.09	53.16362504
2.1	53.21060277
2.11	53.10822297
2.12	53.08003159
2.13	53.04543176
2.14	53.0357069
2.15	52.91784992
2.16	53.10111459
2.17	53.01816013
2.18	52.86514792
2.19	52.90682776
2.2	53.06380266
2.21	53.0193238
2.22	53.00471999
2.23	52.98593658
2.24	52.9933637
2.25	52.92612193
2.26	53.05224233
2.27	53.07188816
2.28	53.0497917
2.29	53.16631476
2.3	53.26560475
2.31	53.21884594
2.32	53.26843588
2.33	53.30442718
2.34	53.25117796
2.35	53.10074037
2.36	53.15593395
2.37	53.16098133
2.38	53.19738918
2.39	53.15148339
2.4	53.14949638
2.41	53.10860127
2.42	53.14253711
2.43	53.03376187
2.44	53.08229615
2.45	53.12624547
2.46	53.16928392
2.47	53.03618464
2.48	53.19898395
2.49	53.13532135
2.5	53.12264263
2.51	53.22895573
2.52	53.27212165
2.53	53.24774146
2.54	53.26416057
2.55	53.30021881
2.56	53.30030515
2.57	53.32003061
2.58	53.32706636
2.59	53.25978965
2.6	53.28825695
2.61	53.31776112
2.62	53.36321094
2.63	53.29317687
2.64	53.3538995
2.65	53.32003061
2.66	53.44639999
2.67	53.31073359
2.68	53.31775493
2.69	53.29366474
2.7	53.31500376
2.71	53.30571293
2.72	53.37524473
2.73	53.38682565
2.74	53.39934111
2.75	53.39231977
2.76	53.43274893
2.77	53.40171754
2.78	53.46897985
2.79	53.43786212
2.8	53.51188561
2.81	53.42326368
2.82	53.47829128
2.83	53.51546193
2.84	53.56793929
2.85	53.50788483
2.86	53.41805032
2.87	53.48056077
2.88	53.54112681
2.89	53.3575286
2.9	53.4783356
2.91	53.54356214
2.92	53.61072061
2.93	53.51946576
2.94	53.56538904
2.95	53.61858806
2.96	53.64972639
2.97	53.5655725
2.98	53.54803019
2.99	53.61631857
3	53.59622296
3.01	53.61054064
3.02	53.66579603
3.03	53.64942198
3.04	53.66851212
3.05	53.64013115
3.06	53.62305611
3.07	53.53902548
3.08	53.64972639
3.09	53.58948542
3.1	53.62546416
3.11	53.72683528
3.12	53.70449085
3.13	53.73935074
3.14	53.6926085
3.15	53.6155779
3.16	53.65401329
3.17	53.6351043
3.18	53.62733997
3.19	53.68084644
3.2	53.69713839
3.21	53.75335791
3.22	53.73859363
3.23	53.70465669
3.24	53.77004247
3.25	53.77236782
3.26	53.77987011
3.27	53.76538479
3.28	53.67227889
3.29	53.64893826
3.3	53.71326681
3.31	53.68364854
3.32	53.67862169
3.33	53.64750396
3.34	53.60633253
3.35	53.57294532
3.36	53.66432322
3.37	53.68137905
3.38	53.61542637
3.39	53.62113809
3.4	53.58223615
3.41	53.68108056
3.42	53.70891187
3.43	53.66345586
3.44	53.73741213
3.45	53.52389198
3.46	53.53012598
3.47	53.67465718
3.48	53.71600357
3.49	53.71499341
3.5	53.68587018
3.51	53.5970417
3.52	53.56438112
3.53	53.52548538
3.54	53.57367815
3.55	53.54029093
3.56	53.58062717
3.57	53.61954964
3.58	53.5404184
3.59	53.50269506
3.6	53.5309939
3.61	53.49064687
3.62	53.55402537
3.63	53.61327671
3.64	53.62305541
3.65	53.63128313
3.66	53.68086659
3.67	53.66403222
3.68	53.60289484
3.69	53.50613722
3.7	53.52887024
3.71	53.45006248
3.72	53.33550401
3.73	53.41664191
3.74	53.57368688
3.75	53.48879268
3.76	53.62937319
3.77	53.62182488
3.78	53.54213508
3.79	53.58995156
3.8	53.65344202
3.81	53.66951505
3.82	53.75778281
3.83	53.59490974
3.84	53.49239946
3.85	53.51571541
3.86	53.50391879
3.87	53.44014728
3.88	53.44059991
3.89	53.39393658
3.9	53.42446405
3.91	53.29443694
3.92	53.41746272
3.93	53.48048158
3.94	53.52867435
3.95	53.49149228
3.96	53.50498601
3.97	53.49675321
3.98	53.3604077
3.99	53.35446354
4	53.41674906
4.01	53.41323214
4.02	53.35019282
4.03	53.38498407
4.04	53.58092308
4.05	53.64881541
4.06	53.62923115
4.07	53.59221359
4.08	53.62329032
4.09	53.3652603
4.1	53.43016762
4.11	53.27199568
4.12	53.22655878
4.13	53.262919
4.14	53.19510938
4.15	53.14527136
4.16	53.31276414
4.17	53.15806513
4.18	53.16818062
4.19	53.22968208
4.2	53.27507649
4.21	53.27780148
4.22	53.25216203
4.23	53.29711204
4.24	53.25252317
4.25	53.26736972
4.26	53.33610398
4.27	53.4085707
4.28	53.18663673
4.29	53.06257926
4.3	53.06819401
4.31	53.17198542
4.32	53.37209782
4.33	53.27356994
4.34	53.316466
4.35	53.38937466
4.36	53.52833349
4.37	53.53373361
4.38	53.46032891
4.39	53.49218891
4.4	53.5546725
4.41	53.71985297
4.42	53.61061268
4.43	53.57722546
4.44	53.5866958
4.45	53.58369893
4.46	53.54809093
4.47	53.58794238
4.48	53.51841058
4.49	53.29522302
4.5	53.24734608
4.51	53.15202424
4.52	53.2474512
4.53	53.26225675
4.54	53.22690652
4.55	53.26380381
4.56	53.41556081
4.57	53.54036384
4.58	53.55637052
4.59	53.57760677
4.6	53.61051229
4.61	53.55661779
4.62	53.55075302
4.63	53.56850295
4.64	53.49326942
4.65	53.44223505
4.66	53.42126254
4.67	53.5355165
4.68	53.51142011
4.69	53.43959269
4.7	53.47889879
4.71	53.59222653
4.72	53.46945531
4.73	53.48732372
4.74	53.43070602
4.75	53.25403544
4.76	53.15388615
4.77	53.23616703
4.78	53.14907232
4.79	53.08588698
4.8	53.1852169
4.81	53.10765835
4.82	53.06415049
4.83	53.10664152
4.84	53.1395025
4.85	53.18895338
4.86	53.19517775
4.87	53.24193778
4.88	53.37297225
4.89	53.44487025
4.9	53.45798605
4.91	53.48743819
4.92	53.28245497
4.93	53.09642308
4.94	53.03134776
4.95	52.94798952
4.96	52.97910593
4.97	53.04696589
4.98	53.06926023
4.99	53.27351225


    };
    \addlegendentry{LITshoe: $Fn_t$}
    \addlegendentry{RITshoe: $Fn_t$}
    \addlegendentry{SITshoe: $Fn_t$}

\end{axis}
\end{tikzpicture}
\caption{vGRFs for a robot static position. Left instrumented shoe (LITshoe) normal force (red), right instrumented shoe (RITshoe) normal force (green), and total (SITshoe) normal force (blue).}%ref}
\label{g2}
\end{figure}


\subsection{Detection of the total GRFs during robot locomotion on two different floors}


For this trial, the data is collected with the humanoid robot NAO walking on two different floors: laboratory floor and carpet. The total GRFs are divided into vertical ground reaction forces (vGRFs), and horizontal ground reaction forces (hGRFs).
The measured vGRFs are illustrated in Fig. \ref{g4}. In order to have a better perception of the measured values, only three random steps are shown. 

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{figure}[h]
\centering
 \begin{tikzpicture}
 \pgfplotsset{width=8cm,compat=newest}
    \begin{axis}[
    title={\textbf{vGRFs (locomotion)}},
    legend style={at={(axis cs:4,100)},anchor=south west},
    axis y line*= left,
    ylabel style={at={(ticklabel cs:0.2)}},
    xlabel={Time[s]},
    ylabel={Force[N]},
    xmin=5,xmax=10,
    ymin=-2,ymax=120,
    xtick={5,6,7,8,9,10},
    ytick={0, 10, 20, 30, 40, 50, 60},
    ]
    \addplot [thick, red] table[
    y]
    {
 5.82	-0.176169023
5.83	-0.179277145
5.84	5.427364807
5.85	8.367564922
5.86	12.16271244
5.87	11.53079093
5.88	16.92841119
5.89	14.89601406
5.9	16.03601325
5.91	15.5454227
5.92	13.78877064
5.93	12.43662335
5.94	10.9725295
5.95	10.29846184
5.96	10.16675068
5.97	9.592113088
5.98	9.824371373
5.99	10.01989185
6	9.459612803
6.01	9.310557609
6.02	9.482804253
6.03	10.42257165
6.04	10.79843115
6.05	13.41750368
6.06	14.00822754
6.07	16.40835051
6.08	18.94019421
6.09	27.02085278
6.1	39.02152088
6.11	53.07959916
6.12	55.92341992
6.13	55.88951307
6.14	46.48891417
6.15	37.95577873
6.16	36.70383771
6.17	35.02667931
6.18	38.41312419
6.19	40.45693513
6.2	42.40724885
6.21	43.61771676
6.22	44.41170921
6.23	39.51781095
6.24	38.02151164
6.25	32.51580461
6.26	34.91136462
6.27	33.54870681
6.28	33.69806177
6.29	36.16524526
6.3	38.14867687
6.31	36.20956337
6.32	38.80814298
6.33	41.26828251
6.34	39.41898368
6.35	36.88285166
6.36	42.47558076
6.37	42.30044891
6.38	43.25110832
6.39	44.28400485
6.4	44.08598651
6.41	44.85283429
6.42	43.10465747
6.43	44.14717612
6.44	40.05611081
6.45	43.61972971
6.46	39.67366293
6.47	30.02433938
6.48	24.48807588
6.49	22.40960021
6.5	25.94303422
6.51	26.42322534
6.52	27.30580163
6.53	27.35090254
6.54	28.85341746
6.55	30.65850962
6.56	31.36240354
6.57	31.84089088
6.58	31.7889217
6.59	29.05935173
6.6	31.2264142
6.61	28.60690846
6.62	28.39024663
6.63	25.33963314
6.64	25.37581133
6.65	20.98259669
6.66	19.46003174
6.67	14.37093163
6.68	6.193566048
6.69	2.575713875
6.7	0.607284186
6.71	-0.071455956
6.72	-0.211374189
6.73	-0.035902053
6.74	-0.240292505
6.75	-0.191377805
6.76	-0.171147946
6.77	-0.1747239
6.78	-0.162114899
6.79	-0.232678449
6.8	-0.253241889
6.81	-0.255723021
6.82	-0.245905839
6.83	-0.173015005
6.84	-0.117894759
6.85	-0.150213001
6.86	-0.11880999
6.87	-0.071497745
6.88	-0.115438987
6.89	-0.09519246
6.9	-0.161978498
6.91	-0.103607897
6.92	-0.093474531
6.93	-0.065835733
6.94	-0.070844132
6.95	-0.119319653
6.96	-0.094508908
6.97	-0.109145692
6.98	-0.101834489
6.99	-0.094508908
7	-0.088736543
7.01	-0.107902173
7.02	-0.013444211
7.03	0.065474996
7.04	10.25053207
7.05	10.41978657
7.06	15.04051436
7.07	16.13955716
7.08	15.86287387
7.09	18.22234325
7.1	17.78646105
7.11	15.87606149
7.12	13.72941964
7.13	11.6193728
7.14	9.074748282
7.15	7.492343944
7.16	6.212624364
7.17	5.260036531
7.18	4.94068093
7.19	4.69166343
7.2	4.79571352
7.21	4.68623631
7.22	4.90635646
7.23	7.569792225
7.24	9.847036333
7.25	12.90659464
7.26	16.19087659
7.27	24.45354707
7.28	38.60203305
7.29	51.27243736
7.3	54.63226863
7.31	54.8200214
7.32	44.46923912
7.33	42.13503954
7.34	35.66271342
7.35	35.06906886
7.36	36.40357294
7.37	39.24192213
7.38	43.71706094
7.39	41.80124698
7.4	41.24600945
7.41	40.50464726
7.42	38.14740244
7.43	32.85955019
7.44	33.07338076
7.45	33.08224117
7.46	33.10619273
7.47	33.87253099
7.48	34.82458517
7.49	35.73433859
7.5	36.69901205
7.51	37.69717563
7.52	38.98955407
7.53	38.48194155
7.54	39.14431599
7.55	40.15789995
7.56	40.48216625
7.57	42.47577061
7.58	38.10650288
7.59	39.57335795
7.6	41.85985825
7.61	41.4944703
7.62	41.03580295
7.63	41.32493307
7.64	40.96674772
7.65	37.73291006
7.66	31.78863848
7.67	31.01573723
7.68	30.19365812
7.69	33.26353954
7.7	33.02973534
7.71	31.96877765
7.72	31.47053466
7.73	30.92614423
7.74	30.81608798
7.75	30.90071512
7.76	29.64140979
7.77	28.51288544
7.78	30.25284723
7.79	28.4540088
7.8	25.52992944
7.81	22.53854586
7.82	19.1298808
7.83	15.44307509
7.84	12.8322114
7.85	10.54479651
7.86	7.475041018
7.87	1.897403666
7.88	0.401154273
7.89	-0.135786575
7.9	-0.133263131
7.91	-0.05981552
7.92	-0.137109273
7.93	-0.240782937
7.94	-0.14793391
7.95	-0.177794964
7.96	-0.139713481
7.97	-0.158617785
7.98	-0.235116464
7.99	-0.305463496
8	-0.264747154
8.01	-0.280308035
8.02	-0.113019594
8.03	-0.235424959
8.04	-0.206594024
8.05	-0.300672607
8.06	-0.207928088
8.07	-0.261466409
8.08	-0.208891771
8.09	-0.241478495
8.1	-0.294194109
8.11	-0.16959118
8.12	-0.125802019
8.13	-0.045555485
8.14	-0.069208743
8.15	-0.082669829
8.16	-0.089981032
8.17	-0.026252367
8.18	-0.047864928
8.19	0.024526373
8.2	0.026648006
8.21	0.008979374
8.22	1.357092039
8.23	6.295366433
8.24	7.131034762
8.25	8.322100806
8.26	9.826078301
8.27	10.49582443
8.28	10.42805638
8.29	11.83395566
8.3	10.40753829
8.31	9.822801003
8.32	8.58328158
8.33	7.587149555
8.34	7.81050596
8.35	7.032925526
8.36	6.386232703
8.37	6.921716755
8.38	7.582467898
8.39	6.772136798
8.4	7.42443403
8.41	8.072975421
8.42	9.641066597
8.43	10.96828538
8.44	13.85514413
8.45	14.83475364
8.46	20.14882475
8.47	27.68126666
8.48	40.16524171
8.49	51.83967632
8.5	52.52115188
8.51	54.97289172
8.52	42.27265066
8.53	42.21780348
8.54	34.47757595
8.55	39.92470483
8.56	38.34039816
8.57	40.08529464
8.58	40.94322132
8.59	38.8183477
8.6	39.8857952
8.61	37.23342311
8.62	41.77335177
8.63	34.44115107
8.64	33.99861899
8.65	32.635147
8.66	34.55951052
8.67	35.95382327
8.68	38.2148244
8.69	33.80650584
8.7	38.52863714
8.71	40.0149342
8.72	42.91190432
8.73	41.66083166
8.74	39.34216085
8.75	45.03175051
8.76	46.20950296
8.77	43.38321349
8.78	41.99225878
8.79	41.48264817
8.8	43.16669856
8.81	44.16472964
8.82	43.17210225
8.83	39.60673888
8.84	29.25647487
8.85	19.52952752
8.86	17.37125167
8.87	17.99820083
8.88	20.76363745
8.89	20.1409641
8.9	21.6362468
8.91	23.09725789
8.92	25.74865218
8.93	28.14903266
8.94	28.95502594
8.95	29.43116498
8.96	29.90051252
8.97	29.81100307
8.98	30.31450612
8.99	31.14056247
9	28.95077843
9.01	28.15244434
9.02	26.10617684
9.03	21.69686335
9.04	18.27775646
9.05	16.49958148
9.06	11.94291269
9.07	3.647106946
9.08	0.949909749
9.09	0.03710741
9.1	-0.251459703
9.11	0.146944653
9.12	0.104007922


    };
    \addplot [thick, blue] table[
    y]
    {

6.39	-1.248192771
6.4	-1.298795181
6.41	-1.265060241
6.42	5.195180723
6.43	26.44819277
6.44	25.06506024
6.45	29.18072289
6.46	28.79277108
6.47	28.42168675
6.48	28.18554217
6.49	27.42650602
6.5	26.36385542
6.51	25.2
6.52	24.79518072
6.53	22.97349398
6.54	21.52289157
6.55	21.64096386
6.56	22.38313253
6.57	24.66024096
6.58	25.31807229
6.59	26.0939759
6.6	25.73975904
6.61	27.02168675
6.62	28.55662651
6.63	30.05783133
6.64	32.48674699
6.65	34.89879518
6.66	37.2939759
6.67	39.57108434
6.68	41.4939759
6.69	40.98795181
6.7	42.23614458
6.71	39.77349398
6.72	38.76144578
6.73	38.45783133
6.74	39.09879518
6.75	39.99277108
6.76	44.00722892
6.77	47.00963855
6.78	48.88192771
6.79	49.05060241
6.8	48.71325301
6.81	48.12289157
6.82	48.51084337
6.83	49.01686747
6.84	48.30843373
6.85	47.98795181
6.86	49.13493976
6.87	50.21445783
6.88	49.3373494
6.89	48.30843373
6.9	49.13493976
6.91	50.21445783
6.92	50.65301205
6.93	49.99518072
6.94	50.01204819
6.95	50.63614458
6.96	51.19277108
6.97	51.1253012
6.98	50.18072289
6.99	49.94457831
7	50.41686747
7.01	50.51807229
7.02	48.12289157
7.03	42.1686747
7.04	37.58072289
7.05	35.37108434
7.06	37.85060241
7.07	38.72771084
7.08	39.41927711
7.09	38.94698795
7.1	40.97108434
7.11	42.18554217
7.12	43.51807229
7.13	43.60240964
7.14	42.94457831
7.15	42.8939759
7.16	42.91084337
7.17	42.75903614
7.18	43.16385542
7.19	42.06746988
7.2	42.6746988
7.21	42.42168675
7.22	44.04096386
7.23	42.91084337
7.24	41.94939759
7.25	37.66506024
7.26	26.36385542
7.27	15.18072289
7.28	1.939759036
7.29	0.219277108
7.3	0.05060241
7.31	-0.168674699
7.32	-0.657831325
7.33	-0.742168675
7.34	-1.079518072
7.35	-1.096385542
7.36	-1.163855422
7.37	-1.180722892
7.38	-0.185542169
7.39	-1.113253012
7.4	-0.691566265
7.41	-1.214457831
7.42	-0.775903614
7.43	-0.708433735
7.44	-0.742168675
7.45	-0.455421687
7.46	-0.539759036
7.47	-0.573493976
7.48	-0.539759036
7.49	-0.607228916
7.5	-0.708433735
7.51	-0.657831325
7.52	-0.640963855
7.53	-1.079518072
7.54	-0.944578313
7.55	-1.130120482
7.56	-1.130120482
7.57	-1.028915663
7.58	-1.231325301
7.59	-1.281927711
7.6	-1.315662651
7.61	-1.36626506
7.62	-1.33253012
7.63	8.079518072
7.64	16.58072289
7.65	15.5686747
7.66	16.90120482
7.67	17.44096386
7.68	17.96385542
7.69	18.33493976
7.7	18.26746988
7.71	17.79518072
7.72	17.6939759
7.73	17.55903614
7.74	17.66024096
7.75	17.66024096
7.76	18.14939759
7.77	19.16144578
7.78	21.26987952
7.79	24.67710843
7.8	26.14457831
7.81	29.46746988
7.82	31.74457831
7.83	34.35903614
7.84	35.37108434
7.85	37.20963855
7.86	37.59759036
7.87	38.81204819
7.88	40.0939759
7.89	41.39277108
7.9	40.17831325
7.91	40.63373494
7.92	38.77831325
7.93	37.85060241
7.94	37.47951807
7.95	38.6939759
7.96	40.63373494
7.97	46.94216867
7.98	48.05542169
7.99	48.02168675
8	49.48915663
8.01	48.8313253
8.02	48.15662651
8.03	47.7686747
8.04	47.73493976
8.05	48.74698795
8.06	49.38795181
8.07	49.96144578
8.08	49.11807229
8.09	48.61204819
8.1	48.98313253
8.11	50.55180723
8.12	49.86024096
8.13	48.8313253
8.14	49.45542169
8.15	49.82650602
8.16	50.18072289
8.17	49.99518072
8.18	49.97831325
8.19	48.57831325
8.2	48.40963855
8.21	48.84819277
8.22	49.25301205
8.23	45.84578313
8.24	41.96626506
8.25	40.95421687
8.26	42.25301205
8.27	41.40963855
8.28	41.34216867
8.29	41.08915663
8.3	41.20722892
8.31	40.97108434
8.32	42.10120482
8.33	42.23614458
8.34	41.71325301
8.35	41.74698795
8.36	41.35903614
8.37	42.57349398
8.38	42.32048193
8.39	41.74698795
8.4	40.63373494
8.41	41.24096386
8.42	40.78554217
8.43	39.01445783
8.44	37.71566265
8.45	32.7060241
8.46	23.22650602
8.47	13.86506024
8.48	1.585542169
8.49	0.404819277
8.5	0.05060241
8.51	-0.118072289
8.52	-0.624096386
8.53	-0.489156627
8.54	-1.214457831
8.55	-1.163855422
8.56	-1.028915663
8.57	-1.130120482
8.58	-1.248192771
8.59	-1.079518072
8.6	-1.012048193
8.61	-1.028915663
8.62	-1.214457831
8.63	-1.130120482
8.64	-1.079518072
8.65	-1.281927711
8.66	-1.028915663
8.67	-1.045783133
8.68	-0.961445783
8.69	-1.130120482
8.7	-1.062650602
8.71	-1.163855422
8.72	-1.146987952
8.73	-1.113253012
8.74	-1.248192771
8.75	-1.265060241
8.76	-1.298795181
8.77	-1.315662651
8.78	-1.315662651
8.79	-1.248192771
8.8	-1.265060241
8.81	-1.265060241
8.82	17.03614458
8.83	25.16626506
8.84	31.5253012
8.85	30.31084337
8.86	32.55421687
8.87	31.96385542
8.88	31.87951807
8.89	30.19277108
8.9	29.43373494
8.91	28.06746988
8.92	25.84096386
8.93	24.5253012
8.94	22.63614458
8.95	21.82650602
8.96	22.53493976
8.97	22.4
8.98	22.75421687
8.99	21.37108434
9	23.02409639
9.01	26.12771084
9.02	27.57831325
9.03	29.4
9.04	31.98072289
9.05	33.19518072
9.06	37.00722892
9.07	38.91325301
9.08	41.79759036
9.09	42.74216867
9.1	43.45060241
9.11	44.53012048
9.12	40.39759036
9.13	40.21204819
9.14	39.14939759
9.15	41.08915663
9.16	43.5686747
9.17	45.59277108
9.18	48.05542169
9.19	49.20240964
9.2	49.96144578
9.21	49.28674699
9.22	48.79759036
9.23	47.90361446
9.24	48.40963855
9.25	48.12289157
9.26	48.54457831
9.27	49.62409639
9.28	49.84337349
9.29	48.42650602
9.3	48.74698795
9.31	49.8939759
9.32	50.55180723
9.33	50.29879518
9.34	49.92771084
9.35	49.65783133
9.36	50.5686747
9.37	51.19277108
9.38	50.70361446
9.39	49.59036145
9.4	49.53975904
9.41	49.77590361
9.42	49.69156627
9.43	48.22409639
9.44	44.90120482
9.45	42.15180723
9.46	42.01686747
9.47	41.98313253
9.48	42.64096386
9.49	42.32048193
9.5	42.06746988
9.51	42.97831325
9.52	43.78795181
9.53	43.93975904
9.54	42.96144578
9.55	42.8939759
9.56	42.97831325
9.57	44.0746988
9.58	43.55180723
9.59	43.66987952
9.6	42.7253012
9.61	42.26987952
9.62	41.71325301
9.63	41.52771084
9.64	36.93975904
9.65	34.69638554
9.66	23.14216867
9.67	14.60722892
9.68	3.795180723
9.69	0.303614458
9.7	0.354216867
9.71	0.134939759



    };
    
   \addplot [thick, black] table[
    y]
    {
    0 57
    10 57
    };
      
    \addlegendentry{Left: $Fn_t$}
    \addlegendentry{Right: $Fn_t$}
  \node[black] at (axis cs: 9.75,110) {B};
 \node[black] at (axis cs: 5.3,5) {A};
\end{axis}
    
    \begin{axis}[
    axis y line*=right,
    axis x line=none,
    ylabel style={at={(ticklabel cs:0.75)}},
    ylabel={Force[N]},
    xmin=5,xmax=10,
    ymin=0,ymax=120,
    xtick={5,6,7,8,9,10},
    ytick={60,70, 80, 90 , 100 , 110,120},
    yticklabels={0,10,20,30,40,50,60},
    ]
     \addplot [thick, red] table[
    y]
    {
   5.77	60.05144362
5.78	60.09292297
5.79	60.07606098
5.8	60.03934647
5.81	60.19323334
5.82	60.22817498
5.83	62.25393255
5.84	73.129333
5.85	76.03240522
5.86	90.14125695
5.87	84.13702818
5.88	95.4614132
5.89	87.9618532
5.9	91.24349154
5.91	89.5046132
5.92	89.53876594
5.93	88.6708963
5.94	86.60601235
5.95	85.41827993
5.96	82.89088801
5.97	79.41186737
5.98	76.13618246
5.99	74.15011627
6	73.00372892
6.01	72.74275213
6.02	72.44435528
6.03	71.21062426
6.04	71.19136832
6.05	70.29326112
6.06	71.62525078
6.07	71.62820455
6.08	72.91169086
6.09	73.48009914
6.1	76.92328424
6.11	82.1462956
6.12	92.60056506
6.13	103.5871752
6.14	108.7232329
6.15	112.4574145
6.16	110.2685507
6.17	103.5538714
6.18	96.78661813
6.19	93.41138252
6.2	94.23268948
6.21	97.7933373
6.22	100.8950631
6.23	104.8480152
6.24	106.2390889
6.25	105.6063551
6.26	105.7376244
6.27	104.2408096
6.28	101.8538906
6.29	96.67636062
6.3	96.12727467
6.31	98.98182247
6.32	101.5549915
6.33	103.1907944
6.34	108.2025515
6.35	106.2393993
6.36	108.2087038
6.37	108.5812513
6.38	108.7599771
6.39	110.5998295
6.4	111.4893181
6.41	112.2069324
6.42	113.1523384
6.43	110.0538174
6.44	105.7339085
6.45	107.1320902
6.46	100.2061353
6.47	90.20569149
6.48	80.32768613
6.49	75.71761167
6.5	75.88289748
6.51	76.97748074
6.52	80.03057421
6.53	81.05197183
6.54	82.7173523
6.55	82.3144908
6.56	85.66669876
6.57	87.42032309
6.58	89.89257858
6.59	90.09545285
6.6	90.79429933
6.61	92.07143683
6.62	94.30931846
6.63	96.06063275
6.64	93.85087751
6.65	92.29580167
6.66	91.81227026
6.67	90.18995498
6.68	90.68103231
6.69	87.13359175
6.7	78.76166539
6.71	70.01772342
6.72	64.31144524
6.73	61.72934599
6.74	60.05123893
6.75	60.34782734
6.76	60.34046422
6.77	60.29616234
6.78	60.2067322
6.79	60.08255878
6.8	60.24826217
6.81	60.37154691
6.82	60.27028369
6.83	60.28419879
6.84	60.18424432
6.85	60.22116299
6.86	60.27937797
6.87	60.206149
6.88	60.45477676
6.89	60.35947156
6.9	60.43759636
6.91	60.32173683
6.92	60.37927075
6.93	60.39935815
6.94	60.1764479
6.95	60.18139481
6.96	60.20352186
6.97	60.24846806
6.98	60.10541815
6.99	60.11051346
7	60.20816832
7.01	63.50607898
7.02	70.45220454
7.03	90.75093733
7.04	97.67677723
7.05	106.1973006
7.06	112.5840624
7.07	104.0378945
7.08	95.61652931
7.09	86.67561293
7.1	86.69716536
7.11	90.35172038
7.12	99.91119992
7.13	106.2228937
7.14	108.5392313
7.15	96.22200894
7.16	85.84228924
7.17	81.43075164
7.18	81.51157683
7.19	82.1268517
7.2	78.51924069
7.21	74.29561352
7.22	70.59391109
7.23	70.1902226
7.24	68.57169602
7.25	68.77702702
7.26	67.55674097
7.27	66.15963478
7.28	64.75349771
7.29	64.57982879
7.3	64.58892803
7.31	66.70003085
7.32	74.45797739
7.33	93.9067631
7.34	105.1691067
7.35	114.3845509
7.36	106.0457612
7.37	103.8024332
7.38	101.8574059
7.39	98.6507879
7.4	96.52476316
7.41	96.33463423
7.42	96.96165836
7.43	101.3529454
7.44	103.00875
7.45	102.6456686
7.46	104.0040814
7.47	104.1293564
7.48	102.4575665
7.49	98.33832022
7.5	97.55234791
7.51	100.9109485
7.52	104.6152259
7.53	106.7258976
7.54	109.8957264
7.55	110.7770041
7.56	109.4577042
7.57	109.7888186
7.58	109.7742182
7.59	108.4562214
7.6	109.1845605
7.61	110.5454813
7.62	109.7579092
7.63	109.2862021
7.64	106.909468
7.65	101.5762225
7.66	91.76216804
7.67	77.5608035
7.68	72.19579757
7.69	69.75797356
7.7	71.25776505
7.71	76.63937382
7.72	77.62734008
7.73	79.95069731
7.74	80.42542229
7.75	82.60133342
7.76	84.07031559
7.77	88.37460884
7.78	89.6871026
7.79	93.1869908
7.8	93.81044872
7.81	94.80136898
7.82	97.59286333
7.83	96.74404649
7.84	95.09046725
7.85	94.9733254
7.86	93.89217068
7.87	94.34262194
7.88	94.20730668
7.89	91.73210036
7.9	89.94050408
7.91	74.12316228
7.92	68.33600191
7.93	62.50834901
7.94	60.4357139
7.95	60.03834411
7.96	59.93817332
7.97	60.32456825
7.98	60.45919626
7.99	60.2002632
8	60.32987067
8.01	59.9358475
8.02	60.38349783
8.03	60.02861303
8.04	60.1386226
8.05	60.07128607
8.06	60.06184
8.07	60.07159735
8.08	60.43006667
8.09	60.08220369
8.1	60.50309586
8.11	60.21918743
8.12	60.58027417
8.13	60.09774337
8.14	60.18836499
8.15	60.14179639
8.16	60.2701734
8.17	60.09529228
8.18	60.07814262
8.19	60.10738611
8.2	60.21814185
8.21	60.50811783
8.22	64.58365176
8.23	76.66227953
8.24	88.69455123
8.25	100.1807371
8.26	98.4097212
8.27	100.6476625
8.28	91.39913241
8.29	87.39697065
8.3	86.77646325
8.31	92.2340547
8.32	99.11138513
8.33	101.0066769
8.34	100.4161564
8.35	91.77515974
8.36	85.00629187
8.37	83.16176398
8.38	83.79441409
8.39	81.41949143
8.4	78.15948297
8.41	75.38887287
8.42	73.49867022
8.43	72.48565088
8.44	72.65071519
8.45	71.83098026
8.46	71.21579983
8.47	70.57557233
8.48	69.58351639
8.49	68.44920719
8.5	68.97998494
8.51	71.71583075
8.52	79.1039907
8.53	95.27103601
8.54	106.2118867
8.55	112.7480797
8.56	107.6201499
8.57	107.6538595
8.58	97.19130236
8.59	97.07101712
8.6	95.2575376
8.61	98.95565344
8.62	100.0459304
8.63	98.94915327
8.64	104.8267027
8.65	104.4201441
8.66	109.1042195
8.67	107.9879326
8.68	106.1167308
8.69	101.7216125
8.7	102.6791246
8.71	105.3656259
8.72	108.401842
8.73	109.3958764
8.74	113.187583
8.75	112.8659278
8.76	114.5919438
8.77	113.5962178
8.78	112.910364
8.79	113.1467691
8.8	112.6473746
8.81	111.6878754
8.82	111.2266877
8.83	109.5860367
8.84	104.9255223
8.85	99.06169172
8.86	86.29722696
8.87	77.67325006
8.88	71.90689209
8.89	70.90885733
8.9	70.88948467
8.91	74.51725126
8.92	75.79471981
8.93	79.56568465
8.94	81.97601022
8.95	84.06029788
8.96	85.51047968
8.97	89.77786019
8.98	91.99646999
8.99	94.81242988
9	95.44440732
9.01	97.09264276
9.02	97.91295907
9.03	101.1444302
9.04	100.2965444
9.05	98.21106
9.06	97.04372653
9.07	96.91543996
9.08	95.24175593
9.09	96.82333187
9.1	94.21852596
9.11	77.9817606
9.12	74.25598817
9.13	64.30193435
9.14	64.04663313
9.15	60.38858063
9.16	60.0627815
9.17	60.10670485
9.18	60.19323334
9.19	60.24400243
9.2	60.25439245
9.21	60.13499549
9.22	60.06155866
9.23	60.34996338
9.24	59.99584376
9.25	60.3855348
9.26	59.99624587
9.27	60.22055931
9.28	60.02835415
9.29	60.51131495
9.3	60.06927808
9.31	60.07387648
9.32	60.05193875
9.33	60.32210641
9.34	59.98450756
9.35	60.0563209
9.36	60.05418408
9.37	60.0224918

    
    };
    
     \addplot [thick, blue] table[
    y]
    {
6.27	59.68433735
6.28	59.7626506
6.29	59.7
6.3	59.77831325
6.31	59.6686747
6.32	59.7313253
6.33	59.6686747
6.34	59.8253012
6.35	59.68433735
6.36	59.7626506
6.37	59.77831325
6.38	59.87228916
6.39	62.53493976
6.4	66.46626506
6.41	70.30361446
6.42	75.50361446
6.43	82.23855422
6.44	87.84578313
6.45	87.61084337
6.46	88.11204819
6.47	85.49638554
6.48	86.70240964
6.49	88.83253012
6.5	91.65180723
6.51	91.10361446
6.52	88.08072289
6.53	85.19879518
6.54	83.0373494
6.55	81.25180723
6.56	80.23373494
6.57	78.90240964
6.58	76.86626506
6.59	76.4746988
6.6	75.81686747
6.61	74.65783133
6.62	73.9373494
6.63	73.95301205
6.64	74.34457831
6.65	75.09638554
6.66	76.00481928
6.67	76.6626506
6.68	78.79277108
6.69	82.83373494
6.7	91.43253012
6.71	106.1554217
6.72	110.9951807
6.73	111.026506
6.74	104.1819277
6.75	99.32650602
6.76	96.49156627
6.77	96.1313253
6.78	95.77108434
6.79	95.63012048
6.8	96.24096386
6.81	96.78915663
6.82	98.54337349
6.83	99.31084337
6.84	99.45180723
6.85	102.6156627
6.86	104.9493976
6.87	104.9650602
6.88	106.2180723
6.89	107.3144578
6.9	107.3614458
6.91	106.3590361
6.92	107.0638554
6.93	107.0638554
6.94	107.4554217
6.95	106.4373494
6.96	106.6096386
6.97	106.0771084
6.98	106.2493976
6.99	106.139759
7	105.4975904
7.01	103.2108434
7.02	97.47831325
7.03	87.67349398
7.04	77.24216867
7.05	72.07349398
7.06	68.07951807
7.07	67.29638554
7.08	67.20240964
7.09	68.50240964
7.1	70.66385542
7.11	74.21927711
7.12	75.75421687
7.13	76.25542169
7.14	78.85542169
7.15	83.85180723
7.16	92.02771084
7.17	97.4626506
7.18	98.7939759
7.19	96.47590361
7.2	95.06626506
7.21	92.1686747
7.22	91.54216867
7.23	90.46144578
7.24	89.41204819
7.25	89.06746988
7.26	87.98674699
7.27	88.56626506
7.28	87.20361446
7.29	85.13614458
7.3	82.20722892
7.31	78.46385542
7.32	73.37349398
7.33	69.61445783
7.34	63.67831325
7.35	62.70722892
7.36	60.9373494
7.37	60.79638554
7.38	60.15421687
7.39	59.7939759
7.4	59.7939759
7.41	59.74698795
7.42	59.87228916
7.43	59.7626506
7.44	59.80963855
7.45	59.7313253
7.46	59.7939759
7.47	59.74698795
7.48	59.74698795
7.49	59.7313253
7.5	59.71566265
7.51	59.74698795
7.52	59.71566265
7.53	59.74698795
7.54	59.7313253
7.55	59.74698795
7.56	59.77831325
7.57	59.87228916
7.58	62.22168675
7.59	64.71204819
7.6	68.98795181
7.61	73.15421687
7.62	77.9939759
7.63	82.84939759
7.64	87.45421687
7.65	87.86144578
7.66	88.48795181
7.67	86.67108434
7.68	86.73373494
7.69	88.89518072
7.7	90.25783133
7.71	89.5373494
7.72	88.2373494
7.73	86.21686747
7.74	83.17831325
7.75	81.42409639
7.76	78.94939759
7.77	77.72771084
7.78	76.16144578
7.79	75.62891566
7.8	74.01566265
7.81	73.01325301
7.82	73.09156627
7.83	72.74698795
7.84	72.71566265
7.85	72.8253012
7.86	73.46746988
7.87	73.85903614
7.88	74.84578313
7.89	76.14578313
7.9	82.66144578
7.91	98.52771084
7.92	110.6349398
7.93	112.5457831
7.94	110.0710843
7.95	102.9289157
7.96	97.50963855
7.97	96.24096386
7.98	96.38192771
7.99	97.00843373
8	99.54578313
8.01	99.67108434
8.02	98.84096386
8.03	99.37349398
8.04	100.046988
8.05	105.0277108
8.06	105.3722892
8.07	106.2024096
8.08	105.3253012
8.09	106.4216867
8.1	107.5180723
8.11	107.2361446
8.12	106.5156627
8.13	107.1421687
8.14	107.5963855
8.15	107.3614458
8.16	106.2807229
8.17	106.2963855
8.18	105.3879518
8.19	104.7771084
8.2	103.0542169
8.21	101.8638554
8.22	94.47108434
8.23	83.1626506
8.24	74.34457831
8.25	70.24096386
8.26	66.99879518
8.27	65.15060241
8.28	65.00963855
8.29	64.66506024
8.3	64.30481928
8.31	67.5313253
8.32	75.44096386
8.33	75.50361446
8.34	76.4746988
8.35	81.58072289
8.36	87.36024096
8.37	94.62771084
8.38	99.06024096
8.39	99.24819277
8.4	93.64096386
8.41	92.56024096
8.42	90.82168675
8.43	89.83493976
8.44	88.75421687
8.45	87.46987952
8.46	87.37590361
8.47	87.31325301
8.48	87.46987952
8.49	85.88795181
8.5	83.97710843
8.51	79.98313253
8.52	76.11445783
8.53	69.12891566
8.54	66.30963855
8.55	63.45903614
8.56	61.97108434
8.57	61.59518072
8.58	61.68915663
8.59	60.53012048
8.6	60.51445783
8.61	60.59277108
8.62	60.23253012
8.63	59.7313253
8.64	59.98192771
8.65	59.7313253
8.66	59.90361446
8.67	59.7939759
8.68	59.90361446
8.69	59.74698795
8.7	59.95060241
8.71	59.96626506
8.72	59.91927711
8.73	59.7626506
8.74	59.74698795
8.75	59.74698795
8.76	59.7626506
8.77	59.91927711
8.78	62.94216867
8.79	65.93373494
8.8	68.8
8.81	74.83012048
8.82	77.74337349
8.83	81.8939759
8.84	84.4313253
8.85	85.7626506
8.86	85.80963855
8.87	85.74698795
8.88	85.87228916
8.89	86.38915663
8.9	87.51686747
8.91	87.0939759
8.92	84.99518072
8.93	82.80240964
8.94	79.65421687
8.95	78.57349398
8.96	77.32048193
8.97	75.50361446
8.98	73.74939759
8.99	73.04457831
9	72.23012048
9.01	71.21204819
9.02	70.69518072
9.03	70.44457831
9.04	70.44457831
9.05	70.91445783
9.06	71.18072289
9.07	71.79156627
9.08	72.30843373
9.09	74.89277108
9.1	81.47108434
9.11	97.60361446
9.12	109.8204819
9.13	112.3421687
9.14	109.4915663
9.15	105.1843373
9.16	97.85421687
9.17	95.58313253
9.18	95.78674699
9.19	96.72650602
9.2	99.78072289
9.21	99.15421687
9.22	96.85180723
9.23	98.84096386
9.24	102.1771084
9.25	103.1168675
9.26	105.5445783
9.27	105.2939759
9.28	105.6542169
9.29	106.813253
9.3	108.0506024
9.31	107.0481928
9.32	106.6253012
9.33	107.0325301
9.34	107.7373494
9.35	107.5024096
9.36	106.8915663
9.37	106.813253
9.38	107.0012048
9.39	106.4843373
9.4	105.3722892
9.41	104.0409639
9.42	100.2349398
9.43	88.86385542
9.44	75.92650602
9.45	71.07108434
9.46	68.14216867
9.47	67.4373494
9.48	66.65421687
9.49	67.20240964
9.5	67.4373494
9.51	70.47590361
9.52	73.84337349
9.53	73.42048193
9.54	74.95542169
9.55	79.85783133
9.56	85.37108434
9.57	93.23373494
9.58	96.47590361
9.59	97.07108434
9.6	93.01445783
9.61	89.25542169
9.62	89.5686747
9.63	88.2373494
9.64	87.93975904
9.65	87.34457831
9.66	87.1253012
9.67	86.98433735
9.68	87.1253012
9.69	85.5746988
9.7	84.85421687
9.71	81.65903614
9.72	78.47951807
9.73	73.21686747
9.74	69.39518072
9.75	63.27108434
9.76	62.86385542
9.77	61.59518072
9.78	60.79638554
    }; 
 
\end{axis}
\end{tikzpicture}
\caption{vGRFs for a walking humanoid robot on two different grounds. Excerpt of 3 steps for each ITshoe. A--laboratory ground; B--carpet.  }%ref}
\label{g4}
\end{figure}


It is noticeable that the step force patterns  from grounds A and B are distinct. On the other hand, for each ITshoe (left and right) and type of ground (A and B), the step patterns are visually quite similar among them. These observations are very interesting since they open the way for automatic detection of grounds based solely on the reaction force patterns. If precise absolute values for each foot are of concern, it may be required to have an additional calibration process by means of a ground truth source (e.g., a force-sensing platform).

The hGRFs can be represented in the sagittal and in the transverse axis as depicted in Fig. \ref{fig:sta12}. Since the horizontal sensors are strategically positioned at \ang{45}, it is quite simple to write the equations that represent the forces in these axis. The total horizontal force in the sagittal and transverse plane can be calculated as follows:
%
\begin{equation}
Fh_{st} = \frac{\sqrt2}{2} \left[(S1 + S2) - (S3+S4)\right] 
\end{equation}


\begin{figure}[h]
\centering
  \includegraphics[width=0.40\textwidth, angle = -00]{TTTT}
\caption{hGRFs axis detail. S1\_45 to S4\_45 represent the four horizontal force sensors.}
\label{fig:sta12}
\end{figure}

%
\begin{equation}
Fh_{st} = \frac{\sqrt2}{2} \left[(S2 + S4) - (S1+S3)\right] 
\end{equation}
%
where $Fh_{st}$ is the total horizontal force in the sagittal plane, $Fh_{tt}$ is the total horizontal force in the transverse plane, and S1 to S4 are the sensors used to measure the tangential forces.


Fig. \ref{g6} and Fig. \ref{g7} illustrate the hGRFs. The data corresponding to the moment when the foot is in the air (swing foot) was removed from the plots. 

\begin{figure}[h]
\centering
 \begin{tikzpicture}
 \pgfplotsset{width=8cm,compat=newest}
    \begin{axis}[
    title={\textbf{hGRFs sagittal plane (locomotion)}},
      legend style={at={(axis cs:4,9.5)},anchor=south west},
    xlabel={Time[s]},
    ylabel={Force[N]},
    xmin=5,xmax=10,
    ymin=-3,ymax=12,
    xtick={5,6,7,8,9,10},
    ytick={ -4,-3, -2,-1, 0 ,1 ,2, 3, 4}
    ]
    \addplot [thick, red] table[
    y]
    {
5.88	-1.537372615
5.89	-0.464333876
5.9	-1.196173634
5.91	-1.09631297
5.92	-1.139375527
5.93	-1.469652637
5.94	-1.087334579
5.95	-1.145871377
5.96	-1.009538791
5.97	-0.906378306
5.98	-0.996720013
5.99	-1.02502423
6	-0.996999159
6.01	-0.630637309
6.02	-0.474543564
6.03	-0.088165755
6.04	0.214399965
6.05	-0.319487782
6.06	0.241459742
6.07	0.285934696
6.08	0.474789936
6.09	0.659026423
6.1	-0.548246944
6.11	0.226794874
6.12	-0.22361342
6.13	-0.321849182
6.14	-0.621011566
6.15	-1.303871742
6.16	-1.561094854
6.17	-1.228823316
6.18	-1.686511461
6.19	-1.309410018
6.2	-1.359728163
6.21	-1.377748078
6.22	-2.29980982
6.23	-1.375405823
6.24	-2.499867522
6.25	-1.293152361
6.26	-2.023284764
6.27	-0.641239576
6.28	-0.148108637
6.29	1.13058139
6.3	1.538908377
6.31	1.751201835
6.32	1.666390213
6.33	1.970507313
6.34	1.319429418
6.35	2.081577126
6.36	2.074585829
6.37	2.229614864
6.38	2.475166711
6.39	2.61088575
6.4	2.675114008
6.41	2.739529712
6.42	3.068929958
6.43	3.198178519
6.44	3.249704412
6.45	3.089042602
6.46	3.288208113
6.47	3.604714492
6.48	3.739892964
6.49	3.35864294
6.5	3.365522717
6.51	3.184432948
6.52	3.230057384
6.53	3.372034797
6.54	3.249201512
6.55	3.274527689
6.56	3.383922483
6.57	3.254856856
6.58	3.332575466
6.59	3.552346578
6.6	3.196478648
6.61	3.558530717
6.62	3.371107622
6.63	3.344173488
6.64	3.000182586
6.65	3.007122597
6.66	2.832088642
6.67	2.845177216
6.68	2.801865102
6.69	2.899084878
6.7	2.543580012


    };
    
      \addplot [thick, blue] table[
    y]
    {
    6.32	-1.354130857
6.33	-1.833877009
6.34	-1.729395044
6.35	-1.644644018
6.36	-2.054504099
6.37	-2.007660434
6.38	-2.089945505
6.39	-2.065152953
6.4	-2.277595652
6.41	-2.140966183
6.42	-2.39712828
6.43	-2.307342476
6.44	-1.973653109
6.45	-2.274417713
6.46	-2.42785794
6.47	-2.495342489
6.48	-2.686663241
6.49	-1.935718913
6.5	-2.279373475
6.51	-2.499967604
6.52	-2.053939548
6.53	-2.199174674
6.54	-2.310812139
6.55	-2.260396104
6.56	-1.732036922
6.57	-1.985405398
6.58	-1.584041979
6.59	-0.099990053
6.6	0.26408439
6.61	0.510576645
6.62	1.232582878
6.63	0.949637212
6.64	1.393534597
6.65	1.489786116
6.66	1.565130535
6.67	1.695305048
6.68	1.907450061
6.69	2.124117318
6.7	2.329513513
6.71	2.217280675
6.72	2.460110784
6.73	2.902302587
6.74	2.634782827
6.75	2.842975623
6.76	2.590913014
6.77	2.024846765
6.78	1.464785909
6.79	1.524738257
6.8	1.322102721
6.81	1.181217224
6.82	1.209028193
6.83	0.956682482
6.84	0.722090435
6.85	1.169099836
6.86	0.265738047
6.87	0.225814325
6.88	-0.034599165
6.89	0.295142621
6.9	0.614110481
6.91	1.232425102
6.92	0.74831856
6.93	1.214333822
6.94	0.620837099
6.95	0.672188424
6.96	0.694364607
6.97	0.587010007
6.98	1.045868278
6.99	0.701153354
7	0.904080451
7.01	0.981930752
7.02	1.122513928
7.03	1.145869417
7.04	1.175163505
7.05	0.953111603
7.06	1.123970211
7.07	1.176888305
7.08	1.256906874
7.09	1.425978552
7.1	1.67131072
7.11	1.585228483
7.12	1.590775794
7.13	1.282431443
7.14	0.606998024

     };
     
    \addplot [thick, red] table[
    y]
    {
    7.05	1.877611889
7.06	1.612182262
7.07	0.648614502
7.08	1.675076411
7.09	0.847617439
7.1	1.029372423
7.11	0.512162142
7.12	0.611292917
7.13	0.644532042
7.14	0.621032716
7.15	0.853617452
7.16	0.601715246
7.17	0.747511133
7.18	0.754139766
7.19	0.947686242
7.2	0.746309373
7.21	1.286591686
7.22	1.623558798
7.23	1.440709674
7.24	1.362642
7.25	1.524867975
7.26	1.437983129
7.27	1.325503286
7.28	1.068820345
7.29	1.183915169
7.3	0.927567291
7.31	0.652794468
7.32	-0.896143468
7.33	-1.459719254
7.34	-1.193428801
7.35	-1.920894713
7.36	-2.371717936
7.37	-1.853760962
7.38	-2.634036932
7.39	-2.400114766
7.4	-2.367687759
7.41	-2.760547697
7.42	-2.727460229
7.43	-2.356615786
7.44	-2.28492095
7.45	-2.380959464
7.46	-1.563419065
7.47	0.032000979
7.48	0.57483828
7.49	0.825048972
7.5	1.226189835
7.51	1.323779676
7.52	1.874305007
7.53	1.769606028
7.54	2.28149162
7.55	2.065362271
7.56	2.408986693
7.57	2.285193739
7.58	2.530492734
7.59	2.58081565
7.6	2.404616142
7.61	2.505050464
7.62	2.601570162
7.63	2.49181859
7.64	2.511513227
7.65	2.698608119
7.66	3.106780887
7.67	3.16014025
7.68	3.354921483
7.69	3.089034564
7.7	3.120506099
7.71	3.18042698
7.72	2.828180091
7.73	2.652668366
7.74	2.737612873
7.75	2.276311879
7.76	2.665741704
7.77	2.880484026
7.78	2.672516961
7.79	2.55560058
7.8	2.626890314
7.81	2.236254568
7.82	2.614763498
7.83	2.830936677
7.84	3.183830983
7.85	3.248611701
7.86	2.292800703
7.87	2.79043786

     };
     
      \addplot [thick, red] table[
    y]
    {
    8.26	0.686710536
8.27	1.14528596
8.28	0.844908857
8.29	0.380919671
8.3	0.919695603
8.31	0.305433504
8.32	0.294249571
8.33	0.558954375
8.34	0.104652059
8.35	0.217779059
8.36	0.714401658
8.37	0.708653566
8.38	0.793797811
8.39	1.244693301
8.4	0.9938684
8.41	1.297793405
8.42	1.431784015
8.43	1.509360439
8.44	1.541609823
8.45	1.748218507
8.46	1.753408115
8.47	1.387656614
8.48	1.558762785
8.49	0.75950365
8.5	1.024974149
8.51	-0.067300171
8.52	-0.477895743
8.53	-1.208883556
8.54	-0.675898644
8.55	-1.458225498
8.56	-0.95874324
8.57	-1.325149635
8.58	-1.63982843
8.59	-1.328157824
8.6	-1.747799054
8.61	-1.041933029
8.62	-1.718592842
8.63	-0.673876418
8.64	-1.075810838
8.65	-0.563098795
8.66	-0.476745565
8.67	1.416881675
8.68	1.285702108
8.69	1.970948429
8.7	1.881617676
8.71	2.210837077
8.72	2.256182035
8.73	2.462536632
8.74	2.610179072
8.75	2.655498258
8.76	2.765861689
8.77	2.997944501
8.78	3.010156331
8.79	3.281925813
8.8	3.287870701
8.81	3.462167676
8.82	3.36538689
8.83	3.326690572
8.84	3.311782651
8.85	3.602747285
8.86	3.810907489
8.87	3.060827457
8.88	3.520890751
8.89	3.178049331
8.9	3.430195209
8.91	3.488149264
8.92	3.391045307
8.93	3.487642185
8.94	3.423168075
8.95	3.520285893
8.96	3.552783357
8.97	3.714131002
8.98	3.681808496
8.99	3.7398239
9	3.681292588
9.01	3.53252435
9.02	3.55761704
9.03	3.240930671
9.04	3.363642493
9.05	3.188285246
9.06	3.176818753
9.07	3.08043241
9.08	2.65194938

     };
  
 
     
      \addplot [thick, blue] table[
    y]
    {
    7.53	-1.350273894
7.54	-1.494329714
7.55	-1.387226745
7.56	-1.580715764
7.57	-1.446631265
7.58	-1.568485556
7.59	-1.3610508
7.6	-1.640674237
7.61	-1.593026477
7.62	-1.687168035
7.63	-1.749116034
7.64	-1.76281014
7.65	-1.736878211
7.66	-1.471475998
7.67	-1.459875778
7.68	-1.723590787
7.69	-1.984635755
7.7	-1.910304153
7.71	-1.911510294
7.72	-1.734410766
7.73	-1.930087272
7.74	-2.007916864
7.75	-2.359353769
7.76	-1.177250163
7.77	-1.803104602
7.78	-1.499538762
7.79	0.136486317
7.8	1.044255198
7.81	0.48898125
7.82	1.236715326
7.83	1.99913247
7.84	1.774577018
7.85	2.072539543
7.86	1.854231136
7.87	2.113753057
7.88	2.001249368
7.89	2.213257059
7.9	2.534845924
7.91	2.650094198
7.92	2.6765279
7.93	2.663031755
7.94	2.999799334
7.95	3.548799401
7.96	3.419137421
7.97	2.812659419
7.98	2.240839439
7.99	1.880902111
8	1.920338644
8.01	2.046990666
8.02	1.85207071
8.03	1.57485343
8.04	1.872693217
8.05	1.700517981
8.06	1.561800751
8.07	1.092483971
8.08	1.228532651
8.09	1.458906266
8.1	1.706092128
8.11	2.050160989
8.12	2.369943704
8.13	1.899645579
8.14	2.068975014
8.15	2.10061462
8.16	2.07536784
8.17	2.005491717
8.18	1.951471842
8.19	2.293853497
8.2	1.988568982
8.21	2.14123195
8.22	2.334120961
8.23	1.986777417
8.24	2.549061084
8.25	2.536756496
8.26	2.346550619
8.27	2.383718317
8.28	2.501800788
8.29	2.275103904
8.3	3.085811046
8.31	2.501440972
8.32	2.924366824
8.33	2.429343753
8.34	1.249602582
8.35	1.458940881

     };
     
      \addplot [thick, blue] table[
    y]
    {
    8.73	-1.085723321
8.74	-1.391937001
8.75	-1.471053241
8.76	-1.852490092
8.77	-1.710852681
8.78	-1.691952672
8.79	-1.603070688
8.8	-1.540218869
8.81	-1.623005714
8.82	-1.573768985
8.83	-1.623485286
8.84	-1.696629967
8.85	-1.824473547
8.86	-1.883976302
8.87	-2.027195066
8.88	-2.148876742
8.89	-1.819516294
8.9	-1.847860506
8.91	-2.054437498
8.92	-2.127357382
8.93	-2.116581967
8.94	-2.172387282
8.95	-2.446276207
8.96	-1.972813885
8.97	-2.111103722
8.98	-1.572220757
8.99	-1.794905627
9	-2.19088359
9.01	-1.515174068
9.02	0.15937617
9.03	0.31330971
9.04	0.859599808
9.05	0.899548875
9.06	0.896865741
9.07	0.892751669
9.08	1.063568858
9.09	1.153254935
9.1	1.132264184
9.11	1.480070414
9.12	1.618485322
9.13	1.731224568
9.14	2.285112142
9.15	2.801189779
9.16	2.035076686
9.17	2.132592653
9.18	0.990099038
9.19	1.319669729
9.2	1.04079208
9.21	1.00229317
9.22	0.565749247
9.23	0.450941459
9.24	0.697818875
9.25	0.788962726
9.26	0.410967047
9.27	-0.097713376
9.28	0.019104647
9.29	0.566481941
9.3	1.02638748
9.31	1.402590102
9.32	1.759778406
9.33	1.778982227
9.34	1.695747281
9.35	1.51282959
9.36	1.258400753
9.37	1.12262905
9.38	1.18225224
9.39	1.461750607
9.4	1.435370575
9.41	1.257356467
9.42	1.301504747
9.43	1.241725015
9.44	1.252144438
9.45	1.267598639
9.46	1.382025089
9.47	1.504431843
9.48	2.269601187
9.49	2.410274139
9.5	1.876069959
9.51	2.28581573
9.52	1.986770643
9.53	1.532141565
9.54	1.175047674
9.55	0.049967805

     };
     
  \addplot [thick, black] table[
    y]
    {
    0 4.3
    10 4.3
    };
  
     
  
     
     
     
    \addlegendentry{Left: $Fh_{st}$}
    \addlegendentry{Right: $Fh_{st}$}
\node[black] at (axis cs: 9.75,11) {B};
 \node[black] at (axis cs: 5.3,-2) {A};


\end{axis}

\begin{axis}[
    axis y line*=right,
    axis x line=none,
    ylabel style={at={(ticklabel cs:0.75)}},
    ylabel={Force[N]},
    xmin=5,xmax=10,
    ymin=-3,ymax=12,
    xtick={5,6,7,8,9,10},
    ytick={5,6,7,8,9,10,11,12},
    yticklabels={-3, -2, -1 ,0,1,2,3,4},
    ]
    
     
    \addplot [thick, red] table[
    y]
    {
    5.83	6.664923072
5.84	7.667849149
5.85	7.879412063
5.86	7.492168708
5.87	7.033653537
5.88	6.946687213
5.89	6.510003484
5.9	6.154089913
5.91	6.456513168
5.92	6.199712503
5.93	6.687118498
5.94	6.212552278
5.95	5.943633871
5.96	6.83466861
5.97	6.749914491
5.98	6.316099777
5.99	6.095286192
6	6.374125839
6.01	6.82847902
6.02	7.1085684
6.03	7.087779865
6.04	7.412812759
6.05	7.251149466
6.06	6.992586885
6.07	8.722822828
6.08	8.345471598
6.09	8.227157794
6.1	8.090483859
6.11	8.201340489
6.12	7.575089095
6.13	7.697308765
6.14	6.587595185
6.15	7.288466241
6.16	5.273299978
6.17	6.45252996
6.18	6.601458423
6.19	7.14585233
6.2	7.102753805
6.21	6.791383085
6.22	7.037668097
6.23	6.869223279
6.24	6.966259861
6.25	6.895435029
6.26	6.870658647
6.27	7.15683564
6.28	7.482679016
6.29	7.715855706
6.3	8.040153328
6.31	8.34704035
6.32	8.912996795
6.33	9.792780463
6.34	10.38459884
6.35	10.53944252
6.36	9.898578923
6.37	10.52696829
6.38	10.59141846
6.39	10.70771291
6.4	10.5465346
6.41	10.79761438
6.42	10.49418912
6.43	10.32107309
6.44	9.925454217
6.45	9.596288189
6.46	9.363265596
6.47	9.241824108
6.48	9.047615481
6.49	8.737486712
6.5	8.867525275
6.51	8.854371555
6.52	8.828156863
6.53	8.94306788
6.54	9.03923289
6.55	9.090054055
6.56	9.141366349
6.57	9.442516849
6.58	9.406517461
6.59	9.795337656
6.6	9.836549678
6.61	9.605308613
6.62	9.657695421
6.63	9.527580793
6.64	9.239608137
6.65	8.714388453
6.66	8.55925046
6.67	8.072490816
6.68	8.013717903
6.69	7.199532451
6.7	7.227765273
6.71	7.71850929
6.72	8.477815411
6.73	8.970537872
6.74	9.35691695
6.75	9.49273317

    };
    
    \addplot [thick, blue] table[
    y]
    {
    6.32	9.249377759
6.33	10.05767232
6.34	8.153496744
6.35	9.420206113
6.36	6.340558897
6.37	7.498827517
6.38	6.34468375
6.39	7.21534043
6.4	8.060328621
6.41	7.375558115
6.42	7.962507709
6.43	7.33165221
6.44	7.957985941
6.45	6.918276802
6.46	7.334988289
6.47	6.914022092
6.48	7.148352233
6.49	7.450578965
6.5	7.502974803
6.51	7.679434955
6.52	7.565827951
6.53	7.376442006
6.54	7.546627997
6.55	7.469338055
6.56	7.756452782
6.57	7.65049996
6.58	7.833972878
6.59	7.70402315
6.6	8.007333286
6.61	8.61885163
6.62	8.477574143
6.63	6.320744234
6.64	6.711703311
6.65	6.95233017
6.66	7.198652043
6.67	7.380200709
6.68	7.472840977
6.69	7.186401461
6.7	7.453051681
6.71	7.0390419
6.72	7.184053504
6.73	7.963952207
6.74	8.141826174
6.75	8.514251217
6.76	8.051649289
6.77	8.584648095
6.78	7.847368229
6.79	9.364902835
6.8	8.612016862
6.81	9.144013116
6.82	9.056458251
6.83	10.31916558
6.84	10.03822783
6.85	10.02219918
6.86	9.704094226
6.87	9.939601847
6.88	8.630231722
6.89	9.572192633
6.9	8.357121093
6.91	9.203411565
6.92	8.880538395
6.93	10.12757315
6.94	10.98163249
6.95	9.105149912
6.96	7.267045805
6.97	7.163601439
6.98	6.912673877
6.99	7.287521467
7	7.336833895
7.01	6.976776025
7.02	7.414822427
7.03	7.701200537
7.04	7.923482678
7.05	8.097826192
7.06	8.258176845
7.07	7.936039169
7.08	6.747175291
7.09	6.333055099
7.1	6.991859375
7.11	7.849495063
7.12	8.091507001
7.13	7.687677357
7.14	7.531965064
7.15	6.978904888
7.16	7.477761607
7.17	6.958899866
7.18	6.669571314
7.19	6.529551037
7.2	7.380963815
7.21	7.574688391
7.22	7.900128353
7.23	6.979225144
7.24	6.724303749
7.25	7.532286403
7.26	8.984729491

    };
    \addplot [thick, red] table[
    y]
    {
    7.03	6.566267868
7.04	8.516613997
7.05	7.724231232
7.06	7.202675139
7.07	7.088156877
7.08	6.936081754
7.09	6.181025333
7.1	6.21958497
7.11	6.380179163
7.12	6.167494305
7.13	6.647997173
7.14	6.343742673
7.15	6.034318752
7.16	6.159055198
7.17	6.145633911
7.18	5.737746436
7.19	5.640471718
7.2	6.171366286
7.21	6.774872558
7.22	6.86175876
7.23	6.745161419
7.24	6.815963822
7.25	7.2392738
7.26	7.42272303
7.27	7.390275143
7.28	7.404558979
7.29	7.587766868
7.3	7.490959474
7.31	7.70998678
7.32	6.660088858
7.33	6.500460943
7.34	5.956156811
7.35	6.421928356
7.36	5.228620514
7.37	6.137329839
7.38	6.266904493
7.39	6.725026522
7.4	6.841001467
7.41	7.151589165
7.42	6.956888314
7.43	6.814448508
7.44	6.762202911
7.45	6.820903035
7.46	6.671679264
7.47	6.996461988
7.48	7.10217716
7.49	7.419218339
7.5	7.640366394
7.51	6.762547386
7.52	8.000290964
7.53	8.970198865
7.54	9.814774122
7.55	9.445241094
7.56	9.795385639
7.57	9.93067281
7.58	10.13694629
7.59	9.314893672
7.6	10.48311448
7.61	10.05648883
7.62	10.78132638
7.63	9.381987475
7.64	9.71834323
7.65	8.920188039
7.66	8.667117948
7.67	8.648253681
7.68	8.943009583
7.69	8.704995987
7.7	8.40560091
7.71	8.146451201
7.72	8.163376244
7.73	7.815195114
7.74	7.779057077
7.75	7.490584222
7.76	7.327016379
7.77	7.556226326
7.78	7.416252174
7.79	7.48314224
7.8	7.669313894
7.81	7.712752001
7.82	7.71147087
7.83	7.426052501
7.84	7.555413556
7.85	7.34213772
7.86	7.174297622
7.87	7.329654005
7.88	6.637342272
7.89	6.947751616
7.9	6.223505644
7.91	6.734489813
7.92	7.193596502
7.93	8.348007391
7.94	8.566022176
7.95	9.447983604

    };
    \addplot [thick, red] table[
    y]
    {8.23	6.161094441
8.24	6.077734886
8.25	7.46619567
8.26	6.008880592
8.27	6.650875802
8.28	6.455525412
8.29	5.901412356
8.3	6.120412129
8.31	6.003909958
8.32	5.681288051
8.33	5.83160093
8.34	6.396529646
8.35	6.599826323
8.36	6.800005591
8.37	6.58683786
8.38	6.282116162
8.39	5.990760306
8.4	6.2621476
8.41	6.303964544
8.42	6.57646059
8.43	6.344120537
8.44	6.228949142
8.45	6.041948755
8.46	6.140035731
8.47	6.284126091
8.48	7.163861295
8.49	7.060856876
8.5	6.916558802
8.51	7.097745581
8.52	6.18578983
8.53	5.418273352
8.54	4.915865832
8.55	4.798461787
8.56	5.117914898
8.57	5.450100688
8.58	5.943982405
8.59	6.923499985
8.6	7.318075064
8.61	7.4987858
8.62	7.232315878
8.63	7.297626243
8.64	7.544059084
8.65	6.799271577
8.66	7.501353986
8.67	6.91840903
8.68	7.626217936
8.69	7.166300831
8.7	6.469069874
8.71	6.934631405
8.72	7.568811544
8.73	7.842230115
8.74	9.43049596
8.75	9.27463518
8.76	9.371087643
8.77	9.468547785
8.78	9.733911796
8.79	9.339131406
8.8	9.786611811
8.81	9.100426509
8.82	10.13186571
8.83	8.966106008
8.84	9.251246013
8.85	7.8373497
8.86	8.627357253
8.87	7.85820303
8.88	8.45678457
8.89	8.64569907
8.9	8.578993231
8.91	8.093886824
8.92	8.565108297
8.93	8.280498065
8.94	8.017339596
8.95	8.13309643
8.96	8.42876569
8.97	8.397366302
8.98	8.387760483
8.99	8.444348513
9	8.154257496
9.01	8.093797878
9.02	8.061121537
9.03	7.91768732
9.04	7.796566059
9.05	7.887171796
9.06	7.594579669
9.07	7.421395866
9.08	7.200636369
9.09	7.188135117
9.1	6.438991301
9.11	6.940297104
9.12	7.634923121
9.13	7.769540228
9.14	8.064155817
9.15	8.884252625

    
    };
    \addplot [thick, blue] table[
    y]
    {
    7.52	8.793152278
7.53	9.087198959
7.54	7.568727182
7.55	6.80318334
7.56	6.050924436
7.57	6.423969505
7.58	6.095624131
7.59	7.476241314
7.6	7.203052944
7.61	7.760134566
7.62	8.145169338
7.63	7.377001489
7.64	7.649923924
7.65	7.015846991
7.66	6.992322754
7.67	7.060561033
7.68	7.393964197
7.69	7.680612435
7.7	7.611880054
7.71	7.569280663
7.72	7.419400544
7.73	7.635296949
7.74	7.643987477
7.75	7.732578688
7.76	7.863566748
7.77	7.479822978
7.78	7.811742754
7.79	7.396127894
7.8	7.598395683
7.81	8.210283329
7.82	8.570567379
7.83	6.355197327
7.84	5.666169042
7.85	6.563114236
7.86	6.280895636
7.87	7.430041032
7.88	7.148567217
7.89	6.658410317
7.9	6.984911127
7.91	6.33001676
7.92	6.715624499
7.93	7.103694541
7.94	7.325565378
7.95	7.748839477
7.96	8.306915483
7.97	8.017811988
7.98	7.690172548
7.99	8.498537768
8	8.367091386
8.01	8.523779742
8.02	8.638492665
8.03	8.913172591
8.04	9.144216781
8.05	9.313541355
8.06	8.486126606
8.07	9.031667555
8.08	8.235604167
8.09	8.285758567
8.1	7.631393219
8.11	8.900538359
8.12	8.292866532
8.13	9.860622041
8.14	10.34111352
8.15	10.24944819
8.16	8.442729233
8.17	7.470089061
8.18	7.103851105
8.19	6.715491491
8.2	6.329953326
8.21	6.3685814
8.22	7.287794446
8.23	7.512630979
8.24	7.753713048
8.25	7.775829651
8.26	7.568473349
8.27	7.60573268
8.28	6.112494709
8.29	5.603520422
8.3	6.070347459
8.31	6.27883617
8.32	6.946436329
8.33	6.634077642
8.34	6.230247999
8.35	6.015609926
8.36	5.806823474
8.37	5.814433709
8.38	5.778960605
8.39	5.276679911
8.4	6.177597504
8.41	6.318873003
8.42	6.196125316
8.43	6.313640243
8.44	6.390094394
8.45	7.158068952
8.46	8.777186983

    };
    \addplot [thick, blue] table[
    y]
    {
    8.72	10.39008927
8.73	9.564445279
8.74	8.626246963
8.75	6.653377893
8.76	6.145756015
8.77	6.0504436
8.78	5.905493402
8.79	6.619160011
8.8	6.447391457
8.81	7.627162835
8.82	8.530934547
8.83	8.181386103
8.84	7.612420091
8.85	7.613200656
8.86	7.127095924
8.87	7.715426663
8.88	7.86501522
8.89	7.956539415
8.9	8.128033866
8.91	8.030436064
8.92	7.874915035
8.93	7.44428337
8.94	7.410076767
8.95	7.683132169
8.96	8.068881043
8.97	8.235070092
8.98	8.105551072
8.99	8.001575742
9	8.069130643
9.01	8.035670811
9.02	7.818632707
9.03	8.026039137
9.04	5.088128852
9.05	5.257325556
9.06	7.24856194
9.07	7.573488396
9.08	6.996431313
9.09	6.721432213
9.1	5.83038371
9.11	6.861799195
9.12	6.581697234
9.13	7.208086191
9.14	7.645952608
9.15	8.346123569
9.16	8.028698883
9.17	8.056067649
9.18	8.569735522
9.19	8.713955283
9.2	8.927916157
9.21	8.992860318
9.22	9.070666877
9.23	9.119663242
9.24	9.91998752
9.25	9.767146805
9.26	9.444947722
9.27	9.654833913
9.28	8.920009305
9.29	8.95998162
9.3	8.400506147
9.31	8.568539719
9.32	9.187465756
9.33	10.01537262
9.34	11.04793611
9.35	10.1581347
9.36	8.250448072
9.37	7.231763965
9.38	6.853939401
9.39	6.727280681
9.4	6.397821221
9.41	5.943667306
9.42	6.473857929
9.43	6.893084957
9.44	7.156044184
9.45	7.351424007
9.46	7.320880824
9.47	7.176238522
9.48	6.805246739
9.49	5.530590544
9.5	6.526752967
9.51	6.6280173
9.52	5.898566545
9.53	6.36181607
9.54	6.157476714
9.55	5.979566967
9.56	5.543224817
9.57	5.626983294
9.58	5.311038927
9.59	4.919851683
9.6	5.913296023
9.61	6.212306377
9.62	5.793411747
9.63	6.283063267
9.64	5.848837756
9.65	6.747197847
9.66	8.805274949

    };
    
    \end{axis}
\end{tikzpicture}
\caption{hGRFs in the sagittal plane for a walking humanoid robot on two different grounds. Excerpt of 3 steps for each ITshoe. A--laboratory ground; B--carpet.}
\label{g6}
\end{figure}











   
\begin{figure}[h]
\centering
 \begin{tikzpicture}
 \pgfplotsset{width=8cm,compat=newest}
    \begin{axis}[
    title={\textbf{hGRFs transverse plane (locomotion)}},
     legend style={at={(axis cs:4,20.5)},anchor=south west},
    xlabel={Time[s]},
    ylabel={Force[N]},
    ylabel style={at={(ticklabel cs:0.25)}},
    xmin=5,xmax=10,
    ymin=-8,ymax=26,
    xtick={5,6,7,8,9,10},
    ytick={-8,-6, -4, -2, 0 , 2, 4, 6}
    ]
    


    \addplot [thick, red] table[
    y]
    {
5.88	-2.95259608
5.89	-3.375377409
5.9	-3.359399027
5.91	-2.885629293
5.92	-2.425417667
5.93	-2.342902659
5.94	-2.385908898
5.95	-2.728791641
5.96	-2.411678715
5.97	-2.462947699
5.98	-2.346525771
5.99	-2.800705648
6	-2.205091413
6.01	-2.806578668
6.02	-2.689367972
6.03	-3.089323776
6.04	-3.522962957
6.05	-2.702192776
6.06	-3.328206382
6.07	-3.490311275
6.08	-3.301272247
6.09	-4.26603278
6.1	-4.944600033
6.11	-5.368500159
6.12	-6.024753149
6.13	-6.291237072
6.14	-6.879340096
6.15	-7.278405691
6.16	-6.877741363
6.17	-6.468071303
6.18	-5.970993884
6.19	-5.749057062
6.2	-5.386583088
6.21	-4.938373992
6.22	-4.068472693
6.23	-4.563225393
6.24	-3.816120077
6.25	-3.720437825
6.26	-3.171656732
6.27	-3.041990381
6.28	-1.7096403
6.29	-1.227243035
6.3	-1.153758282
6.31	-1.262672152
6.32	-1.125431145
6.33	-1.054612338
6.34	-0.919356619
6.35	-1.642518467
6.36	-1.700189839
6.37	-1.752108227
6.38	-2.010565714
6.39	-2.055810803
6.4	-2.3138926
6.41	-2.339591384
6.42	-2.68189727
6.43	-2.81101136
6.44	-2.888483004
6.45	-2.766269171
6.46	-2.978609264
6.47	-3.282210003
6.48	-3.417926358
6.49	-3.100935591
6.5	-2.990857785
6.51	-2.771320038
6.52	-2.855795865
6.53	-3.049395837
6.54	-2.861899882
6.55	-2.97810342
6.56	-3.126887488
6.57	-2.97160717
6.58	-2.932771609
6.59	-3.191394111
6.6	-2.745455645
6.61	-3.262240919
6.62	-3.036504319
6.63	-3.269834513
6.64	-3.056379189
6.65	-2.868658837
6.66	-2.771327662
6.67	-2.732255793
6.68	-2.662594516
6.69	-2.616238604
6.7	-2.376884496



    };
     \addplot [thick, blue] table[
    y]
    {

6.32	2.015208347
6.33	0.202800948
6.34	0.92120185
6.35	1.342937635
6.36	0.9842831
6.37	0.845163823
6.38	1.264737467
6.39	1.307402629
6.4	1.392178999
6.41	1.717987787
6.42	2.097637709
6.43	2.413152145
6.44	1.907193942
6.45	2.4663618
6.46	2.612552926
6.47	2.188602819
6.48	2.447935238
6.49	2.880571509
6.5	2.651389452
6.51	2.676004753
6.52	3.329084738
6.53	3.68812061
6.54	4.149028204
6.55	4.627400234
6.56	5.229662136
6.57	5.593428975
6.58	5.975933156
6.59	4.974705759
6.6	4.365618355
6.61	3.907841998
6.62	3.92897853
6.63	3.402431622
6.64	3.224186592
6.65	3.392732643
6.66	2.862199068
6.67	2.340114608
6.68	2.017872064
6.69	1.839112168
6.7	0.909249912
6.71	0.222863827
6.72	0.157035011
6.73	-0.194843871
6.74	-0.527495271
6.75	-0.612501073
6.76	-0.456892579
6.77	0.179148028
6.78	2.12929342
6.79	2.389060847
6.8	2.603123986
6.81	2.867497309
6.82	2.780031679
6.83	3.133563962
6.84	3.421115523
6.85	3.589582558
6.86	4.245414743
6.87	4.146244533
6.88	4.095295254
6.89	3.590214036
6.9	3.763788739
6.91	2.78895489
6.92	2.51431256
6.93	2.368017073
6.94	2.657954081
6.95	2.712521784
6.96	2.772855469
6.97	2.850659716
6.98	2.519467773
6.99	2.733445771
7	2.936372867
7.01	2.413352275
7.02	2.652143535
7.03	2.466838907
7.04	2.546534445
7.05	2.535554943
7.06	2.630902642
7.07	2.228160466
7.08	2.346336537
7.09	2.23525766
7.1	1.848615748
7.11	1.746269089
7.12	1.240930114
7.13	0.855466665
7.14	0.141875744

     };
     
    \addplot [thick, red] table[
    y]
    {
  7.05	-1.53064083
7.06	-4.5634179
7.07	-4.08166188
7.08	-3.713638775
7.09	-4.567816193
7.1	-4.463092156
7.11	-3.91939824
7.12	-3.849007577
7.13	-3.439286353
7.14	-2.77762843
7.15	-3.192236832
7.16	-2.575077057
7.17	-1.989954393
7.18	-2.048743468
7.19	-2.268908049
7.2	-1.665170343
7.21	-1.527904786
7.22	-2.109944581
7.23	-2.512824669
7.24	-2.500226491
7.25	-2.75373324
7.26	-3.201627402
7.27	-3.364603533
7.28	-4.124376869
7.29	-4.7583867
7.3	-5.414443155
7.31	-5.780249405
7.32	-7.333782036
7.33	-7.109921482
7.34	-7.023860005
7.35	-6.284429749
7.36	-5.495773888
7.37	-5.532322536
7.38	-4.751912095
7.39	-4.595572238
7.4	-4.161782564
7.41	-3.728323113
7.42	-3.514455305
7.43	-3.32457499
7.44	-3.149314549
7.45	-3.026792401
7.46	-2.776657378
7.47	-0.574581333
7.48	-0.973977417
7.49	-1.368570623
7.5	-1.599517692
7.51	-1.837321448
7.52	-1.686504694
7.53	-2.191463613
7.54	-2.143969156
7.55	-2.553227235
7.56	-2.454698133
7.57	-2.565223758
7.58	-2.939848092
7.59	-3.17313597
7.6	-3.452533509
7.61	-4.100652478
7.62	-4.236695922
7.63	-4.178566908
7.64	-4.158872271
7.65	-4.319886942
7.66	-4.363205554
7.67	-4.064750871
7.68	-4.037850223
7.69	-3.954524853
7.7	-4.155248885
7.71	-3.746667078
7.72	-4.227911504
7.73	-4.222459101
7.74	-4.13788217
7.75	-4.263789568
7.76	-4.093839344
7.77	-3.763183025
7.78	-4.008930413
7.79	-3.945250242
7.8	-4.132690734
7.81	-4.471359889
7.82	-3.834488309
7.83	-3.774398689
7.84	-2.800294539
7.85	-2.747310848
7.86	-1.726702711
7.87	-2.442928917


     };
     
      \addplot [thick, red] table[
    y]
    {
 8.26	-2.881485286
8.27	-3.301747204
8.28	-3.718038303
8.29	-2.862576855
8.3	-3.062578852
8.31	-2.526557417
8.32	-2.449097163
8.33	-2.674950577
8.34	-2.728809164
8.35	-1.995942622
8.36	-2.492968635
8.37	-2.356819436
8.38	-2.142310078
8.39	-2.878743293
8.4	-2.446432611
8.41	-2.568199479
8.42	-3.118263391
8.43	-3.221651095
8.44	-3.253900479
8.45	-3.382134028
8.46	-3.609139988
8.47	-4.155120465
8.48	-4.770531695
8.49	-5.441712232
8.5	-5.642654532
8.51	-6.075938687
8.52	-7.986213869
8.53	-7.633254794
8.54	-7.410047759
8.55	-6.797914698
8.56	-6.840589671
8.57	-6.109598061
8.58	-5.455876389
8.59	-5.454718811
8.6	-4.9573748
8.61	-5.103995248
8.62	-4.361865939
8.63	-4.453981932
8.64	-4.183658857
8.65	-4.187672113
8.66	-3.622557694
8.67	-1.303422369
8.68	-1.237981238
8.69	-1.767149644
8.7	-1.482755115
8.71	-1.811974517
8.72	-1.882727342
8.73	-2.192730469
8.74	-2.483679656
8.75	-2.450892648
8.76	-2.508961166
8.77	-2.75435303
8.78	-2.806088605
8.79	-3.025428703
8.8	-3.044548172
8.81	-3.257427596
8.82	-3.22557842
8.83	-3.173976463
8.84	-3.575141844
8.85	-3.853604251
8.86	-3.723393933
8.87	-3.064460205
8.88	-3.198520733
8.89	-2.907436343
8.9	-3.159851162
8.91	-3.230710857
8.92	-3.081984341
8.93	-3.282633163
8.94	-3.218293523
8.95	-3.276156538
8.96	-3.321425171
8.97	-3.444055897
8.98	-3.437679141
8.99	-3.534276994
9	-3.528309537
9.01	-3.457781962
9.02	-3.665436202
9.03	-3.375098996
9.04	-3.445515904
9.05	-3.400290823
9.06	-3.075996144
9.07	-2.836437527
9.08	-2.420187781


     };

   
      \addplot [thick, blue] table[
    y]
    {
 7.53	2.921175745
7.54	3.314723859
7.55	2.434373818
7.56	2.538103869
7.57	2.336007703
7.58	1.894237166
7.59	2.048054095
7.6	2.159033245
7.61	2.280583725
7.62	2.444699641
7.63	2.323505154
7.64	2.513646599
7.65	2.079897786
7.66	1.954697412
7.67	2.801924729
7.68	2.742849366
7.69	2.78143575
7.7	3.49318756
7.71	3.326303369
7.72	3.407607095
7.73	4.053186348
7.74	4.589275693
7.75	4.570803076
7.76	5.090194483
7.77	5.407202798
7.78	6.203599782
7.79	4.788672259
7.8	4.449561494
7.81	4.335291585
7.82	4.06867247
7.83	4.199919626
7.84	3.929969824
7.85	3.406778949
7.86	3.372825295
7.87	2.917625378
7.88	2.443649031
7.89	2.182306377
7.9	1.54155169
7.91	0.583382818
7.92	0.577287679
7.93	0.446102651
7.94	0.343474381
7.95	0.612323893
7.96	0.95941117
7.97	1.139949539
7.98	1.483470571
7.99	3.223473571
8	3.479793699
8.01	3.803044423
8.02	3.573987437
8.03	3.843401663
8.04	4.171358009
8.05	4.662699066
8.06	4.970614535
8.07	5.034077123
8.08	4.872102639
8.09	4.573717357
8.1	4.252378633
8.11	4.103433816
8.12	3.37472631
8.13	3.717358107
8.14	3.762383372
8.15	3.719316163
8.16	3.880032325
8.17	3.845097326
8.18	3.805733683
8.19	3.542237345
8.2	3.930585681
8.21	3.880217192
8.22	3.706004885
8.23	4.091255789
8.24	3.536775175
8.25	3.435911635
8.26	3.694129179
8.27	3.656407528
8.28	3.166199722
8.29	3.395967203
8.3	2.358923804
8.31	2.067752329
8.32	1.266860778
8.33	0.949320679
8.34	0.789926428
8.35	0.497223478


     };
     
      \addplot [thick, blue] table[
    y]
    {
8.73	1.642796765
8.74	2.312778774
8.75	2.378251598
8.76	1.998422935
8.77	2.010931609
8.78	1.428960724
8.79	0.906336054
8.8	1.41742876
8.81	1.110521472
8.82	1.815855111
8.83	2.246908539
8.84	2.268755565
8.85	2.288717426
8.86	2.472815913
8.87	2.650317488
8.88	2.596105777
8.89	2.851563505
8.9	2.783453602
8.91	2.930929886
8.92	3.342800202
8.93	3.505401529
8.94	3.95145493
8.95	4.417397301
8.96	4.685415884
8.97	4.956405338
8.98	6.375505658
8.99	6.191782383
9	5.509038577
9.01	4.612580039
9.02	4.540000103
9.03	4.22522533
9.04	3.829456768
9.05	3.518568937
9.06	3.341180182
9.07	2.73762087
9.08	2.395943583
9.09	2.264725595
9.1	2.022026682
9.11	1.917571553
9.12	1.851346816
9.13	1.57107119
9.14	0.749315449
9.15	0.452024138
9.16	1.049492945
9.17	1.002419848
9.18	1.434945602
9.19	2.779234553
9.2	3.17852943
9.21	3.423579986
9.22	3.464589414
9.23	3.531724097
9.24	3.930427425
9.25	4.544518978
9.26	4.607527338
9.27	4.542263236
9.28	4.361058095
9.29	4.065388908
9.3	3.921024641
9.31	3.420226287
9.32	2.600599727
9.33	2.46878173
9.34	2.596619184
9.35	2.915560209
9.36	3.14350929
9.37	3.062963603
9.38	3.338666294
9.39	3.022922426
9.4	3.200377946
9.41	3.630654115
9.42	3.48642717
9.43	3.576865162
9.44	3.623029803
9.45	3.376206058
9.46	3.453279101
9.47	3.222436675
9.48	2.452534875
9.49	2.260865099
9.5	2.412070293
9.51	1.851249034
9.52	1.683123407
9.53	1.030287438
9.54	0.006460876
9.55	0.319230696


     };
     

    
     
    \addlegendentry{Left: $Fh_{tt}$}
    \addlegendentry{Right: $Fh_{tt}$}
 \node[black] at (axis cs: 9.75,24) {B};
 \node[black] at (axis cs: 5.3,-6) {A};
\end{axis}
    
    \begin{axis}[
    axis y line*=right,
    axis x line=none,
    ylabel style={at={(ticklabel cs:0.75)}},
    ylabel={Force[N]},
    xmin=5,xmax=10,
    ymin=-8,ymax=26,
    xtick={5,6,7,8,9,10},
    ytick={9,11, 13, 15, 17 , 19, 21, 23},
    yticklabels={-8,-6,-4,-2,0,2,4,6},
    ]

 \addplot [thick, red] table[
    y]
    {
    5.83	14.85118813
5.84	10.26061797
5.85	8.574446788
5.86	8.469527712
5.87	8.552434622
5.88	7.921656976
5.89	9.032392331
5.9	9.296218301
5.91	9.212384096
5.92	9.510456628
5.93	9.831268552
5.94	10.91683738
5.95	12.5452855
5.96	13.57840131
5.97	13.73911009
5.98	13.61488947
5.99	13.53725969
6	13.6732831
6.01	14.33513507
6.02	14.70570651
6.03	14.72582269
6.04	14.89550718
6.05	14.90337856
6.06	14.65808331
6.07	16.39772704
6.08	16.2415095
6.09	15.99456574
6.1	15.87097534
6.11	16.18896318
6.12	16.04625146
6.13	15.94715365
6.14	14.47183606
6.15	15.08075475
6.16	13.02633786
6.17	14.0874484
6.18	14.32851301
6.19	14.8594558
6.2	14.64645503
6.21	14.34816785
6.22	14.63370348
6.23	14.4389078
6.24	14.53612817
6.25	14.45221979
6.26	14.37455788
6.27	14.29917419
6.28	14.70223231
6.29	14.83202718
6.3	14.83346338
6.31	14.94722167
6.32	14.86819043
6.33	13.38695159
6.34	14.13503731
6.35	13.88998863
6.36	14.12943268
6.37	13.56718516
6.38	13.72266876
6.39	13.55461727
6.4	13.53552004
6.41	13.70938508
6.42	13.8329382
6.43	13.67198678
6.44	13.68971139
6.45	13.92813452
6.46	13.74615958
6.47	14.14144339
6.48	14.21923232
6.49	14.46416054
6.5	14.64708463
6.51	14.64719824
6.52	14.62098355
6.53	14.40215506
6.54	14.21470927
6.55	13.99463561
6.56	13.69636804
6.57	13.12070286
6.58	12.43989958
6.59	12.4030279
6.6	11.54015443
6.61	10.69027655
6.62	10.74247957
6.63	10.52078016
6.64	10.09127778
6.65	9.938680656
6.66	9.963955427
6.67	10.20362532
6.68	10.29909809
6.69	10.91862318
6.7	10.87923284
6.71	11.79695524
6.72	11.53276992
6.73	12.06860254
6.74	12.4782081
6.75	13.98194958

};
 \addplot [thick, blue] table[
    y]
    {6.32	19.14726029
6.33	19.8990569
6.34	19.93227618
6.35	20.60314428
6.36	20.50584844
6.37	21.91314047
6.38	20.86818499
6.39	21.21814724
6.4	20.41520785
6.41	19.77893399
6.42	18.69953537
6.43	17.59522376
6.44	17.12854406
6.45	17.07653208
6.46	16.58337263
6.47	16.76307394
6.48	16.62775034
6.49	16.50840805
6.5	16.47943087
6.51	16.5712062
6.52	16.80109263
6.53	16.788986
6.54	16.76560368
6.55	16.93256076
6.56	16.97980647
6.57	17.16101161
6.58	17.36849855
6.59	17.79237576
6.6	18.00699286
6.61	17.5266068
6.62	18.02089802
6.63	18.30804287
6.64	18.64424424
6.65	19.44301886
6.66	22.17038678
6.67	22.27339788
6.68	22.39068471
6.69	22.16119884
6.7	22.42873043
6.71	22.63276479
6.72	22.99817277
6.73	23.53269535
6.74	23.29337085
6.75	22.9471483
6.76	22.30539007
6.77	21.93519132
6.78	21.9328834
6.79	21.31264686
6.8	21.09319062
6.81	20.53949339
6.82	20.33004227
6.83	20.6290168
6.84	20.60668871
6.85	20.25187127
6.86	20.58020895
6.87	20.33612052
6.88	21.06226151
6.89	20.41962399
6.9	20.45538609
6.91	20.11390826
6.92	20.96680711
6.93	21.78457683
6.94	21.82077758
6.95	19.79887514
6.96	18.0841164
6.97	17.46874665
6.98	16.79619981
6.99	16.17807637
7	15.94455927
7.01	15.89684559
7.02	15.01022549
7.03	15.13242965
7.04	17.01955378
7.05	17.01394583
7.06	16.15220194
7.07	16.4166604
7.08	17.46552499
7.09	17.46278968
7.1	17.88299017
7.11	17.71891567
7.12	18.22034087
7.13	18.81172066
7.14	19.44057893
7.15	19.54149465
7.16	21.12337271
7.17	20.93243329
7.18	21.18315857
7.19	21.00022017
7.2	20.55874732
7.21	20.43353932
7.22	20.29567921
7.23	19.12986074
7.24	18.34905612
7.25	17.94131148

    };
    
    \addplot [thick, red] table[
    y]
    {7.03	15.45867659
7.04	10.05068408
7.05	8.67652588
7.06	8.930425312
7.07	8.994665725
7.08	8.544609752
7.09	9.559392619
7.1	9.299420042
7.11	9.085186227
7.12	9.583274338
7.13	10.06451236
7.14	11.13961257
7.15	11.69612623
7.16	13.08632506
7.17	12.90575834
7.18	12.7564173
7.19	12.60754357
7.2	13.46387258
7.21	14.04029283
7.22	14.72277316
7.23	14.55531195
7.24	14.43261805
7.25	15.21840765
7.26	15.15473989
7.27	15.12265958
7.28	15.17545889
7.29	15.12426576
7.3	14.87100723
7.31	15.36313482
7.32	14.84580255
7.33	14.68507191
7.34	13.73554557
7.35	14.17478245
7.36	12.92932423
7.37	13.85093331
7.38	14.03284213
7.39	14.16461078
7.4	14.34600343
7.41	14.66967467
7.42	14.51404065
7.43	14.28019984
7.44	14.22758667
7.45	14.23395263
7.46	14.04547823
7.47	14.18892926
7.48	14.2172459
7.49	14.14729447
7.5	14.27796046
7.51	12.81891739
7.52	12.52140649
7.53	12.08560881
7.54	12.60769022
7.55	12.75433113
7.56	12.40862412
7.57	12.37658207
7.58	12.39064576
7.59	12.42034446
7.6	12.49079969
7.61	12.68391359
7.62	12.53230303
7.63	12.6424191
7.64	12.32125499
7.65	12.12957201
7.66	11.99278349
7.67	12.07711725
7.68	13.81554285
7.69	14.50888455
7.7	14.39247526
7.71	14.35170243
7.72	13.80067073
7.73	13.15726557
7.74	12.39782239
7.75	11.66983896
7.76	11.35092271
7.77	11.73419455
7.78	10.88473395
7.79	10.11314002
7.8	9.70555543
7.81	9.310034335
7.82	9.076190062
7.83	9.04931813
7.84	9.010431024
7.85	9.106749277
7.86	9.235603514
7.87	9.145313213
7.88	9.863301754
7.89	9.916267387
7.9	10.52032847
7.91	10.60139365
7.92	10.35119767
7.93	11.42839383
7.94	11.55923473
7.95	12.50275432

    };
    
    \addplot [thick, red] table[
    y]
    {8.23	15.22340542
8.24	13.10338968
8.25	8.973547303
8.26	11.06458344
8.27	9.292670612
8.28	10.09041747
8.29	10.77210774
8.3	10.8007356
8.31	10.77379655
8.32	10.96789995
8.33	11.40237498
8.34	12.23838232
8.35	13.97711726
8.36	14.19166659
8.37	14.34097847
8.38	14.11549317
8.39	13.7071206
8.4	14.00761558
8.41	14.34557549
8.42	14.42475902
8.43	14.51675069
8.44	14.19555081
8.45	14.24166493
8.46	14.12045609
8.47	14.28905945
8.48	15.01124079
8.49	14.88059898
8.5	14.5170051
8.51	15.34850941
8.52	14.73692375
8.53	13.69428532
8.54	12.73432143
8.55	12.73503683
8.56	12.79226775
8.57	13.15080441
8.58	13.56636866
8.59	14.45466903
8.6	14.90084312
8.61	15.17277107
8.62	14.91938469
8.63	14.98487885
8.64	15.10047627
8.65	14.32952168
8.66	14.83626991
8.67	14.26604092
8.68	14.8186852
8.69	14.38475139
8.7	12.17861683
8.71	12.42488255
8.72	12.31069318
8.73	12.41678253
8.74	12.58570789
8.75	12.71400927
8.76	12.57789859
8.77	12.46914646
8.78	12.372766
8.79	12.313563
8.8	12.78776185
8.81	12.24347384
8.82	12.84903738
8.83	12.49596216
8.84	12.43317639
8.85	12.11502399
8.86	12.00296772
8.87	12.34300853
8.88	14.15563722
8.89	14.27968538
8.9	14.18883412
8.91	14.45319995
8.92	13.93022144
8.93	12.60054737
8.94	12.11974718
8.95	11.87412712
8.96	11.39562737
8.97	11.47959062
8.98	10.94164626
8.99	10.55945887
9	9.623645019
9.01	9.240875346
9.02	9.169499744
9.03	8.948483217
9.04	9.213986992
9.05	9.214662029
9.06	8.985649731
9.07	9.393689996
9.08	9.444928055
9.09	9.575059244
9.1	10.21853747
9.11	10.20146388
9.12	10.54816412
9.13	11.43188588
9.14	11.03120129
9.15	12.46717809

    };
    
    \addplot [thick, blue] table[
    y]
    {7.52	19.22599286
7.53	20.37268717
7.54	20.41404712
7.55	20.1039579
7.56	20.2106994
7.57	21.05177094
7.58	20.14762428
7.59	20.18982296
7.6	19.72265374
7.61	19.07070321
7.62	18.81928054
7.63	18.19677367
7.64	17.90121497
7.65	17.8106312
7.66	17.36583612
7.67	17.18225863
7.68	17.08469048
7.69	17.17215611
7.7	17.19611773
7.71	17.23002997
7.72	17.28892659
7.73	17.71071456
7.74	17.77165437
7.75	18.01601978
7.76	18.05742662
7.77	18.36084056
7.78	18.49852426
7.79	18.63907447
7.8	18.31983725
7.81	17.78369806
7.82	18.32941962
7.83	17.97723459
7.84	18.03962373
7.85	19.55314935
7.86	20.21527
7.87	22.31238631
7.88	22.27178356
7.89	21.96260828
7.9	22.4977226
7.91	22.69258276
7.92	23.25470227
7.93	23.80964094
7.94	23.56435249
7.95	22.99881336
7.96	22.61650186
7.97	22.08600821
7.98	22.23669241
7.99	21.71655291
8	21.16230786
8.01	20.61970674
8.02	20.29646214
8.03	20.16208842
8.04	20.04538934
8.05	20.03101008
8.06	20.40711214
8.07	20.83976912
8.08	21.22553256
8.09	20.65968264
8.1	20.76272521
8.11	20.32571018
8.12	20.54732474
8.13	21.01083181
8.14	22.5216627
8.15	20.98728578
8.16	18.76007016
8.17	17.61326979
8.18	16.63414018
8.19	16.05208416
8.2	15.92199329
8.21	16.02486707
8.22	15.65014377
8.23	16.7316999
8.24	17.21152125
8.25	17.04134166
8.26	16.43418464
8.27	17.20375115
8.28	16.9067668
8.29	17.4582497
8.3	17.82061393
8.31	17.70326267
8.32	18.16982575
8.33	18.13537182
8.34	18.55866128
8.35	18.85093071
8.36	19.2327489
8.37	19.02577599
8.38	19.16806559
8.39	19.12190707
8.4	18.77600523
8.41	18.6544425
8.42	18.95280179
8.43	18.25392161
8.44	17.94014043
8.45	17.55600755

    };
    
    \addplot [thick, blue] table[
    y]
    {8.72	19.520416
8.73	20.07576812
8.74	20.01392513
8.75	20.79247233
8.76	20.70012995
8.77	20.64812644
8.78	20.76353423
8.79	20.23805002
8.8	20.10681132
8.81	19.34129211
8.82	18.39307851
8.83	18.03976341
8.84	17.88836489
8.85	17.33669015
8.86	17.15324998
8.87	17.18923921
8.88	17.21447987
8.89	17.19295587
8.9	17.15187819
8.91	17.09574835
8.92	17.20403212
8.93	17.33913092
8.94	17.60506638
8.95	17.83094692
8.96	18.06262027
8.97	18.38788123
8.98	18.67651999
8.99	18.85679597
9	19.10968951
9.01	19.08459463
9.02	18.56776278
9.03	18.36846023
9.04	18.13083201
9.05	18.70776115
9.06	20.19475568
9.07	22.53846481
9.08	23.3863236
9.09	23.18087837
9.1	22.63411882
9.11	21.80316606
9.12	22.28251944
9.13	22.85051924
9.14	23.93983934
9.15	23.47491973
9.16	22.49764148
9.17	22.3093254
9.18	22.26080912
9.19	22.09606023
9.2	21.40624756
9.21	21.23405161
9.22	20.68733263
9.23	20.72166762
9.24	20.15152872
9.25	20.10838862
9.26	20.52703923
9.27	20.88587714
9.28	21.49468756
9.29	21.5441476
9.3	21.7669683
9.31	21.70059778
9.32	21.40449359
9.33	21.44867567
9.34	22.60898489
9.35	20.92362524
9.36	18.49334584
9.37	17.64980138
9.38	17.07323436
9.39	16.4671715
9.4	16.15341103
9.41	16.47952826
9.42	16.83338507
9.43	16.88132439
9.44	16.63753977
9.45	17.01561447
9.46	16.76617435
9.47	16.7185473
9.48	17.54259405
9.49	16.87819231
9.5	17.27367314
9.51	18.35593363
9.52	17.73253056
9.53	17.80544581
9.54	18.23215574
9.55	18.05895774
9.56	18.72657352
9.57	19.21011152
9.58	19.27117641
9.59	19.22058313
9.6	18.7004751
9.61	18.80766554
9.62	18.86121745
9.63	17.92894524
9.64	18.04025968
9.65	17.91093149

    };
\addplot [thick, black] table[
    y]
    {
    0 6.8
    10 6.8
    };
\end{axis}
\end{tikzpicture}
\caption{hGRFs in the transverse plane for a walking humanoid robot on two different grounds. Excerpt of 3 steps for each ITshoe. A--laboratory ground; B--carpet.}
\label{g7}
\end{figure}

Relatively to the hGRFs in the sagittal plane, a similarity between the data from the LITshoe and the RITshoe is visible for both grounds. This pattern may be useful to recognise different robot walking situations, since the robot balance will tightly affect these forces. The hGRFs in the transverse axis, between the LITshoe and the RITshoe, seems to be opposite but similar in absolute value. This behaviour appears to be correct since the positive direction in this axis for the RITshoe is the negative direction for the LITshoe as stated in Fig. \ref{fig:sta12}.

All these indicators suggest that automatic detection of different floors, among other studies in gait analysis and expectedly also ground friction coefficients, can be done by observing the ground vertical and horizontal reaction forces using these instrumented shoes.
Automatic separation of steps from full sequence of data is fairly easy since the alternation of the stance and swing foot is detectable by the activity of normal forces.

Anyway, with combined or separate steps, the next challenge is to process data and conclude about the floor the robot is walking on. Several tools appear, and some have actually been preliminary tried, suited for that task, such as temporal and spectral analysis, or even machine learning techniques.

\section{Conclusions and Future Work}

This paper presents a system that is able to measure the GRFs, in real-time, of a walking humanoid robot. The ITshoe is a cost-effective, self-contained and low-power battery-powered device that does not affect the balance of the robot. The ITshoe design simplifies its use with a variety of humanoid robots, since it only requires a change in its coupling geometry, that can be extracted from the robot foot in question. Nevertheless, more research, refinements and extensions are desired to improve its functionality and accuracy.

Using computational techniques models, either in spectral analysis or machine learning, in the near future, it is expected to be possible to distinguish different floors by analysing the patterns, obtained by the force sensors, for different grounds. Additionally, it is also expected that this data can be used for several other studies within this theme, to improve the performance and balance of biped humanoid robots.






\end{document}
